UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): October 12, 2005 (October 12, 2005)

Blackboard Inc.

(Exact Name of Registrant as Specified in Charter)			
Delaware	000-50784	52-2081178	
(State or Other Juris-	(Commission	(IRS Employer	
diction of Incorporation)	File Number)	Identification No.)	
1899 L Street, N.W., Washington, D.C.		20036	
(Address of Principal Executive O	ffices)	(Zip Code)	
	elephone number, including area code: (ne or Former Address, if Changed Since		
Check the appropriate box below if the Founder any of the following provisions (<i>see</i> Ge	_	asly satisfy the filing obligation of the registrant	
☐ Written communications pursuant to Ru	ile 425 under the Securities Act (17 CFI	R 230.425)	
☐ Soliciting material pursuant to Rule 14a	-12 under the Exchange Act (17 CFR 2	40.14a-12)	
☐ Pre-commencement communications pu	ursuant to Rule 14d-2(b) under the Exch	ange Act (17 CFR 240.14d-2(b))	
☐ Pre-commencement communications pu	ursuant to Rule 13e-4(c) under the Exch	ange Act (17 CFR 240.13e-4(c))	

Item 1.01. Entry into a Material Definitive Agreement

On October 12, 2005, the Registrant entered into an Agreement and Plan of Merger (the "Agreement") with College Acquisition Sub, Inc., a wholly owned subsidiary of the Registrant (the "Transitory Subsidiary"), and WebCT, Inc. ("WebCT"). The Agreement contemplates that the Registrant will acquire WebCT for a purchase price of \$180 million in cash, less certain expenses and other deductions set forth in the Agreement, pursuant to the merger of the Transitory Subsidiary with and into WebCT (the "Merger"). The Registrant will pay the purchase price from a combination of its working capital and financing to be provided pursuant to the \$80,000,000 Senior Secured Credit Facilities Commitment Letter described below. The Merger is subject to customary closing conditions. The Registrant and WebCT each have a right to terminate the Agreement if a court or governmental agency has issued a final, non-appealable order preventing the consummation of the Merger. In addition, if the Agreement is terminated other than by mutual consent or as a result of a breach by WebCT, the Registrant will be required to pay WebCT \$15,000,000 in liquidated damages.

The foregoing description of the Agreement does not purport to be a complete statement of the parties' rights under the Agreement and is qualified in its entirety by reference to the full text of the Agreement, which is filed with this report as Exhibit 99.1.

On October 12, 2005, the Registrant entered into an \$80,000,000 Senior Secured Credit Facilities Commitment Letter (together with the terms and conditions attached thereto, the "Commitment Letter") with Credit Suisse, Cayman Islands Branch ("Credit Suisse"). The Commitment Letter provides for a \$70,000,000 senior secured term loan facility repayable over six years and a \$10,000,000 senior secured revolving credit facility due and payable in full at the end of five years (together, the "Facilities"). The Commitment Letter is between the Registrant, as Borrower, and Credit Suisse, as sole administrative agent, collateral agent, bookrunner and lead arranger for the Facilities. If the Facilities are rated at or above the ratings threshold set forth in the Commitment Letter, the interest on the Facilities will accrue at one of the following rates selected by the Registrant (a) Adjusted LIBOR plus 2.25%-2.50% or (b) ABR plus 1.25%-1.50%. If the Facilities are rated below such ratings threshold, the interest on the Facilities will accrue and be payable quarterly in arrears at one of the following rates selected by the Registrant (a) Adjusted LIBOR plus 2.50%-2.75% or (b) ABR plus 1.50%-1.75%. "ABR" is the Alternate Base Rate, which is the higher of Credit Suisse's Prime Rate and the Federal Funds Effective Rate plus \(^{1}\)2 of 1.0%. In addition, the Registrant and Credit Suisse have agreed to "flex" terms, allowing Credit Suisse to alter certain provisions of the Facilities, including but not limited to the interest rate, if advisable to ensure a successful syndication of the Facilities.

Under the Commitment Letter, in connection with the Facilities, the Registrant has agreed to pay a commitment fee, payable quarterly after the closing date of the Merger at an initial rate of 0.50% per annum of the undrawn amount of the revolving credit facility.

The Facilities will be guaranteed by all domestic subsidiaries of the Registrant and secured by perfected first priority security interests in, and mortgages on, substantially all tangible and intangible assets of the Registrant (including the capital stock of each subsidiary) and each subsidiary. In addition, the Facilities will contain customary negative covenants applicable to the Registrant and its subsidiaries with respect to their operations and financial condition.

The foregoing description of the Commitment Letter does not purport to be a complete statement of the parties' rights under the Commitment Letter and is qualified in its entirety by reference to the full text of the Commitment Letter, which is filed with this report as Exhibit 99.2.

Item 2.02. Results of Operations and Financial Condition

On October 12, 2005, the Registrant issued a press release reaffirming its financial guidance for the remainder of 2005 and providing initial full year guidance for 2006.

The information in this Item 2.02 and the press release filed as Exhibit 99.3 in Item 9.01 of this Form 8-K shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or otherwise subject to the liabilities of that section, nor shall they be deemed incorporated by reference in any filing under the Securities Act of 1933, as amended, or the Exchange Act, except as expressly set forth by specific reference in such a filing.

Item 7.01. Regulation Disclosure

On October 12, 2005, the Registrant issued a press release announcing that it had entered into the Agreement.

The information in this Item 7.01 and the press release filed as Exhibit 99.4 in Item 9.01 of this Form 8-K shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or otherwise subject to the liabilities of that section, nor shall they be deemed incorporated by reference in any filing under the Securities Act of 1933, as amended, or the Exchange Act, except as expressly set forth by specific reference in such a filing.

Item 9.01. Financial Statements and Exhibits

(d) Exhibits

See Exhibit Index attached hereto.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

BLACKBOARD INC.

Date: October 12, 2005 By: /s/ Matthew H. Small

Senior Vice President for Legal, General

Counsel and Secretary

EXHIBIT INDEX

Exhibit No.	Description
99.1	Agreement and Plan of Merger, dated October 12, 2005, by and among the Registrant, WebCT, Inc. and College Acquisition Sub, Inc.
99.2	Commitment Letter, dated October 12, 2005, by and between the Registrant and Credit Suisse, Cayman Islands Branch
99.3	Press Release regarding financial guidance dated October 12, 2005
99.4	Press Release regarding the Agreement and Plan of Merger dated October 12, 2005

Exhibit 99.1

EXECUTION COPY

AGREEMENT AND PLAN OF MERGER

BY AND AMONG

BLACKBOARD INC.,

WEBCT, INC.,

AND

COLLEGE ACQUISITION SUB, INC.

October 12, 2005

TABLE OF CONTENTS

	Page
ARTICLE I	
THE MERGER	1
1.1 The Merger	1
1.2 The Closing	1
1.3 Actions at the Closing	1
1.4 Additional Action	1
1.5 Conversion of Shares	2
1.6 Dissenting Shares	3
1.7 Exchange of Shares	4
1.8 Options and Warrants	5
1.9 Escrow	6
1.10 Certificate of Incorporation and By-laws	7
1.11 No Further Rights	7
1.12 Closing of Transfer Books	7
1.13 Buyer Financing	7
1.14 Employment Agreements	7
1.15 Non-Competition Agreements	7
1.16 Taxes	7
ARTICLE II	
REPRESENTATIONS AND WARRANTIES OF THE COMPANY	8
2.1 Organization, Qualification and Corporate Power	8
2.2 Capitalization	8
2.3 Authorization of Transaction	10
2.4 Noncontravention	10
2.5 Subsidiaries	11
2.6 Financial Statements	11
2.7 Absence of Certain Changes	12
2.8 Undisclosed Liabilities	12
2.9 Tax Matters	12
2.10 Assets	13
2.11 Owned Real Property	14
2.12 Real Property Leases	14
2.13 Intellectual Property	15
2.14 Inventory	17
2.15 Contracts	17
2.16 Accounts Receivable	19
2.17 Powers of Attorney	19
2.18 Insurance	19
2.19 Litigation	19
2.20 Warranties	20
2.21 Employees	20
-i-	

	Page
2.22 Employee Benefits	21
2.23 Environmental Matters	23
2.24 Legal Compliance	24
2.25 Customers and Suppliers	24
2.26 Permits	24
2.27 Certain Business Relationships With Affiliates	24
2.28 Financial Advisory Fees	25
2.29 Books and Records	25
2.30 Non-Compete Agreements	25
2.31 Government Contracts	25
2.32 Disclosure	26
ARTICLE III	
REPRESENTATIONS AND WARRANTIES OF THE BUYER AND THE TRANSITORY SUBSIDIARY	26
3.1 Organization, Qualification and Corporate Power	26
3.2 Authorization of Transaction	26
3.3 Noncontravention	27
3.4 Buyer Financial Condition	27
3.5 Absence of Material Adverse Change	27
3.6 Litigation	27
3.7 Interim Operations of the Transitory Subsidiary	28
ARTICLE IV	
COVENANTS	28
4.1 Closing Efforts	28
4.2 Governmental and Third-Party Notices and Consents	28
4.3 Stockholder Approval	29
4.4 Operation of Business	30
4.5 Access to Information	32
4.6 Notice of Breaches	33
4.7 Exclusivity	33
4.8 Expenses	34
4.9 Indemnification	34
4.10 Non-Solicitation	35
4.11 FIRPTA	35
4.12 280G Covenant	35
4.13 Termination of Stockholder Agreements	35
ARTICLE V	
CONDITIONS TO CONSUMMATION OF THE MERGER	35
5.1 Conditions to Each Party's Obligations	35
5.2 Conditions to Obligations of the Buyer and the Transitory Subsidiary	36
5.3 Conditions to Obligations of the Company	37
-ii-	

A DOWNEY D. LVI	Page
ARTICLE VI	
INDEMNIFICATION	38
6.1 Indemnification by the Indemnifying Stockholders	38
6.2 Indemnification by the Buyer	39
6.3 Indemnification Claims	39
6.4 Survival of Representations and Warranties	41
6.5 Limitations	42 43
6.6 Limitation on Liability of Indemnification Representatives ARTICLE VII	43
TERMINATION	44
7.1 Termination of Agreement	44
7.2 Effect of Termination	45
7.3 Liquidated Damages	45
ARTICLE VIII	45
8.1 Exclusions from Liability	45
8.2 Liability Cap	45
8.3 Exceptions to Limitations of Liability ARTICLE IX	46
DEFINITIONS	46
DEFINITIONS ARTICLE X	40
MISCELLANEOUS	55
10.1 Press Releases and Announcements	55
10.2 No Third Party Beneficiaries	55
10.3 Entire Agreement	55
10.4 Succession and Assignment	56
10.5 Counterparts and Facsimile Signature	56
10.6 Headings	56
10.7 Notices	56
10.8 Governing Law 10.9 Amendments and Waivers	57 57
10.10 Severability	57
10.11 Submission to Jurisdiction	57
10.12 Construction	57
10.13 Further Assurances	58
Exhibit A — Form of Commitment Letter	
Exhibit B — Form of Non-Competition Agreement	
Exhibit C — Opinion of Counsel to the Company	
Exhibit D — Opinion of Counsel to the Buyer and the Transitory Subsidiary	
Exhibit E — Escrow Agreement	
-iii-	

AGREEMENT AND PLAN OF MERGER

Agreement (the "<u>Agreement</u>") entered into as of October 12, 2005 by and among Blackboard Inc., a Delaware corporation (the "<u>Buyer</u>"), College Acquisition Sub, Inc., a Delaware corporation and a wholly-owned subsidiary of the Buyer (the "<u>Transitory Subsidiary</u>"), and WebCT, Inc., a Delaware corporation (the "<u>Company</u>").

This Agreement contemplates a merger of the Transitory Subsidiary into the Company. In such merger, the stockholders of the Company will receive cash in exchange for their capital stock of the Company.

Now, therefore, in consideration of the representations, warranties and covenants herein contained, the Parties agree as follows.

ARTICLE I THE MERGER

- 1.1 <u>The Merger</u>. Upon and subject to the terms and conditions of this Agreement, the Transitory Subsidiary shall merge with and into the Company at the Effective Time. From and after the Effective Time, the separate corporate existence of the Transitory Subsidiary shall cease and the Company shall continue as the Surviving Corporation. The Merger shall have the effects set forth in Section 259 of the Delaware General Corporation Law.
- 1.2 <u>The Closing</u>. The Closing shall take place at the offices of Wilmer Cutler Pickering Hale and Dorr LLP in Washington, D.C., commencing at 9:00 a.m. local time on the Closing Date.
 - 1.3 Actions at the Closing. At the Closing:
- (a) the Company shall deliver to the Buyer and the Transitory Subsidiary the various certificates, instruments and documents referred to in Section 5.2;
- (b) the Buyer and the Transitory Subsidiary shall deliver to the Company the various certificates, instruments and documents referred to in Section 5.3;
 - (c) the Surviving Corporation shall file with the Secretary of State of the State of Delaware the Certificate of Merger;
- (d) the Buyer or the Surviving Corporation shall pay (by check or by wire transfer) to each Company Stockholder the Initial Cash to which such Company Stockholder is entitled pursuant to Section 1.5; and
- (e) the Buyer, the Indemnification Representatives and the Escrow Agent shall execute and deliver the Escrow Agreement and the Buyer or the Transitory Subsidiary shall deposit the aggregate Escrow Cash with the Escrow Agent in accordance with Section 1.9.
- 1.4 <u>Additional Action</u>. The Surviving Corporation may, at any time after the Effective Time, take any action, including executing and delivering any document, in the name

and on behalf of either the Company or the Transitory Subsidiary, in order to consummate the transactions contemplated by this Agreement.

- 1.5 <u>Conversion of Shares</u>. At the Effective Time, by virtue of the Merger and without any action on the part of any Party or the holder of any of the following securities:
- (a) Each Company Share issued and outstanding immediately prior to the Effective Time (other than Company Shares owned beneficially by the Buyer or the Transitory Subsidiary, Dissenting Shares and Company Shares held in the Company's treasury) shall be converted into and represent the right to receive (subject to the provisions of Section 1.9) an amount of cash equal to the Total Consideration Per Share for the respective class or series (and any issuance date within a series) of Company Shares.
- (b) Company Stockholders shall be entitled to receive as of the Closing for each Company Share held by them, a portion of the cash into which such Company Share was converted pursuant to this Section 1.5 equal to the Initial Cash Per Share, rounded in the aggregate for each Company Stockholder to the nearest \$0.01 (the "Initial Cash"); the remainder of the cash into which their Company Shares were converted pursuant to this Section 1.5 (the "Escrow Cash") shall be deposited in escrow pursuant to Section 1.9 and shall be held and disposed of in accordance with the terms of the Escrow Agreement. The Escrow Cash and the Initial Cash are herein referred to as the "Consideration".
- (c) On or before the fifth business day prior to Closing, the Company will deliver to the Buyer a schedule (the "<u>Distribution Schedule</u>") setting forth the Initial Cash, the Initial Cash Per Share, the Escrow Cash and the Total Consideration Per Share attributable to each class or series (and any issuance date within a series) of Company Shares in accordance with the Company's certificate of incorporation then in effect. The Buyer shall be entitled to rely exclusively on the Distribution Schedule in making distributions of Consideration pursuant to this Section 1.5. The Distribution Schedule will provide that the Escrow Cash will be allocated to each class and series (and issuance date within a series) of Company Shares pro rata in accordance with the allocation of the Initial Cash Per Share.
 - (d) For purposes of this Agreement, each of the following terms shall have the meaning set forth below:
 - "Banking Fee Deduction" shall mean one-half of the aggregate amount of any financial advisory and investment banking fees paid or incurred by the Company in connection with or as a result of this Agreement and the Merger.
 - "Initial Cash Amount" shall mean the Total Consideration less \$18 million, subject to Section 1.9(b).
 - "Initial Cash Per Share" shall mean the portion of the Initial Cash Amount to which each Company Share is entitled as set forth on the Distribution Schedule.
 - "<u>Total Consideration</u>" shall mean an amount equal to \$180 million minus (i) the Banking Fee Deduction, (ii) any payments made by the Company or any Subsidiary or

-2-

due from the Company or any Subsidiary to employees of the Company in connection with the Merger, excluding any payments set forth on Schedule 1.5 hereto, and (iii) any cash payments made by the Company or any Subsidiary prior to or at the Effective Time in connection with the termination of the Options and Warrants as contemplated by Section 1.8.

"Total Consideration Per Share" shall mean the portion of the Total Consideration to which each Company Share is entitled as set forth on the Distribution Schedule (it being understood that such Total Consideration Per Share may vary among classes and series of Company Shares and among holders of any class or series depending on the applicable liquidation preference thereof; provided that the Total Consideration Per Share will be identical for each Company Share of the same class or series of Company Shares, except with respect to any series issued with multiple issuance dates, in which case, the Total Consideration Per Share will be identical for each Company Share with the same issuance date).

- (e) Each Company Share held in the Company's treasury immediately prior to the Effective Time and each Company Share owned beneficially by the Buyer or the Transitory Subsidiary shall be cancelled and retired without payment of any consideration therefor.
- (f) Each share of common stock, \$0.01 par value per share, of the Transitory Subsidiary issued and outstanding immediately prior to the Effective Time shall be converted into and thereafter evidence one share of common stock, \$0.01 par value per share, of the Surviving Corporation.

1.6 Dissenting Shares.

(a) Dissenting Shares shall not be converted into or represent the right to receive the Consideration, unless the Company Stockholder holding such Dissenting Shares shall have forfeited his, her or its right to appraisal under the Delaware General Corporation Law or properly withdrawn, his, her or its demand for appraisal. If such Company Stockholder has so forfeited or withdrawn his, her or its right to appraisal of Dissenting Shares, then (i) as of the occurrence of such event, such holder's Dissenting Shares shall cease to be Dissenting Shares and shall be converted into and represent the right to receive the Consideration issuable in respect of such Company Shares pursuant to Section 1.5, and (ii) promptly following the occurrence of such event, the Buyer shall pay (by check or by wire transfer) to the Payment Agent, if such event occurs prior to six months after the Closing Date, or to such holder if such event occurs thereafter for each such Company Share a portion of the cash into which such Company Share was converted pursuant to Section 1.5 equal to the Initial Cash Per Share, rounded in the aggregate for each such Company Stockholder to the nearest \$0.01 (which cash shall be considered Initial Cash for all purposes of this Agreement); the remainder of the cash into which such holder's Company Shares were converted pursuant to Section 1.5 shall be deposited in escrow pursuant to Section 1.9 and shall be held and disposed of in accordance with the terms of the Escrow Agreement (which cash shall be considered Escrow Cash for all purposes of this Agreement).

-3-

(b) The Company shall give the Buyer (i) prompt notice of any written demands for appraisal of any Company Shares, withdrawals of such demands, and any other instruments that relate to such demands received by the Company and (ii) the opportunity to direct all negotiations and proceedings with respect to demands for appraisal under the Delaware General Corporation Law. The Company shall not, except with the prior written consent of the Buyer, make any payment with respect to any demands for appraisal of Company Shares or offer to settle or settle any such demands.

1.7 Exchange of Shares.

- (a) Prior to the Effective Time, the Buyer shall appoint the Payment Agent to effect the payment of the Initial Cash to the holders of Company Shares. On the Closing Date, the Buyer shall deliver to the Payment Agent, in trust for the benefit of holders of Certificates, and holders of Options who exercise such Options prior to the Effective Time as contemplated in Section 1.8 but do not hold Certificates, the Initial Cash Amount. As soon as practicable after the Effective Time, the Buyer shall cause the Payment Agent to send a notice and a transmittal form to each holder of a Certificate, and each holder of an Option who exercised such Option prior to the Effective Time as contemplated in Section 1.8 but does not hold a Certificate, advising such holder of the effectiveness of the Merger and the procedure for surrendering to the Payment Agent such Certificate in exchange for the Initial Cash payable to such Certificate holder pursuant to Section 1.5. Each holder of a Certificate, upon proper surrender thereof to the Payment Agent in accordance with the instructions in such notice, shall be entitled to receive in exchange therefor (subject to any taxes required to be withheld) the Initial Cash payable to such holder pursuant to Section 1.5. Until properly surrendered, each such Certificate shall be deemed for all purposes to evidence only the right to receive the Consideration payable pursuant to Section 1.5. Holders of Certificates shall not be entitled to receive the Consideration to which they would otherwise be entitled until such Certificates are properly surrendered, provided, however, that to the extent permitted under Delaware law and the Company's certificate of incorporation and by-laws, the parties acknowledge and agree that holders of Options who exercise such Options prior to the Effective Time as contemplated in Section 1.8 shall not receive Certificates in connection with such exercises but shall nevertheless be entitled to receive Consideration payable pursuant to Section 1.5.
- (b) If any Consideration is to be issued in the name of a person other than the person in whose name the Certificate surrendered in exchange therefor is registered, it shall be a condition to the payment of such Consideration that (i) the Certificate so surrendered shall be transferable, and shall be properly assigned, endorsed or accompanied by appropriate stock powers, (ii) such transfer shall otherwise be proper and (iii) the person requesting such transfer shall pay to the Payment Agent any transfer or other taxes payable by reason of the foregoing or establish to the satisfaction of the Payment Agent that such taxes have been paid or are not required to be paid. Notwithstanding the foregoing, neither the Payment Agent nor any Party shall be liable to a holder of Company Shares for any Consideration payable to such holder pursuant to Section 1.5 that are delivered to a public official pursuant to applicable abandoned property, escheat or similar laws.
- (c) In the event any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming such Certificate to be lost, stolen or

-4

destroyed, the Payment Agent shall pay in exchange for such lost, stolen or destroyed Certificate the Initial Cash issuable in exchange therefor pursuant to Section 1.5. The Board of Directors of the Buyer may, in its discretion and as a condition precedent to the issuance thereof, require the owner of such lost, stolen or destroyed Certificate to give the Buyer or the Payment Agent a bond in such sum as it may reasonably direct (it being agreed that any amount required by the Payment Agent shall be deemed reasonable) as indemnity against any claim that may be made against the Buyer with respect to the Certificate alleged to have been lost, stolen or destroyed.

- (d) Any portion of the Initial Cash Amount which remains undistributed to the holders of Certificates for six months after the Effective Time shall be delivered to the Buyer, upon demand, and any holder of Certificates who has not previously complied with this Section 1.7 shall thereafter look only to the Buyer, as a general unsecured creditor, for payment of its claim for Initial Cash.
- (e) To the extent permitted by applicable law, none of the Buyer, the Transitory Subsidiary, the Company, the Surviving Corporation or the Payment Agent shall be liable to any holder of Certificates, as the case may be, for such portion of the Initial Cash delivered to a public official pursuant to any applicable abandoned property, escheat or similar law. If any Certificate shall not have been surrendered prior to two years after the Effective Time (or immediately prior to such earlier date on which any cash payable to the holder of such Certificate pursuant to this Article I would otherwise escheat to or become the property of any Governmental Entity), any such cash in respect of such Certificate shall, to the extent permitted by applicable law, become the property of the Surviving Corporation, free and clear of all claims or interest of any person previously entitled thereto.
- (f) Each of the Buyer and the Surviving Corporation shall be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Agreement to any holder of Certificates such amounts as it reasonably determines that it is required to deduct and withhold with respect to the making of such payment under the Code, or any other applicable provision of law. To the extent that amounts are so withheld by the Surviving Corporation or the Buyer, as the case may be, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the holder of the Certificates in respect of which such deduction and withholding was made by the Surviving Corporation or the Buyer, as the case may be.

1.8 Options and Warrants.

(a) The Company shall cause the termination, as of the Effective Time, of all then unexercised Options. The Company may permit the holders of Options to exercise such Options at any time prior to the Effective Time on a net-exercise basis. To the extent permitted under Delaware law and the Company's certificate of incorporation and by-laws, the holders of such Options who so exercise such Options prior to the Effective Time shall be included on the Distribution Schedule and shall be entitled to receive Consideration in accordance with Section 1.5 without the requirement that the holders surrender Certificates pursuant to Section 1.7. For the avoidance of doubt, any net exercise of Options prior to the Effective Time shall not affect the Total Consideration, subject to Section 1.16. Any cash payment by the Company or any

-5-

Subsidiary in connection with the termination of any Option, however, shall reduce the Total Consideration as provided in clause (iii) of the definition of Total Consideration.

- (b) The Company shall cause the termination, as of the Effective Time, of all then unexercised Warrants. The Company may permit the holders of Warrants to exercise such Warrants at any time prior to the Effective Time on a net-exercise basis. For the avoidance of doubt, any net exercise of Warrants prior to the Effective Time shall not affect the Total Consideration. Any cash payment by the Company or any Subsidiary in connection with the termination of any Warrant, however, shall reduce the Total Consideration as provided in clause (iii) of the definition of Total Consideration.
- (c) The Company shall obtain, prior to the Closing, the consent from each holder of an Option or a Warrant to the termination of such Option or Warrant pursuant to this Section 1.8 (unless such consent is not required under the terms of the applicable agreement, instrument or plan).

1.9 Escrow.

- (a) On the Closing Date, the Buyer shall deposit with the Escrow Agent the aggregate Escrow Cash, for the purpose of securing the indemnification obligations of the Indemnifying Stockholders set forth in this Agreement. The aggregate Escrow Cash, along with any interest earned thereon while it is held in escrow, is referred to herein as the "Escrow Fund." The Escrow Fund shall be held by the Escrow Agent under the Escrow Agreement pursuant to the terms thereof. The Escrow Fund shall be held as a trust fund and shall not be subject to any lien, attachment, trustee process or any other judicial process of any creditor of any party, and shall be held and disbursed solely for the purposes and in accordance with the terms of the Escrow Agreement.
 - (b) Notwithstanding anything herein to the contrary,
- (i) in the event that there are Dissenting Shares representing greater than 5% and not more than 10% of the outstanding Common Shares immediately prior to the Effective Time (calculated on a basis as if all outstanding Preferred Shares were converted into Common Shares), then the Initial Cash Amount shall be reduced by \$500,000 and the aggregate Escrow Cash shall be increased by \$500,000; or
- (ii) in the event that there are Dissenting Shares representing greater than 10% of the outstanding Common Shares immediately prior to the Effective Time (calculated on a basis as if all outstanding Preferred Shares were converted into Common Shares), then the Initial Cash Amount shall be reduced by \$1,000,000 and the aggregate Escrow Cash shall be increased by \$1,000,000;

provided, however, that any amounts placed in escrow pursuant to clause (i) or (ii) above shall be (x) available to cover only Damages described in Section 6.1(e) and shall not be available for any other purpose under Article VI or otherwise, and (y) released from escrow promptly upon any such amount no longer being required to indemnify the Buyer with respect to Damages described in Section 6.1(e).

-6-

(c) The adoption of this Agreement and the approval of the Merger by the stockholders of the Company shall constitute approval of the Escrow Agreement and of all of the arrangements relating thereto, including the placement of the aggregate Escrow Cash in escrow, the appointment of the Indemnification Representatives and the grant to the Indemnification Representatives of the power and authority to act on behalf of the Indemnifying Stockholders in such capacity pursuant to the Escrow Agreement.

1.10 Certificate of Incorporation and By-laws.

- (a) The certificate of incorporation of the Surviving Corporation immediately following the Effective Time shall be the same as the certificate of incorporation of the Transitory Subsidiary immediately prior to the Effective Time, except that (i) the name of the corporation set forth therein shall be changed to the name of the Company and (ii) the identity of the incorporator shall be deleted.
- (b) The by-laws of the Surviving Corporation immediately following the Effective Time shall be the same as the by-laws of the Transitory Subsidiary immediately prior to the Effective Time, except that the name of the corporation set forth therein shall be changed to the name of the Company.
- 1.11 No Further Rights. From and after the Effective Time, no Company Shares shall be deemed to be outstanding, and holders of Certificates shall cease to have any rights with respect thereto, except as provided herein or by law.
- 1.12 <u>Closing of Transfer Books</u>. At the Effective Time, the stock transfer books of the Company shall be closed and no transfer of Company Shares shall thereafter be made. If, after the Effective Time, Certificates are presented to the Buyer or the Surviving Corporation, they shall be cancelled and exchanged for Consideration in accordance with Section 1.5, subject to Section 1.9 and to applicable law in the case of Dissenting Shares.
- 1.13 <u>Buyer Financing</u>. The Company hereby acknowledges that the Buyer has delivered to the Company a financing commitment letter substantially in the form attached hereto as Exhibit A (the "Commitment Letter").
- 1.14 <u>Employment Agreements</u>. The Buyer hereby acknowledges that the Transitory Subsidiary and the Company employees listed on Schedule 1.14 hereto have entered into employment or consulting agreements (or amendments to existing such agreements), conditioned upon the closing of the Merger.
- 1.15 <u>Non-Competition Agreements</u>. The Buyer hereby acknowledges that the Buyer and the Company employees listed on Schedule 1.15 hereto have entered into non-competition agreements substantially in the form attached hereto as <u>Exhibit B</u>.
- 1.16 <u>Taxes</u>. Notwithstanding any other provision in this Agreement, the Buyer, the Surviving Corporation, the Company, the Transitory Subsidiary, the Payment Agent and the Escrow Agent shall have the right to deduct and withhold Taxes from any payment of

-7-

Consideration to be made hereunder (including any payments to be made under the Escrow Agreement) if such withholding is required by law with respect to such payment or with respect to either the issuance or transfer of the Company Shares for which such payment is being made and to collect any necessary Tax forms, including Form W-9 or the appropriate series of Form W-8, as applicable, or any similar information, from the Company Stockholders and any other recipients of payments hereunder or contemplated hereby. To the extent that amounts are so withheld, such withheld amounts shall be treated for all purposes of this Agreement as having been delivered and paid to the Company Stockholder or other recipient of payments in respect of which such deduction and withholding was made.

ARTICLE II REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company represents and warrants to the Buyer that, except as set forth in the Disclosure Schedule, the statements contained in this Article II are true and correct as of the date of this Agreement and will be true and correct as of the Closing as though made as of the Closing, except to the extent such representations and warranties are specifically made as of a particular date (in which case such representations and warranties will be true and correct as of such date). The Disclosure Schedule shall be arranged in sections and subsections corresponding to the numbered and lettered sections and subsections contained in this Article II. The disclosures in any section or subsection of the Disclosure Schedule shall qualify only the corresponding section or subsection in this Article II and such other sections or subsections to the extent expressly cross-referenced or otherwise expressly stated therein.

2.1 Organization, Qualification and Corporate Power. The Company is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. The Company is duly qualified to conduct business and is in good standing under the laws of each jurisdiction listed in Section 2.1 of the Disclosure Schedule, which jurisdictions constitute the only jurisdictions in which the nature of the Company's businesses or the ownership or leasing of its properties requires such qualification, except where the failure to be so qualified or in good standing would not have, individually or in the aggregate, a Company Material Adverse Effect. The Company has all requisite corporate power and authority to carry on the businesses in which it is engaged and to own and use the properties owned and used by it. The Company has furnished to the Buyer complete and accurate copies of its certificate of incorporation and by-laws. Except as set forth in Section 2.1 of the Disclosure Schedule, the Company is not in default under or in violation of any provision of its certificate of incorporation or by-laws.

2.2 Capitalization.

(a) The authorized capital stock of the Company as of the date of this Agreement consists of (i) 41,400,000 Common Shares, of which 1,767,375 shares were issued and outstanding; and (ii) 24,745,778 Preferred Shares, of which (A) 58,891 shares have been designated as Series A Convertible Preferred Stock, all of which were issued and outstanding; (B) 280,000 shares have been designated as Series B Convertible Preferred Stock, of which 274,964 shares were issued and outstanding; (C) 49,785 shares have been designated as Series B1 Convertible Preferred Stock, none of which were issued and outstanding; (D) 3,805,658 shares have been designated as Series C Convertible Participating Preferred Stock, of which

-8-

3,805,648 were issued and outstanding; (E) 406,555 shares have been designated as Series C1 Convertible Participating Preferred Stock, none of which were issued and outstanding; (F) 724,888 shares have been designated as Series D Convertible Preferred Stock, none of which were issued and outstanding; (G) 4,750,000 shares have been designated as Series E Convertible Preferred Stock, of which 4,670,688 shares were issued and outstanding; (H) 700,000 shares have been designated as Series E1 Convertible Preferred Stock, none of which were issued and outstanding; (I) 1,470,000 shares have been designated as Series F Convertible Preferred Stock, of which 1,132,180 shares were issued and outstanding; (J) 8,500,000 shares have been designated as Series G Convertible Preferred Stock, of which 6,633,309 shares were issued and outstanding; (K) one share has been designated as Special Voting Preferred Stock and such share was issued and outstanding; and (L) 4,000,000 of which remain undesignated.

- (b) Section 2.2 of the Disclosure Schedule sets forth a complete and accurate list, as of the date of the Agreement, of the holders of capital stock of the Company, showing the number of shares of capital stock, and the class or series of such shares, held by each stockholder and (for shares other than Common Stock) the number of Common Shares (if any) into which such shares are convertible. None of the outstanding Common Shares are subject to a repurchase or redemption right on the part of the Company. All of the issued and outstanding shares of capital stock of the Company have been duly authorized and validly issued and are fully paid and nonassessable. All of the issued and outstanding shares of capital stock of the Company have been offered, issued and sold by the Company in compliance with all applicable federal and state securities laws.
- (c) Section 2.2 of the Disclosure Schedule sets forth a complete and accurate list, as of the date of this Agreement of: (i) all Company Stock Plans, indicating for each Company Stock Plan the number of Common Shares issued to date under such Plan, the number of Common Shares subject to outstanding Options under such Plan and the number of Common Shares reserved for future issuance under such Plan; (ii) all holders of outstanding Options, indicating with respect to each Option the Company Stock Plan under which it was granted, the number of Common Shares subject to such Option, the exercise price, the date of grant, and the vesting schedule (including any acceleration provisions with respect thereto); and (iii) all holders of outstanding Warrants, indicating with respect to each Warrant the agreement or other document under which it was granted, the number of shares of capital stock, and the class or series of such shares, subject to such Warrant, the exercise price, the date of issuance and the expiration date thereof. The Company has delivered or, if not delivered, made available to the Buyer complete and accurate copies of all Company Stock Plans, forms of all stock option agreements evidencing Options and all Warrants. All of the shares of capital stock of the Company subject to Options and Warrants will be, upon issuance pursuant to the exercise of such instruments, duly authorized, validly issued, fully paid and nonassessable.
- (d) Except as set forth in this Section 2.2 or in Section 2.2 of the Disclosure Schedule, (i) no subscription, warrant, option, convertible security or other right (contingent or otherwise) to purchase or acquire any shares of capital stock of the Company is authorized or outstanding, (ii) the Company has no obligation (contingent or otherwise) to issue any subscription, warrant, option, convertible security or other such right, or to issue or distribute to holders of any shares of its capital stock any evidences of indebtedness or assets of the Company, (iii) the Company has no obligation (contingent or otherwise) to purchase, redeem or

-9-

otherwise acquire any shares of its capital stock or any interest therein or to pay any dividend or to make any other distribution in respect thereof, and (iv) there are no outstanding or authorized stock appreciation, phantom stock or similar rights with respect to the Company.

- (e) Except as set forth in Section 2.2 of the Disclosure Schedule, there is no agreement, written or oral, between the Company and any holder of its securities, or, to the best of the Company's knowledge, among any holders of its securities, relating to the sale or transfer (including agreements relating to rights of first refusal, co-sale rights or "drag-along" rights), registration under the Securities Act, or voting, of the capital stock of the Company.
- (f) Except as set forth in Section 2.2 of the Disclosure Schedule, the Company has not repurchased any capital stock of the Company.
- 2.3 <u>Authorization of Transaction</u>. The Company has all requisite power and authority to execute and deliver this Agreement and to perform its obligations hereunder. The execution and delivery by the Company of this Agreement and, subject to obtaining the Requisite Stockholder Approval, the consummation by the Company of the transactions contemplated hereby have been duly and validly authorized by all necessary corporate action on the part of the Company. Without limiting the generality of the foregoing, the Board of Directors of the Company, at a meeting duly called and held, by the requisite vote of all directors (which vote results shall be set forth on Section 2.3 of the Disclosure Schedule) (i) determined that the Merger is fair and in the best interests of the Company and its stockholders, (ii) adopted this Agreement in accordance with the provisions of the Delaware General Corporation Law, and (iii) directed that this Agreement and the Merger be submitted to the stockholders of the Company for their adoption and approval and resolved to recommend that the stockholders of the Company vote in favor of the adoption of this Agreement and the approval of the Merger. This Agreement has been duly and validly executed and delivered by the Company and constitutes a valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, except to the extent that such enforceability may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the enforcement of creditors' rights generally and to general equitable principles.
- 2.4 <u>Noncontravention</u>. Subject to compliance with the applicable requirements of the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "<u>HSR Act</u>"), as well as premerger notification laws and regulations of other Governmental Entities, if applicable, and the filing of the Certificate of Merger as required by the Delaware General Corporation Law, and except as set forth in Section 2.4 of the Disclosure Schedule, neither the execution and delivery by the Company of this Agreement, nor the consummation by the Company of the transactions contemplated hereby, will (a) conflict with or violate any provision of the certificate of incorporation or by-laws of the Company or the charter, by-laws or other organizational document of any Subsidiary, (b) require on the part of the Company or any Subsidiary any notice to or filing with, or any permit, authorization, consent or approval of, any Governmental Entity, (c) conflict with, result in a breach of, constitute (with or without due notice or lapse of time or both) a default under, result in the acceleration of obligations under, create in any party the right to terminate, modify or cancel, or require any notice, consent or waiver under, any contract or instrument to which the Company or any Subsidiary is a party or by which the Company or any Subsidiary is bound or to which any of their respective assets is subject, (d) result in the

-10-

imposition of any Security Interest upon any assets of the Company or any Subsidiary or (e) violate any order, writ, injunction, decree, statute, rule or regulation applicable to the Company, any Subsidiary or any of their respective properties or assets, except in the case of clause (c) above as would not have, individually or in the aggregate, a Company Material Adverse Effect.

2.5 Subsidiaries.

- (a) Section 2.5 of the Disclosure Schedule sets forth: (i) the name of each Subsidiary; (ii) the number and type of outstanding equity securities of each Subsidiary and a list of the holders thereof; (iii) the jurisdiction of organization of each Subsidiary; (iv) the names of the officers and directors of each Subsidiary; and (v) the jurisdictions in which each Subsidiary is qualified or holds licenses to do business as a foreign corporation or other entity.
- (b) Each Subsidiary is a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation. Each Subsidiary is duly qualified to conduct business and is in good standing under the laws of each jurisdiction in which the nature of its businesses or the ownership or leasing of its properties requires such qualification, except where such failure to be so qualified or in good standing would not have, individually or in the aggregate, a Company Material Adverse Effect. Each Subsidiary has all requisite power and authority to carry on the businesses in which it is engaged and to own and use the properties owned and used by it. The Company has delivered or, if not delivered, made available to the Buyer complete and accurate copies of the charter, by-laws or other organizational documents of each Subsidiary. No Subsidiary is in default under or in violation of any provision of its charter, by-laws or other organizational documents. All of the issued and outstanding shares of capital stock of each Subsidiary are duly authorized, validly issued, fully paid, nonassessable and free of preemptive rights. All shares of each Subsidiary that are held of record or owned beneficially by either the Company or any Subsidiary are held or owned free and clear of any restrictions on transfer (other than restrictions under the Securities Act and state securities laws), claims, Security Interests, options, warrants, rights, contracts, calls, commitments, equities and demands. Except as set forth in Section 2.5(a) of the Disclosure Schedule, there are no outstanding or authorized options, warrants, rights, agreements or commitments to which the Company or any Subsidiary is a party or which are binding on any of them providing for the issuance, disposition or acquisition of any capital stock of any Subsidiary. There are no outstanding stock appreciation, phantom stock or similar rights with respect to any Subsidiary. There are no voting trusts, proxies or other agreements or understandings with respect to the voting of any capital stock of any Subsidiary.
- (c) The Company does not control directly or indirectly or have any direct or indirect equity participation or similar interest in any corporation, partnership, limited liability company, joint venture, trust or other business association or entity that is not a Subsidiary.
- 2.6 <u>Financial Statements</u>. True and correct copies of the Financial Statements are attached to Section 2.6 of the Disclosure Schedule. The Financial Statements have been prepared in accordance with GAAP applied on a consistent basis (except to the extent set forth in the footnotes thereto) throughout the periods covered thereby, fairly present the consolidated financial condition, results of operations and cash flows of the Company and the Subsidiaries as

-11-

of the respective dates thereof and for the periods referred to therein and are consistent with the books and records of the Company and the Subsidiaries; <u>provided</u>, <u>however</u>, that the Financial Statements referred to in clause (b) of the definition of such term are subject to normal recurring year-end adjustments consistent with past practice and do not include footnotes. Without limiting the generality of the foregoing, the deferred revenues reflected on the Financial Statements have been recorded in accordance with GAAP applied on a consistent basis throughout the periods covered thereby (except to the extent set forth in the footnotes thereto).

- 2.7 <u>Absence of Certain Changes</u>. Since the Most Recent Balance Sheet Date, (a) there has occurred no event or development which, individually or in the aggregate, has had, or could reasonably be expected to have in the future, a Company Material Adverse Effect, and (b) except as set forth in Section 4.4 of the Disclosure Schedule, neither the Company nor any Subsidiary has taken any of the actions set forth in paragraphs (a) through (o) of Section 4.4.
- 2.8 <u>Undisclosed Liabilities</u>. Except as set forth in Section 2.8 of the Disclosure Schedule, none of the Company and its Subsidiaries has any liability (whether known or unknown, whether absolute or contingent, whether liquidated or unliquidated and whether due or to become due), except for (a) liabilities shown on the Most Recent Balance Sheet or described in the footnotes thereto, (b) liabilities which have arisen since the Most Recent Balance Sheet Date in the Ordinary Course of Business and (c) contractual and other liabilities incurred in the Ordinary Course of Business which are not required by GAAP to be reflected on a balance sheet (other than, in the case of (b) and (c), contractual and other liabilities in connection with the transactions contemplated by this Agreement).

2.9 Tax Matters.

- (a) Except as set forth in Section 2.9 of the Disclosure Schedule, each of the Company and the Subsidiaries has filed on a timely basis all U.S. federal, state, Canadian and provincial Tax Returns and other material Tax Returns that it was required to file, and all such Tax Returns were complete and accurate in all material respects. Neither the Company nor any Subsidiary is or has ever been a member of a group of corporations with which it has filed (or been required to file) consolidated, combined or unitary Tax Returns, other than a group of which only the Company and the Subsidiaries are or were members. Each of the Company and the Subsidiaries has paid on a timely basis all Taxes that were due and payable. The unpaid Taxes of the Company and the Subsidiaries for tax periods through the Most Recent Balance Sheet Date do not exceed the accruals and reserves for Taxes (excluding accruals and reserves for deferred Taxes established to reflect timing differences between book and Tax income) set forth on the Most Recent Balance Sheet. Neither the Company nor any Subsidiary has any actual or potential liability for any Tax obligation of any taxpayer (including any affiliated group of corporations or other entities that included the Company or any Subsidiary during a prior period) other than the Company and the Subsidiaries. Except as set forth on Section 2.9(a) of the Disclosure Schedule, all Taxes that the Company or any Subsidiary is or was required by law to withhold or collect have been duly withheld or collected and, to the extent required, have been paid to the proper Governmental Entity.
- (b) The Company has delivered to the Buyer complete and correct copies of all federal income Tax Returns, examination reports and statements of deficiencies assessed

-12-

against or agreed to by the Company or any Subsidiary for all periods from and after December 31, 2001. The federal income Tax Returns of the Company and each Subsidiary have been audited by the Internal Revenue Service or are closed by the applicable statute of limitations for all taxable years through December 31, 2001, but such taxable years may be subject to Internal Revenue Service examination to the extent that net operating loss carry-forwards are utilized (or reduce taxable income) in future tax years. The Company has delivered or made available to the Buyer complete and accurate copies of all other Tax Returns of the Company and the Subsidiaries together with all related examination reports and statements of deficiency for all periods from and after December 31, 2001. No examination or audit of any Tax Return of the Company or any Subsidiary by any Governmental Entity is currently in progress or to the knowledge of the Company, threatened or contemplated. Neither the Company nor any Subsidiary has been informed in writing by any jurisdiction that the jurisdiction believes that the Company or Subsidiary was required to file any Tax Return that was not filed. Neither the Company nor any Subsidiary has waived any statute of limitations with respect to Taxes or agreed to an extension of time with respect to a Tax assessment or deficiency.

- (c) Neither the Company nor any Subsidiary: (i) is a "consenting corporation" within the meaning of Section 341(f) of the Code, and none of the assets of the Company or the Subsidiaries are subject to an election under Section 341(f) of the Code; (ii) has been a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code during the applicable period specified in Section 897(c)(l)(A)(ii) of the Code; (iii) except as set forth in Section 2.9(c) of the Disclosure Schedule, has made any payments, is obligated to make any payments, or is a party to any agreement that could obligate it to make any payments that may be treated as an "excess parachute payment" under Section 280G of the Code; (iv) is or will be obligated to make any gross-up payments in connection with the payments described in clause (iii) above; (v) has any actual or potential liability for any Taxes of any person (other than the Company and its Subsidiaries) under Treasury Regulation Section 1.1502-6 (or any similar provision of federal, state, local, or foreign law), or as a transferee or successor, by contract, or otherwise; or (vi) is or has been required to make a basis reduction pursuant to Treasury Regulation Section 1.1502-20(b) or Treasury Regulation Section 1.337(d)-2(b).
- (d) None of the assets of the Company or any Subsidiary: (i) is "tax-exempt use property" within the meaning of Section 168(h) of the Code; or (ii) directly or indirectly secures any debt the interest on which is tax exempt under Section 103(a) of the Code.
- (e) Neither the Company nor any Subsidiary has undergone a change in its method of accounting resulting in an adjustment to its taxable income pursuant to Section 481 of the Code.
- (f) Except as set forth in Section 2.9(f) of the Disclosure Schedule or as a result of the transactions contemplated by this Agreement, since August 30, 2000, the Company has not had an "ownership change" within the meaning of Section 382(g) of the Code.

2.10 Assets.

(a) The Company or the applicable Subsidiary is the true and lawful owner, and has good title to, all of the assets (tangible or intangible) reflected on the Most Recent

-13-

Balance Sheet or acquired by the Company or any Subsidiary since the Most Recent Balance Sheet Date, free and clear of all Security Interests. Each of the Company and the Subsidiaries owns or leases all tangible assets sufficient for the conduct of its businesses as presently conducted. Each such tangible asset is free from material defects, has been maintained in accordance with normal industry practice, is in good operating condition and repair (subject to normal wear and tear) and is suitable for the purposes for which it presently is used.

- (b) Section 2.10(b) of the Disclosure Schedule lists individually, as of the Most Recent Balance Sheet Date, each fixed asset (within the meaning of GAAP) of the Company or the Subsidiaries with a net book value in excess of \$10,000, indicating the cost, accumulated book depreciation (if any) and the net book value of each such fixed asset as of the Most Recent Balance Sheet Date.
- (c) Each item of equipment, motor vehicle and other asset that the Company or a Subsidiary has possession of pursuant to a lease agreement or other contractual arrangement is in such condition that, upon its return to its lessor or owner and payment of all amounts due (other than with respect to the condition of such asset) under the applicable lease or contract, the obligations of the Company or such Subsidiary to such lessor or owner will have been discharged in full in all material respects.
 - 2.11 Owned Real Property. Neither the Company nor any Subsidiary owns any Owned Real Property.
- 2.12 <u>Real Property Leases</u>. Section 2.12 of the Disclosure Schedule lists all Leases and lists the term of such Lease, whether such Lease contains any extension and expansion options, and the annual rent payable thereunder. The Company has delivered or, if not delivered, made available to the Buyer complete and accurate copies of the Leases. With respect to each Lease:
- (a) such Lease is in full force and effect and is legal, valid, binding, and enforceable against the Company and, to the knowledge of the Company, the other parties thereto, except to the extent that such enforceability may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the enforcement of creditors' rights generally and to general equitable principles;
- (b) except as set forth in Section 2.12(b) of the Disclosure Schedule, such Lease will continue to be legal, valid, binding, enforceable and in full force and effect immediately following the Closing in accordance with the terms thereof as in effect immediately prior to the Closing, except to the extent that such enforceability may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the enforcement of creditors' rights generally and to general equitable principles;
- (c) neither the Company nor any Subsidiary nor, to the knowledge of the Company, any other party, is in breach or violation of, or default under, any such Lease, and no event has occurred, is pending or, to the knowledge of the Company, is threatened, which, after the giving of notice, with lapse of time, or otherwise, would constitute a breach or default by the Company or any Subsidiary or, to the knowledge of the Company, any other party under such

-14-

Lease, other than any such breaches, violations or defaults that would not have, individually or in the aggregate, a Company Material Adverse Effect;

- (d) there are no disputes, oral agreements or forbearance programs in effect as to such Lease;
- (e) except as set forth in Section 2.12(e) of the Disclosure Schedule, neither the Company nor any Subsidiary has assigned, transferred, conveyed, mortgaged, deeded in trust or encumbered any interest in the leasehold or subleasehold;
- (f) to the knowledge of the Company, all facilities leased or subleased thereunder are supplied with utilities and other services adequate for the operation of said facilities; and
- (g) the Company has no knowledge of any Security Interest, easement, covenant or other restriction applicable to the real property subject to such lease which would reasonably be expected to materially impair the current uses or the occupancy by the Company or a Subsidiary of the property subject thereto.

2.13 Intellectual Property.

- (a) Section 2.13(a) of the Disclosure Schedule lists (i) each patent, patent application, copyright registration or application therefor, mask work registration or application therefor, and trademark, service mark and domain name registration or application therefor of the Company or any Subsidiary and (ii) each Customer Deliverable of the Company or any Subsidiary.
- (b) To the knowledge of the Company, each of the Company and the Subsidiaries owns or has the right to use all Patents and Trademarks included in the Company Intellectual Property necessary (i) to use, produce, market and distribute the Customer Deliverables and (ii) to operate the Internal Systems as currently operated. Each of the Company and the Subsidiaries owns or has the right to use all Intellectual Property, other than the Patents and Trademarks included in the Company Intellectual Property, necessary (i) to use, produce, market and distribute the Customer Deliverables and (ii) to operate the Internal Systems as currently operated. Except as set forth in Section 2.13(b) of the Disclosure Schedule, each item of Company Intellectual Property will be owned or available for use by the Surviving Corporation immediately following the Closing on substantially similar terms and conditions as it was immediately prior to the Closing. The Company or the appropriate Subsidiary has taken commercially reasonable measures to protect the proprietary nature of each item of Company Intellectual Property, and to maintain in confidence all trade secrets and confidential information, that it owns or uses. No other person or entity has any rights to any of the Company Intellectual Property owned by the Company or the Subsidiaries (except pursuant to agreements or licenses specified in Section 2.13(d) of the Disclosure Schedule or agreements with academic institutions in the Ordinary Course of Business and agreements with customers set forth in Section 2.25 of the Disclosure Schedule), and, to the knowledge of the Company, no other person or entity is infringing, violating or misappropriating any of the Company Intellectual Property.

-15-

- (c) To the knowledge of the Company, none of the Customer Deliverables, or the marketing, distribution, provision or use thereof, infringes or violates, or constitutes a misappropriation of, any Patent or Trademark rights of any person or entity. None of the Customer Deliverables, or the marketing, distribution, provision or use thereof, infringes or violates, or constitutes a misappropriation of, any Intellectual Property rights (other than Patent or Trademark rights) of any person or entity. To the knowledge of the Company, none of the Internal Systems, or the use thereof, infringes or violates, or constitutes a misappropriation of, any Patent or Trademark rights of any person or entity; provided that for the purposes of this sentence, "knowledge" shall mean the actual knowledge of Carol Vallone, John Giordano, Barbara Ross, Peter Segall, Christopher Vento, Karen Gage and Robert Bean, without any inquiry. None of the Internal Systems, or the use thereof, infringes or violates, or constitutes a misappropriation of, any Intellectual Property rights (other than Patent or Trademark rights) of any person or entity. Section 2.13(c) of the Disclosure Schedule lists any complaint, claim or notice, or written threat thereof, received by the Company or any Subsidiary alleging any such infringement, violation or misappropriation; and the Company has delivered or, if not delivered, made available to the Buyer complete and accurate copies of all written documentation in the Possession relating to claims or disputes known to the Company concerning any Company Intellectual Property.
- (d) Section 2.13(d) of the Disclosure Schedule identifies each person or entity to which the Company or a Subsidiary has licensed, distributed or otherwise granted any rights to any third party with respect to, any Company Intellectual Property owned by or licensed to the Company or a Subsidiary, excluding academic institutions with agreements with the Company entered into in the Ordinary Course of Business and customers set forth on Section 2.25 of the Disclosure Schedule.
- (e) Section 2.13(e) of the Disclosure Schedule identifies each item of Company Intellectual Property which the Company has licensed from a party other than the Company or a Subsidiary, and the license or agreement pursuant to which the Company or a Subsidiary uses it (excluding off-the-shelf or freely or commercially downloadable software programs).
- (f) Section 2.13(f) of the Disclosure Schedule lists all Open Source Materials that the Company or a Subsidiary has distributed, the Customer Deliverables with which it has been distributed and whether the Open Source Materials have been modified. Except as set forth in the Disclosure Schedule, neither the Company nor any Subsidiary has (i) incorporated any Open Source Materials into, or combined Open Source Materials with, any Customer Deliverables, (ii) distributed Open Source Materials in connection with any Customer Deliverables, or (iii) used Open Source Materials that (with respect to either clause (i), (ii) or (iii) above) (A) create, or purport to create, obligations for the Company or a Subsidiary with respect to software developed or distributed by the Company or a Subsidiary or (B) grant, or purport to grant, to any third party any rights or immunities under intellectual property rights. Without limiting the generality of the foregoing, neither the Company nor any Subsidiary has distributed any Open Source Materials that require, as a condition of distribution of such Open Source Materials, that other software incorporated into, derived from or distributed with such Open

-16-

Source Materials be (1) disclosed or distributed in source code form, (2) licensed for the purpose of making derivative works, or (3) redistributable at no charge.

- (g) Neither the Company nor any Subsidiary has disclosed the source code for the Software or other confidential information constituting, embodied in or pertaining to the Software to any person or entity, except pursuant to the agreements listed in Sections 2.13(f) and 2.13(g) of the Disclosure Schedule, and the Company has taken commercially reasonable measures to prevent disclosure of such source code.
- (h) All of the Intellectual Property (including all Patents and Trademarks listed on Section 2.13(a) of the Disclosure Schedule, but excluding all other Patents and Trademarks) incorporated in the Customer Deliverables has been (i) created by employees of the Company or a Subsidiary within the scope of their employment by the Company or a Subsidiary or by independent contractors of the Company or a Subsidiary who have executed agreements expressly assigning all right, title and interest in such Intellectual Property to the Company or a Subsidiary, or (ii) licensed pursuant to an agreement set forth in Section 2.13(e) of the Disclosure Schedule. No portion of such Intellectual Property was jointly developed with any third party.
 - (i) The Customer Deliverables conform in all material respects to the written documentation and specifications therefor.
 - 2.14 Inventory. Neither the Company nor any Subsidiary had any tangible inventory as of the Most Recent Balance Sheet Date.

2.15 Contracts.

- (a) Section 2.15 of the Disclosure Schedule lists the following agreements (written or oral) to which the Company or any Subsidiary is a party as of the date of this Agreement:
- (i) any agreement (or group of related agreements) for the lease of personal property from or to third parties providing for lease payments in excess of \$25,000 per annum;
- (ii) any agreement (or group of related agreements), excluding agreements in the Ordinary Course of Business with the customers of the Company or any Subsidiary set forth on Section 2.25 of the Disclosure Schedule, for the purchase or sale of products or for the furnishing or receipt of services pursuant to which (A) the Company has paid, or may be obligated to pay, more than the sum of \$20,000 and has not yet received the products or services for which such payment has been or will be made, (B) the Company has received, or may be entitled to receive, more than the sum of \$20,000 and has not yet delivered the products or services for which it has received or will receive such payment, or (C) in which the Company or any Subsidiary has granted manufacturing rights, "most favored nation" pricing provisions or marketing or distribution rights relating to any products or territory or has agreed to purchase a minimum quantity of goods or services or has agreed to purchase goods or services exclusively from a certain party;

-17-

- (iii) any agreement concerning the establishment or operation of a partnership, joint venture or limited liability company;
- (iv) any agreement (or group of related agreements) under which it has created, incurred, assumed or guaranteed (or may create, incur, assume or guarantee) indebtedness (including capitalized lease obligations) involving more than \$50,000 or under which it has imposed (or may impose) a Security Interest on any of its assets, tangible or intangible;
- (v) any agreement for the disposition of any significant portion of the assets or business of the Company or any Subsidiary (other than sales of products in the Ordinary Course of Business) or any agreement for the acquisition of the assets or business of any other entity (other than purchases of inventory or components in the Ordinary Course of Business);
- (vi) any agreement concerning confidentiality or noncompetition other than agreements with customers, employees or consultants in the Ordinary Course of Business;
 - (vii) any employment or consulting agreement;
- (viii) any agreement presently in effect to which any current or former officer, director or stockholder of the Company or an Affiliate thereof is a party;
- (ix) any agreement, license, franchise, permit, or authorization presently in effect that may be, by its terms, terminated, materially impaired, or materially adversely affected by reason of the execution of this Agreement and all other agreements contemplated hereby, the consummation of the Merger, or the consummation of the transactions contemplated hereby;
- (x) any agreement which contains any provisions requiring the Company or any Subsidiary to indemnify any other party (excluding indemnities contained in agreements for the purchase, sale or license of Customer Deliverables entered into in the Ordinary Course of Business) pursuant to which the amount of indemnification exceeds \$300,000;
- (xi) any other agreement (or group of related agreements) involving more than \$10,000 and not entered into in the Ordinary Course of Business;
- (xii) any agreement pursuant to which the Company has granted, or may be obligated to grant in the future, to any party a source code license or option or other right to use or acquire any source code of the Company, including any agreements which provide for source code escrow arrangements; and
- (xiii) any other non-customer agreement under which the consequences of a default or termination would reasonably be expected to have a Company Material Adverse Effect.
- (b) The Company has delivered or made available to the Buyer a complete and accurate copy of each agreement listed in Section 2.13 or Section 2.15 of the Disclosure

-18-

Schedule and each agreement with those customers set forth on Section 2.25 of the Disclosure Schedule. With respect to each such agreement: (i) the agreement is in full force and effect, is legal, valid, binding and enforceable against the Company and, to the Company's knowledge, against each other party thereto, except to the extent that such enforceability may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the enforcement of creditors' rights generally and to general equitable principles; (ii) the agreement will continue to be legal, valid, binding and enforceable and in full force and effect immediately following the Closing in accordance with the terms thereof as in effect immediately prior to the Closing; and (iii) neither the Company nor any Subsidiary nor, to the knowledge of the Company, any other party, is in breach or violation of, or default under, any such agreement, and, to the knowledge of the Company, no event has occurred, is pending or, is threatened, which, after the giving of notice, with lapse of time, or otherwise, would constitute a breach or default by the Company or any Subsidiary or, to the knowledge of the Company, any other party under such agreement, other than such breaches, violations or defaults that would not have, individually or in the aggregate, a Company Material Adverse Effect.

- 2.16 Accounts Receivable. All accounts receivable of the Company and the Subsidiaries reflected on the Most Recent Balance Sheet are valid receivables subject to no setoffs or counterclaims known to the Company. A complete and accurate list of the accounts receivable reflected on the Most Recent Balance Sheet, showing the aging thereof, is included in Section 2.16 of the Disclosure Schedule. All accounts receivable of the Company and the Subsidiaries that have arisen since the Most Recent Balance Sheet Date are valid receivables subject to no setoffs or counterclaims known to the Company. Neither the Company nor any Subsidiary has received any written notice from an account debtor stating that any account receivable is subject to any contest, setoff or counterclaim by such account debtor.
 - 2.17 Powers of Attorney. There are no outstanding powers of attorney executed on behalf of the Company or any Subsidiary.
- 2.18 <u>Insurance</u>. Section 2.18 of the Disclosure Schedule lists each insurance policy (including fire, theft, casualty, comprehensive general liability, workers compensation, business interruption, environmental, product liability and automobile insurance policies and bond and surety arrangements) to which the Company or any Subsidiary is a party, all of which are in full force and effect. There is no material claim pending under any such policy as to which coverage has been questioned, denied or disputed by the underwriter of such policy. All premiums due and payable under all such policies have been paid. To the knowledge of the Company, neither the Company nor any Subsidiary may be liable for retroactive premiums or similar payments, and the Company and the Subsidiaries are otherwise in compliance in all material respects with the terms of such policies. The Company has no knowledge of any threatened termination of, or premium increase with respect to, any such policy, other than premium increases in the Ordinary Course of Business. Each such policy will continue to be enforceable and in full force and effect immediately following the Closing in accordance with the terms thereof as in effect immediately prior to the Closing.
- 2.19 <u>Litigation</u>. As of the date of this Agreement, there is no Legal Proceeding which is pending or has been threatened in writing, or, to the knowledge of the Company, has been otherwise threatened against the Company or any Subsidiary which (a) seeks either money

-19-

damages or equitable relief or (b) in any manner challenges or seeks to prevent, enjoin, alter or delay the transactions contemplated by this Agreement. There are no judgments, orders or decrees outstanding against the Company or any Subsidiary.

2.20 <u>Warranties</u>. No product or service manufactured, sold, leased, licensed or delivered by the Company or any Subsidiary is subject to any guaranty, warranty, right of return, right of credit or other indemnity other than (i) the applicable standard terms and conditions of sale or license, substantially in the forms set forth in Section 2.20 of the Disclosure Schedule and (ii) manufacturers' warranties for which neither the Company nor any Subsidiary has any liability. Since January 1, 2002, the Company and the Subsidiaries have not incurred any expenses in fulfilling their obligations under their guaranty, warranty, right of return and indemnity provisions.

2.21 Employees.

- (a) Section 2.21 of the Disclosure Schedule contains a list, as of the date of this Agreement, of all employees (such term meaning permanent and temporary, full-time and part-time employees) of the Company and each Subsidiary, along with the position and the annual rate of compensation of each such person, and the wage rates for non-salaried employees (by classification). Except as set forth in Section 2.21(a) of the Disclosure Schedule, each current employee of the Company or any Subsidiary, and each person who has been employed by the Company or any Subsidiary within the last two years, has entered into a confidentiality and assignment of inventions agreement with the Company or such Subsidiary, a copy or form of which has previously been delivered or, if not delivered, made available to the Buyer. Section 2.21 of the Disclosure Schedule contains a list of all employees of the Company or any Subsidiary who are a party to a non-competition agreement with the Company or any Subsidiary; copies of such agreements have previously been delivered or, if not delivered, made available to the Buyer. Section 2.21 of the Disclosure Schedule contains a list of all employees of the Company or any Subsidiary who are not citizens of the United States. All of the Company's employees who work in the United States are legally authorized to do so. The Company has no plans to terminate the employment of any key employee of the Company or any Subsidiary. No employee has notified the Company in writing that such employee intends to resign or retire as a result of the transactions contemplated by this Agreement or otherwise prior to six months after the Closing.
- (b) Neither the Company nor any Subsidiary is a party to or bound by any collective bargaining agreement, nor has any of them experienced any strikes, grievances, claims of unfair labor practices or other collective bargaining disputes. The Company has no knowledge of any organizational effort made or threatened, either currently or within the past two years, by or on behalf of any labor union with respect to employees of the Company or any Subsidiary. Except as provided in Section 2.21(b) of the Disclosure Schedule, the Company does not have any contract of employment or other employment severance or similar agreement with any of its employees or any established policy or practice relating thereto, and all of its employees are employees-at-will.
- (c) Any persons engaged by the Company as independent contractors, rather than employees, have been properly classified as such and have been so engaged in accordance

-20-

with all applicable federal, foreign, state or local laws, except to the extent that a failure to have been so classified or engaged would, individually or in the aggregate, have a Company Material Adverse Effect. Hours worked by and payments made to employees of the Company have not been in violation of the Fair Labor Standards Act or any other applicable federal, foreign, state or local laws dealing with such matters. Except as set forth in Section 2.21(c) of the Disclosure Schedule, the Company is not and never has been engaged in any dispute or litigation with an employee or former employee regarding matters pursuant to intellectual property or assignment of inventions.

(d) The Company has complied in all material respects with all applicable federal, foreign, state or local laws, ordinances, rules and regulations and requirements relating to the employment of labor, including but not limited to the provisions thereof relating to wages, hours, collective bargaining and ensuring equality of opportunity for employment and advancement of minorities, women and other individuals protected by applicable law. There are no claims pending, threatened in writing, or, to the knowledge of the Company, otherwise threatened to be brought, in any court or administrative agency by any former or current Company employees for compensation, pending severance benefits, vacation time, vacation pay or pension benefits, or any other claim pending, threatened in writing, or, to the knowledge of the Company, otherwise threatened in any court or administrative agency from any current or former employee or any other Person arising out of the Company's status as employer, whether in the form of claims for employment discrimination, harassment, unfair labor practices, grievances, wrongful discharge, or otherwise. The Company has not undergone any wage and hour investigations by any Governmental Entity in the last two years.

2.22 Employee Benefits.

- (a) Section 2.22(a) of the Disclosure Schedule contains a complete and accurate list of all Company Plans. Complete and accurate copies of (i) all Company Plans which have been reduced to writing, (ii) written summaries of all unwritten Company Plans, (iii) all related trust agreements, insurance contracts and summary plan descriptions, and (iv) all annual reports filed on IRS Form 5500, 5500C or 5500R and (for all funded plans) all plan financial statements for the last three plan years for each Company Plan, have been delivered or, if not delivered, made available to the Buyer.
- (b) Each Company Plan has been administered in all material respects in accordance with its terms and each of the Company, the Subsidiaries and the ERISA Affiliates has in all material respects met its obligations with respect to each Company Plan and has made all required contributions thereto. The Company, each Subsidiary, each ERISA Affiliate and each Company Plan are in compliance in all material respects with the currently applicable provisions of ERISA and the Code and the regulations thereunder (including Section 4980 B of the Code, Subtitle K, Chapter 100 of the Code and Sections 601 through 608 and Section 701 et seq. of ERISA). All filings and reports as to each Company Plan required to have been submitted to the Internal Revenue Service or to the United States Department of Labor have been duly submitted. No Company Plan has assets that include securities issued by the Company or any ERISA Affiliate.

-21-

- (c) There are no Legal Proceedings (except claims for benefits payable in the normal operation of the Company Plans and proceedings with respect to qualified domestic relations orders) against or involving any Company Plan or asserting any rights or claims to benefits under any Company Plan that could give rise to any material liability.
- (d) The Company Plan that is intended to be qualified under Section 401(a) of the Code is maintained pursuant to a prototype plan document that has received a favorable opinion letter from the Internal Revenue Service; such opinion letter has not been revoked, and no act or omission has occurred that would adversely affect the qualification of such Company Plan or materially increase its cost. Each Company Plan that is required to satisfy Section 401(k)(3) or Section 401(m)(2) of the Code has been tested for compliance with, and satisfies the requirements of Section 401(k)(3) and Section 401(m)(2) of the Code for each plan year ending prior to the Closing Date.
- (e) Neither the Company, any Subsidiary, nor any ERISA Affiliate has ever maintained an Employee Benefit Plan subject to Section 412 of the Code or Title IV of ERISA.
- (f) At no time has the Company, any Subsidiary or any ERISA Affiliate been obligated to contribute to any "multiemployer plan" (as defined in Section 4001(a)(3) of ERISA).
- (g) There are no unfunded obligations under any Company Plan providing benefits after termination of employment to any employee of the Company or any Subsidiary (or to any beneficiary of any such employee), including but not limited to retiree health coverage and deferred compensation, but excluding continuation of health coverage required to be continued under Section 4980B of the Code or other applicable law and insurance conversion privileges under state law. The assets of each Company Plan that is funded are reported at their fair market value on the books and records of such Company Plan.
- (h) No act or omission has occurred and no condition exists with respect to any Company Plan that would subject the Company, any Subsidiary or any ERISA Affiliate to (i) any material fine, penalty, tax or liability of any kind imposed under ERISA or the Code or (ii) any contractual indemnification or contribution obligation protecting any fiduciary, insurer or service provider with respect to any Company Plan.
- (i) No Company Plan is funded by, associated with or related to a "voluntary employee's beneficiary association" within the meaning of Section 501(c)(9) of the Code.
- (j) Each Company Plan is amendable and terminable unilaterally by the Company at any time without liability or expense to the Company or such Company Plan as a result thereof (other than for benefits accrued through the date of termination or amendment and reasonable administrative expenses related thereto) and no Company Plan, plan documentation or agreement, summary plan description or other written communication distributed generally to employees by its terms prohibits the Company from amending or terminating any such Company Plan.
- (k) Section 2.22(k) of the Disclosure Schedule discloses each: (i) agreement with any stockholder, director, executive officer or other key employee of the Company or any

-22-

Subsidiary (A) the benefits of which are contingent, or the terms of which are altered, upon the occurrence of a transaction involving the Company or any Subsidiary of the nature of any of the transactions contemplated by this Agreement, (B) providing any term of employment or compensation guarantee or (C) providing severance benefits or other benefits after the termination of employment of such director, executive officer or key employee; (ii) agreement, plan or arrangement under which any person may receive payments from the Company or any Subsidiary that may be subject to the tax imposed by Section 4999 of the Code or included in the determination of such person's "parachute payment" under Section 280G of the Code; and (iii) agreement or plan binding the Company or any Subsidiary, including any stock option plan, stock appreciation right plan, restricted stock plan, stock purchase plan, severance benefit plan or Company Plan, any of the benefits of which will be increased, or the vesting of the benefits of which will be calculated on the basis of any of the transactions contemplated by this Agreement or the value of any of the benefits of which will be calculated on the basis of any of the transactions contemplated by this Agreement.

(1) Section 2.22(1) of the Disclosure Schedule sets forth the policy of the Company and any Subsidiary with respect to accrued vacation, accrued sick time and earned time off, as well as the amount of such liabilities as of the Most Recent Balance Sheet Date.

2.23 Environmental Matters.

- (a) Each of the Company and the Subsidiaries has complied with all applicable Environmental Laws other than any failures to comply that would not have, individually or in the aggregate, a Company Material Adverse Effect. There is no pending or, to the knowledge of the Company, threatened civil or criminal litigation, written notice of violation, formal administrative proceeding, or investigation, inquiry or information request by any Governmental Entity, relating to any Environmental Law involving the Company or any Subsidiary.
- (b) Neither the Company nor any Subsidiary has any liabilities or obligations arising from the release of any Materials of Environmental Concern into the environment.
- (c) Neither the Company nor any Subsidiary is a party to or bound by any court order, administrative order, consent order or other agreement between the Company and any Governmental Entity entered into in connection with any legal obligation or liability arising under any Environmental Law.
- (d) Set forth in Section 2.23(d) of the Disclosure Schedule is a list of all documents (whether in hard copy or electronic form) that contain any environmental reports, investigations and audits relating to premises currently or previously owned or operated by the Company or a Subsidiary (whether conducted by or on behalf of the Company or a Subsidiary or a third party, and whether done at the initiative of the Company or a Subsidiary or directed by a Governmental Entity or other third party) which the Company has possession of or access to. A complete and accurate copy of each such document has been provided to the Buyer.

-23-

- (e) The Company has no knowledge of any material environmental liability of any solid or hazardous waste transporter or treatment, storage or disposal facility that has been used by the Company or any Subsidiary.
- 2.24 <u>Legal Compliance</u>. Each of the Company and the Subsidiaries is currently conducting, and has at all times conducted, its business in compliance with each applicable law (including rules and regulations thereunder) of any federal, state, local or foreign government, or any Governmental Entity, except for any violations or failures to comply that, individually or in the aggregate, have not had and would not reasonably be expected to have a Company Material Adverse Effect. Neither the Company nor any Subsidiary has received any written notice, written communication or, to the knowledge of the Company, other notice or communication from any Governmental Entity alleging noncompliance with any applicable law, rule or regulation.
- 2.25 <u>Customers and Suppliers</u>. Section 2.25 of the Disclosure Schedule sets forth a list of (a) each customer that accounted for more than \$10,000 in revenues of the Company during either the last full fiscal year or the interim period through the Most Recent Balance Sheet Date and (b) each supplier that is the sole supplier of any significant product or service to the Company or a Subsidiary. No such customer or supplier has given the Company or any Subsidiary written notice of non-renewal during the twelve months prior to the date hereof. No purchase order or commitment of the Company or any Subsidiary is in excess of normal requirements, nor are prices provided therein in excess of current market prices for the products or services to be provided thereunder.
- 2.26 <u>Permits</u>. Section 2.26 of the Disclosure Schedule sets forth a list of all material Permits issued to or held by the Company or any Subsidiary. Such listed Permits are the only Permits that are required for the Company and the Subsidiaries to conduct their respective businesses as presently conducted or as proposed to be conducted, other than such Permits the failure to possess which would not have, individually or in the aggregate, a Company Material Adverse Effect. Each such Permit is in full force and effect; the Company or the applicable Subsidiary is in compliance with the terms of each such Permit; and, to the knowledge of the Company, no suspension or cancellation of such Permit is threatened and there is no basis for believing that such Permit will not be renewable upon expiration. Each such Permit will continue in full force and effect immediately following the Closing.
- 2.27 Certain Business Relationships With Affiliates. No Affiliate of the Company or of any Subsidiary (a) owns any property or right, tangible or intangible, which is used in the business of the Company or any Subsidiary except as set forth in Section 2.27(a) of the Disclosure Schedule, (b) to the knowledge of the Company, has any claim or cause of action against the Company or any Subsidiary, or (c) owes any money to, or is owed any money by, the Company or any Subsidiary (other than compensation and benefits owed by the Company to officers of the Company in the Ordinary Course of Business or pursuant to agreements set forth in Section 2.15 of the Disclosure Schedule). Section 2.27 of the Disclosure Schedule describes any commercial transactions or relationships between the Company or a Subsidiary and any Affiliate thereof, which occurred or have existed since the beginning of the time period covered by the Financial Statements.

-24-

- 2.28 <u>Financial Advisory Fees</u>. Other than as set forth on Section 2.28 of the Disclosure Schedule, neither the Company nor any Subsidiary has any liability or obligation to pay any fees or commissions to any financial advisor, broker, finder or agent with respect to the transactions contemplated by this Agreement.
- 2.29 <u>Books and Records</u>. The minute books and other similar records of the Company and each Subsidiary contain complete and accurate records of all actions taken at any meetings of the Company's or such Subsidiary's stockholders, Board of Directors or any committee thereof and of all written consents executed in lieu of the holding of any such meeting. The books and records of the Company and each Subsidiary accurately reflect in all material respects the assets, liabilities, business, financial condition and results of operations of the Company or such Subsidiary and have been maintained in accordance with good business and bookkeeping practices. Section 2.29 of the Disclosure Schedule contains a list of all bank accounts and safe deposit boxes of the Company and the Subsidiaries and the names of persons having signature authority with respect thereto or access thereto.
- 2.30 <u>Non-Compete Agreements</u>. Except as set forth on Section 2.30 of the Disclosure Schedule, neither the Company nor any Subsidiary is party to any agreement (written or oral) in the nature of a non-compete or exclusivity agreement or that otherwise limits or restricts the ability of the Company, or would after the Closing limit or restrict the ability of the Surviving Corporation or the Buyer, to compete or otherwise conduct its business in any manner or place.

2.31 Government Contracts.

- (a) Neither the Company nor any Subsidiary has been suspended or debarred from bidding on contracts or subcontracts with any Governmental Entity; to the knowledge of the Company, no such suspension or debarment has been threatened or initiated; and the consummation of the transactions contemplated by this Agreement will not result in any such suspension or debarment of the Company, any Subsidiary or the Buyer (assuming that no such suspension or debarment will result solely from the identity of the Buyer). Neither the Company nor any Subsidiary has been or is now being audited or investigated by the United States Government Accounting Office, the United States Department of Defense or any of its agencies, the Defense Contract Audit Agency, the contracting or auditing function of any Governmental Entity with which it is contracting, the United States Department of Justice, the Inspector General of the United States Governmental Entity, or any prime contractor with a Governmental Entity; nor, to the knowledge of the Company, has any such audit or investigation been threatened. To the knowledge of the Company, there is no valid basis for (i) the suspension or debarment of the Company or any Subsidiary from bidding on contracts or subcontracts with any Governmental Entity or (ii) any claim (including any claim for return of funds to the Government) pursuant to an audit or investigation by any of the entities named in the foregoing sentence. The Company has no agreements, contracts or commitments, which require it to obtain or maintain a security clearance with any Governmental Entity.
- (b) Other than as set forth on Section 2.31 of the Disclosure Schedule, neither the Company nor any Subsidiary is a party to any contract with a U.S. federal Governmental Entity. To the knowledge of the Company, no basis exists for any of the following with respect to any of its contracts or subcontracts with any Governmental Entity: (i) a Termination for

-25-

Default (as provided in 48 C.F.R. Ch.1 §52.249-8, 52.249-9 or similar sections), (ii) a Termination for Convenience (as provided in 48 C.F.R. Ch.1 §52.241-1, 52.249-2 or similar sections), or (iii) a Stop Work Order (as provided in 48 C.F.R. Ch.1 §52.212-13 or similar sections). The Company has no reason to believe that funding may not be provided under any presently existing contract or subcontract with any Governmental Entity in the upcoming federal fiscal year.

2.32 <u>Disclosure</u>. No representation or warranty by the Company contained in this Agreement, and no statement contained in the Disclosure Schedule or any other document, certificate or other instrument delivered or to be delivered by or on behalf of the Company pursuant to this Agreement, contains or will contain any untrue statement of a material fact or omits or will omit to state any material fact necessary, in light of the circumstances under which it was or will be made, in order to make the statements herein or therein not misleading.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE BUYER AND THE TRANSITORY SUBSIDIARY

Each of the Buyer and the Transitory Subsidiary represents and warrants to the Company that the statements contained in this Article III are true and correct as of the date of this Agreement and will be true and correct as of the Closing as though made as of the Closing, except to the extent such representations and warranties are specifically made as of a particular date (in which case such representations and warranties will be true and correct as of such date):

- 3.1 Organization, Qualification and Corporate Power. Each of the Buyer and the Transitory Subsidiary is a corporation duly organized, validly existing and in good standing under the laws of the state of its incorporation. The Buyer is duly qualified to conduct business and is in good standing under the laws of each jurisdiction in which the nature of its businesses or the ownership or leasing of its properties requires such qualification, except where the failure to be so qualified or in good standing would not have a Buyer Material Adverse Effect. The Buyer has all requisite corporate power and authority to carry on the businesses in which it is engaged and to own and use the properties owned and used by it. The Buyer has furnished or made available to the Company complete and accurate copies of its certificate of incorporation and by-laws.
- 3.2 <u>Authorization of Transaction</u>. Each of the Buyer and the Transitory Subsidiary has all requisite power and authority to execute and deliver this Agreement and (in the case of the Buyer) the Escrow Agreement and to perform its obligations hereunder and thereunder. The execution and delivery by the Buyer and the Transitory Subsidiary of this Agreement and (in the case of the Buyer) the Escrow Agreement and the consummation by the Buyer and the Transitory Subsidiary of the transactions contemplated hereby and thereby have been duly and validly authorized by all necessary corporate action on the part of the Buyer and Transitory Subsidiary, respectively. This Agreement has been duly and validly executed and delivered by the Buyer and the Transitory Subsidiary and constitutes a valid and binding obligation of the Buyer and the Transitory Subsidiary, enforceable against them in accordance with its terms, except to the extent that such enforceability may be subject to applicable bankruptcy, insolvency,

-26-

reorganization, moratorium or similar laws affecting the enforcement of creditors' rights generally and to general equitable principles.

- 3.3 Noncontravention. Subject to compliance with the applicable requirements of the Securities Act and any applicable state securities laws, the Exchange Act, the HSR Act as well as premerger notification laws and regulations of other Governmental Entities, if applicable, and the filing of the Certificate of Merger as required by the Delaware General Corporation Law, neither the execution and delivery by the Buyer or the Transitory Subsidiary of this Agreement or (in the case of the Buyer) the Escrow Agreement, nor the consummation by the Buyer or the Transitory Subsidiary of the transactions contemplated hereby or thereby, will (a) conflict with or violate any provision of the charter or by-laws of the Buyer or the Transitory Subsidiary, (b) require on the part of the Buyer or the Transitory Subsidiary any filing with, or permit, authorization, consent or approval of, any Governmental Entity, (c) conflict with, result in breach of, constitute (with or without due notice or lapse of time or both) a default under, result in the acceleration of obligations under, create in any party any right to terminate, modify or cancel, or require any notice, consent or waiver under, any contract or instrument to which the Buyer or the Transitory Subsidiary is a party or by which either is bound or to which any of their assets are subject, except for (i) any conflict, breach, default, acceleration, termination, modification or cancellation which would not adversely affect the consummation of the transactions contemplated hereby or (ii) any notice, consent or waiver the absence of which would not adversely affect the consummation of the transactions contemplated hereby, or (d) violate any order, writ, injunction, decree, statute, rule or regulation applicable to the Buyer or the Transitory Subsidiary or any of their properties or assets, except in the case of clause (c) above as would not have, individually or in the aggregate, a Buyer Material Adverse Effect.
- 3.4 <u>Buyer Financial Condition</u>. The balance sheet of the Buyer as of June 30, 2005 as filed with the Securities and Exchange Commission in the Buyer's Form 10-Q for the quarter then ended (the "<u>June 30, 2005 Balance Sheet</u>") (i) complied as to form in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto when filed, (ii) was prepared in accordance with GAAP (except as may be indicated therein or in the notes thereto, and except as permitted by Form 10-Q under the Exchange Act), and (iii) fairly presents the consolidated financial condition of the Buyer as of June 30, 2005.
- 3.5 <u>Absence of Material Adverse Change</u>. Since June 30, 2005, there has not occurred any materially adverse change in the financial condition of the Buyer as reflected on the June 30, 2005 Balance Sheet.
- 3.6 <u>Litigation</u>. Except as disclosed in the Buyer Reports, as of the date of this Agreement, there is no Legal Proceeding which is pending or, to the Buyer's knowledge, threatened against the Buyer or any subsidiary of the Buyer which, if determined adversely to the Buyer or such subsidiary, could have, individually or in the aggregate, a Buyer Material Adverse Effect or which in any manner challenges or seeks to prevent, enjoin, alter or delay the transactions contemplated by this Agreement.

-27-

3.7 <u>Interim Operations of the Transitory Subsidiary</u>. The Transitory Subsidiary was formed solely for the purpose of engaging in the transactions contemplated by this Agreement and has engaged in no business activities other than as contemplated by this Agreement.

ARTICLE IV COVENANTS

4.1 <u>Closing Efforts</u>. Each of the Parties shall use its Reasonable Best Efforts to take all actions and to do all things necessary, proper or advisable to consummate the transactions contemplated by this Agreement, including using its Reasonable Best Efforts to ensure that (i) its representations and warranties remain true and correct in all material respects through the Closing Date and (ii) the conditions to the obligations of the other Parties to consummate the Merger are satisfied. Without limiting the foregoing, the Buyer (a) shall not modify or amend the Commitment Letter to value the principal amount thereof less than the amount sufficient to consummate the transactions contemplated hereby and (b) shall not terminate the Commitment Letter unless the Buyer shall have obtained a replacement financing commitment of not less than the amount sufficient to consummate the transactions contemplated hereby.

4.2 Governmental and Third-Party Notices and Consents.

(a) Until such time as this Agreement is terminated by either or both Parties pursuant to Section 7.1, each Party shall cooperate and shall use all necessary efforts and take all necessary actions to obtain, at its expense (except as provided in Section 4.8), all waivers, permits, consents, approvals or other authorizations from Governmental Entities, and to effect all registrations, filings and notices with or to Governmental Entities, as may be required for such Party to consummate the transactions contemplated by this Agreement and to otherwise comply with all applicable laws and regulations in connection with the consummation of the transactions contemplated by this Agreement. Without limiting the generality of the foregoing, each of the Parties shall promptly file any Notification and Report Forms and related material that it may be required to file with the Federal Trade Commission and the Antitrust Division of the United States Department of Justice under the HSR Act, and thereafter shall use all necessary efforts and take all necessary actions to obtain an early termination of the applicable waiting period, to make any further filings or information submissions pursuant thereto that may be necessary, proper or advisable, to respond to formal requests for additional information or documentary material pursuant to 16 C.F.R. 803.20 under the HSR Act and to avoid, contest and/or overcome any administrative or judicial action or order that would restrict, prevent or prohibit the consummation of the Merger. Nothing contained in this Agreement would require the Buyer or the Transitory Subsidiary to agree to divest, license or otherwise become subject to any limitations on, their respective rights effectively to acquire, control or operate the business of the Company, or divest itself of all or any portion of the Company's business or the business, assets or operations of the Buyer or the Transitory Subsidiary, or any of their respective Affiliates. Each Party shall cooperate with the other Parties hereto in connection with making any filing under the HSR Act or premerger filings with other Governmental Entities, including, with respect to the Party making a particular filing, giving due consideration to all reasonable additions, deletions or changes suggested in connection therewith. The Company shall provide promptly following the request of Buyer any information required in connection with the preparation of merger filings with, and any information requested by, any Governmental Entity

-28-

in respect of such filings. The Buyer shall be entitled to direct and oversee any proceedings or negotiations with any Governmental Entity relating to the government review of the Merger. The Company shall promptly inform Buyer of any communication with, and any proposed understanding, undertaking or agreement with, any Governmental Authority in respect of any such filings, investigation or other inquiry. The Company shall, to the extent practicable, provide reasonable prior notice of any meeting or communication with any Governmental Entity regarding any filings, investigation or other inquiry reviewing the transaction, such that the Buyer has reasonable opportunity to attend and participate at such a meeting or communication. The Company shall consult and cooperate with Buyer in connection with any analyses, appearances, presentations, memoranda, briefs, arguments, opinions and proposals made or submitted by or on behalf of the Company in connection with all meetings, actions and proceedings under or relating to any such application or filing. For the avoidance of doubt, all strategic decisions, presentations, submissions and any other documentation prepared in connection with the transactions contemplated by this Agreement (other than communication by the Company's counsel in the ordinary course with representatives of Governmental Entities, of which such counsel shall keep Buyer's counsel fully informed) shall be made or prepared by the Buyer, or reviewed by the Buyer to the extent the submission must be made by the Company, and the Company shall fully cooperate with Buyer in connection with the same. Neither the Buyer nor any subsidiary thereof is, or will become prior to the Effective Time, party to any agreement pursuant to which it is obligated, or may become obligated, to acquire any other entity, business or assets of an entity or business, the acquisition of which would reasonably be expected to materially delay or prevent the receipt of necessary HSR Act clearance of the transactions contemplated by this Agreement.

(b) The Company shall use its Reasonable Best Efforts to obtain, at its expense, all such waivers, consents or approvals from third parties, and to give all such notices to third parties, as are required to be listed in Section 2.4 of the Disclosure Schedule.

4.3 Stockholder Approval.

(a) The Company shall, as promptly as practicable, in accordance with its certificate of incorporation and by-laws and the applicable requirements of the Delaware General Corporation Law, solicit the written consents of the Company Stockholders for the Requisite Stockholder Approval. Within 20 days after the date of receipt of the Requisite Stockholder Approval, the Company shall send, pursuant to Sections 228 and 262(d) of the Delaware General Corporation Law, a written notice to all stockholders of the Company that did not execute such written consent informing them that this Agreement and the Merger were adopted and approved by the stockholders of the Company and that appraisal rights are available for their Company Shares pursuant to Section 262 of the Delaware General Corporation Law (which notice shall include a copy of such Section 262), and shall promptly inform the Buyer of the date on which such notice was sent. In connection with such notice, the Company shall provide to its stockholders the Disclosure Statement, which shall include (A) a summary of the Merger and this Agreement (which summary shall include a summary of the terms relating to the indemnification obligations of the Indemnifying Stockholders, the escrow arrangements and the authority of the Indemnification Representatives, and a statement that the adoption of this Agreement by the stockholders of the Company shall constitute approval of such terms) and (B) a statement that appraisal rights are available for the Company Shares pursuant to

-29-

Section 262 of the Delaware General Corporation Law. The Buyer agrees to cooperate with the Company in the preparation of the Disclosure Statement and shall give the Company any comments on the Disclosure Statement within 3 business days of receipt of such Disclosure Statement. The Company agrees not to distribute the Disclosure Statement until the Buyer has had a reasonable opportunity to review and comment on the Disclosure Statement and the Disclosure Statement has been approved by the Buyer (which approval may not be unreasonably withheld, conditioned or delayed).

- (b) The Company's board of directors shall not (i) subject to its fiduciary duties, withhold, withdraw or modify, in any manner adverse to the Buyer, the approval or recommendation by the Company's board of directors with respect to the adoption of the Merger Agreement; (ii) cause or permit the Company to enter into any letter of intent, memorandum of understanding, agreement in principle, acquisition agreement, merger agreement or similar agreement providing for the consummation of any merger, reorganization, consolidation, recapitalization, business combination, liquidation, dissolution, share exchange, sale of stock, sale of material assets or similar business transaction involving the Company, any Subsidiary or any division of the Company; or (iii) adopt, approve or recommend any merger, reorganization, consolidation, recapitalization, business combination, liquidation, dissolution, share exchange, sale of stock, sale of material assets or similar business transaction involving the Company, any Subsidiary or any division of the Company.
- 4.4 Operation of Business. Except (i) as contemplated by this Agreement, (ii) as set forth in Section 4.4 of the Disclosure Schedule or (iii) as otherwise consented to by the Buyer, during the period from the date of this Agreement to the Closing, the Company shall (and shall cause each Subsidiary to) conduct its operations in the Ordinary Course of Business and in compliance with all applicable laws and regulations (except for any failure to comply that, individually or in the aggregate, has not had and would not have a Company Material Adverse Effect) and, to the extent consistent therewith, use its Reasonable Best Efforts to preserve intact its current business organization, keep its physical assets in good working condition, keep available the services of its current officers and employees and preserve its relationships with customers, suppliers and others having business dealings with it to the end that its goodwill and ongoing business shall not be impaired in any material respect. Without limiting the generality of the foregoing, prior to the Closing, except as expressly set forth in this Agreement and except as set forth in Section 4.4 of the Disclosure Schedule, the Company shall not (and shall cause each Subsidiary not to), without the written consent of the Buyer:
- (a) issue or sell any stock or other securities of the Company or any Subsidiary or any options, warrants or rights to acquire any such stock or other securities (except pursuant to the conversion or exercise of Preferred Shares, Options or Warrants outstanding on the date hereof and the grant of stock options to employees in the Ordinary Course of Business or as otherwise consented to by the Buyer, which consent shall not be unreasonably withheld or delayed), or amend any of the terms of (including the vesting of) any Options, Warrants or restricted stock agreements (except as contemplated in Section 1.8), or repurchase or redeem any stock, Options, Warrants or other securities of the Company (except (i) in connection with a net exercise of Options or (ii) from former employees, directors or consultants in accordance with agreements providing for the repurchase of shares at their original issuance price in connection with any termination of employment with or services to the Company);

-30-

- (b) split, combine or reclassify any shares of its capital stock; or declare, set aside or pay any dividend or other distribution (whether in cash, stock or property or any combination thereof) in respect of its capital stock;
- (c) other than in the Ordinary Course of Business, create, incur or assume any indebtedness (including obligations in respect of capital leases); assume, guarantee, endorse or otherwise become liable or responsible (whether directly, contingently or otherwise) for the obligations of any other person or entity; or make any loans, advances or capital contributions to, or investments in, any other person or entity, other than out-of-pocket expenses advances to employees in the Ordinary Course of Business;
- (d) enter into, adopt or amend any Employee Benefit Plan or any employment or severance agreement or arrangement of the type described in Section 2.22(k) or (except for normal increases in the Ordinary Course of Business for employees who are not Affiliates) increase in any manner the compensation or fringe benefits of, or materially modify the employment terms of, its directors, officers or employees, generally or individually, or pay any bonus or other benefit to its directors, officers or employees (except for existing payment obligations listed in Section 2.22 of the Disclosure Schedule or any payments contemplated in Section 4.4(n) of the Disclosure Schedule) or hire any new officers or (except in the Ordinary Course of Business) any new employees;
- (e) acquire, sell, lease, license or dispose of any assets or property (including any shares or other equity interests in or securities of any Subsidiary or any corporation, partnership, association or other business organization or division thereof), other than purchases and sales of assets and licenses of Customer Deliverables in the Ordinary Course of Business;
 - (f) mortgage or pledge any of its property or assets or subject any such property or assets to any Security Interest;
 - (g) discharge or satisfy any Security Interest or pay any obligation or liability other than in the Ordinary Course of Business;
 - (h) amend its charter, by-laws or other organizational documents, except as contemplated on Schedule 4.4(h);
- (i) change its accounting methods, principles or practices, except insofar as may be required by a generally applicable change in GAAP, or make any new elections, or changes to any current elections, with respect to Taxes;
- (j) enter into, amend, terminate or waive any rights under any contract or agreement of a nature required to be listed in Section 2.12, Section 2.13 or Section 2.15 of the Disclosure Schedule, other than in the Ordinary Course of Business; or to take or omit to take any action that would constitute a violation of or default under any such contract or agreement;
- (k) make or commit to make any capital expenditure, in excess of \$50,000 per item or \$250,000 in the aggregate, other than in the Ordinary Course of Business;

-31-

- (l) institute any Legal Proceeding other than in the Ordinary Course of Business or settle any Legal Proceeding for any amount in excess of \$100,000;
- (m) take any action or fail to take any action otherwise permitted by this Agreement with the knowledge that such action or failure to take action would result in (i) any of the representations and warranties of the Company set forth in this Agreement becoming untrue or (ii) any of the conditions to the Merger set forth in Article V not being satisfied;
- (n) other than as set forth on Section 4.4(n) of the Disclosure Schedule, make any payments to its employees in connection with the Merger; or
 - (o) agree in writing or otherwise to take any of the foregoing actions.

Notwithstanding the foregoing, the Company shall be permitted to purchase, at a cost of up to \$100,000, a tail policy under the Company's current directors' and officers' liability insurance policy.

4.5 Access to Information.

- (a) The Company shall (and shall cause each Subsidiary to) permit representatives of the Buyer to have full access (at all reasonable times, and in a manner so as not to interfere with the normal business operations of the Company and the Subsidiaries), subject to compliance with applicable law (including but not limited to antitrust law) to all premises, properties, financial, tax and accounting records (including the work papers of the Company's independent accountants), contracts, other records and documents, and personnel, of or pertaining to the Company and each Subsidiary.
- (b) As soon as practicable, but in any event within 30 days, after the end of each month ending prior to the Closing (other than the last month of any fiscal year, which shall be subject to Section 4.5(c)), beginning with September 2005, the Company shall furnish to the Buyer an unaudited income statement for such month and a balance sheet as of the end of such month, prepared on a basis consistent with the financial statements described in clause (b) of the definition of Financial Statements. The Company shall use Reasonable Best Efforts to prepare such financial statements such that they present fairly in all material respects the financial condition and results of operations of the Company and the Subsidiaries on a consolidated basis as of the dates thereof and for the periods covered thereby, and such financial statements shall be consistent with the books and records of the Company and the Subsidiaries; provided that such financial statements will be subject to normal recurring year-end adjustments consistent with past practice and will not include footnotes.
- (c) Within 90 days after the end of each fiscal year ending prior to the Closing, beginning with 2005, the Company shall use Reasonable Best Efforts to furnish to the Buyer the audited consolidated balance sheets and statements of income, changes in stockholders' equity and cash flows of the Company as of the end of and for such fiscal year.
- (d) Any information obtained by the Buyer from the Company or any Subsidiary with respect to the Company and the Subsidiaries (whether pursuant to this Section 4.5 or otherwise) shall be subject, to the extent applicable, to the terms and conditions of

-32-

the Confidentiality Agreement dated February 11, 2005 between the Company and the Buyer, the terms of which as amended by Section 10.3 are incorporated herein by reference.

4.6 Notice of Breaches.

- (a) From the date of this Agreement until the Closing, the Company shall promptly deliver to the Buyer supplemental information concerning events or circumstances occurring subsequent to the date hereof which would render any representations or warranties of the Company in this Agreement or the Disclosure Schedule that are qualified as to materiality inaccurate or incomplete in any respect at any time after the date of this Agreement until the Closing or would render any other representation or warranty of the Company in this Agreement or the Disclosure Schedule inaccurate or incomplete in any material respect at any time after the date of this Agreement until the Closing, except to the extent such representations and warranties are specifically made as of a particular date (in which case this Section 4.6(a) shall apply with respect to inaccuracy or incompletion as of such date). No such supplemental information shall be deemed to avoid or cure any misrepresentation or breach of warranty or constitute an amendment of any representation, warranty or statement in this Agreement or the Disclosure Schedule.
- (b) From the date of this Agreement until the Closing, the Buyer shall promptly deliver to the Company supplemental information concerning events or circumstances occurring subsequent to the date hereof which would render any representations or warranties of the Buyer in this Agreement that are qualified as to materiality inaccurate or incomplete in any respect at any time after the date of this Agreement until the Closing or would render any other representation or warranty of the Buyer in this Agreement inaccurate or incomplete in any material respect at any time after the date of this Agreement until the Closing, except to the extent such representations and warranties are specifically made as of a particular date (in which case this Section 4.6(b) shall apply with respect to inaccuracy or incompletion as of such date). No such supplemental information shall be deemed to avoid or cure any misrepresentation or breach of warranty or constitute an amendment of any representation or warranty in this Agreement.

4.7 Exclusivity.

(a) The Company shall not, and the Company shall require each of its officers, directors, employees, representatives and agents not to, directly or indirectly, (i) initiate, solicit, engage in, encourage or otherwise facilitate any inquiry, proposal, negotiation offer or discussion with any party (other than the Buyer) concerning any merger, reorganization, consolidation, recapitalization, business combination, liquidation, dissolution, share exchange, sale of stock, sale of material assets or similar business transaction involving the Company, any Subsidiary or any division of the Company, or (ii) subject to the following sentence, furnish any non-public information concerning the business, properties or assets of the Company, any Subsidiary or any division of the Company to any party, other than (A) pursuant to the existing contractual obligations of the Company set forth on Schedule 4.7, (B) the Buyer or (C) existing or potential Power Links partners, customers, consultants, vendors or suppliers in the Ordinary Course of Business; provided that, subject to the following sentence, nothing herein shall be deemed to prohibit any director of the Company who is a representative of a Company

-33-

Stockholder on the Company's board of directors, from disclosing any such information to such Company Stockholder's officers, directors, managers, general partners or financial or legal advisors if such recipients reasonably need to know such information and are bound by a professional duty or a contractual obligation of confidentiality with respect to such information. Notwithstanding anything herein to the contrary, the Company shall not, and the Company shall require each of its officers, directors, employees, representatives and agents not to, directly or indirectly furnish any non-public information concerning the business, properties or assets of the Company, any Subsidiary or any division of the Company to any party with any intention to initiate, solicit, encourage or otherwise facilitate any inquiry, proposal, offer or discussion of the nature described in clause (i) above.

(b) The Company shall immediately notify any party with which discussions or negotiations of the nature described in clause (i) of paragraph (a) above were pending at the date of this Agreement that the Company is terminating such discussions or negotiations. If the Company receives any proposal, offer or written inquiry of the nature described in clause (i) of paragraph (a) above, the Company shall, within one business day after such receipt, notify the Buyer of such inquiry, proposal or offer, including the identity of the other party and the terms of such inquiry, proposal or offer.

4.8 Expenses.

- (a) Except as set forth herein, in the definition of Total Consideration and in Article VI and the Escrow Agreement, each of the Parties shall bear its own costs and expenses (including legal and accounting fees and expenses and investment banking and financial advisor fees) incurred in connection with this Agreement and the transactions contemplated hereby. For the avoidance of doubt, any and all such costs and expenses of the Company and its Subsidiaries shall remain an obligation of the Surviving Corporation and the Subsidiaries (and not of the Company Stockholders) following the Merger, except as provided in Article VI of this Agreement.
- (b) Notwithstanding the foregoing, the Buyer shall reimburse the Company on a monthly basis for all reasonable legal fees and expenses of the Company's antitrust counsel, incurred by the Company for services relating to antitrust issues in connection with the Merger since August 15, 2005.
- 4.9 <u>Indemnification</u>. The Buyer shall not, for a period of five years after the Closing, take any action to alter or impair any exculpatory or indemnification provisions now existing in the certificate of incorporation or by-laws of the Company for the benefit of any individual who served as a director or officer of the Company at any time prior to the Closing, except for any changes which may be required to conform with changes in applicable law and any changes which do not affect the application of such provisions to acts or omissions of such individuals prior to the Closing. The Buyer shall not terminate any tail policy under the Company's directors' and officers' liability insurance presently in effect or acquired by the Company prior to Closing in accordance with the terms hereof; provided, however, that Buyer shall have no obligation to make any payments in connection therewith.

-34-

- 4.10 Non-Solicitation. From the date of this Agreement until the earlier of (i) the Closing and (ii) 12 months following the termination of this Agreement, neither the Buyer nor the Company shall, without the consent of the other party, directly or indirectly, recruit, solicit, hire or retain any person who is an employee of the other party or who has been an employee of the other party during the period commencing 12 months prior to the date of this Agreement, or induce, or attempt to induce, an employee of the other party to terminate his or her employment with, or otherwise cease his or her relationship with, the other party; provided, however, that this Section 4.10 shall not apply to discussions or agreements between the Buyer and any employee of the Company regarding employment by the Buyer or the Surviving Corporation following the successful completion of the Merger.
- 4.11 <u>FIRPTA</u>. Prior to the Closing, (i) the Company shall deliver to the Buyer and to the Internal Revenue Service notices, in a form reasonably satisfactory to Buyer, that the Company Shares are not "U.S. real property interests" in accordance with Treasury Regulations under Sections 897 and 1445 of the Code, or (ii) each of the Company Stockholders shall deliver to the Buyer certifications, in a form reasonably satisfactory to Buyer, that they are not foreign persons in accordance with the Treasury Regulations under Section 1445 of the Code. If the Buyer does not receive either the notices or the certifications described above on or before the Closing Date, the Buyer, the Surviving Corporation, the Transitory Subsidiary, or the Payment Agent shall be permitted to withhold from the payments to be made pursuant to this Agreement any required withholding tax under Section 1445 of the Code.
- 4.12 280G Covenant. Prior to the Closing Date, if so requested by any "disqualified individual" (as defined in Section 280G(c) of the Code), the Company shall submit to a stockholder vote the right of such "disqualified individual" (as defined in Section 280G(c) of the Code) to receive any and all payments (or other benefits) that could be deemed "parachute payments" under Section 280G(b) of the Code, in a manner that satisfies the stockholder approval requirements of Section 280G(b)(5)(B) of the Code and the Treasury Regulations promulgated thereunder. Consistent with the requirements of Section 280G(b)(5)(B) and the Treasury Regulations, such vote shall establish the "disqualified individual's" right to the payment, benefit or other compensation, and before the Closing Date, the Company shall provide adequate disclosure to all Company Stockholders of all material facts concerning all payments that, but for such vote, could be deemed "parachute payments" to a "disqualified individual" under Section 280G of the Code in a manner that satisfies Section 280G(b)(5)(B)(ii) of the Code and the Treasury Regulations promulgated thereunder.
- 4.13 <u>Termination of Stockholder Agreements</u>. The Company shall use Reasonable Best Efforts to cause the termination prior to the Closing of the Amended and Restated Investor Rights Agreement, dated as of August 30, 2000, by and among the Company and the other parties named therein and the Fifth Amended and Restated Stockholders Agreement, dated as of August 30, 2000, by and among the Company and the other parties named therein.

ARTICLE V CONDITIONS TO CONSUMMATION OF THE MERGER

5.1 <u>Conditions to Each Party's Obligations</u>. The respective obligations of each Party to consummate the Merger are subject to the satisfaction of the following conditions:

-35-

- (a) this Agreement and the Merger shall have received the Requisite Stockholder Approval;
- (b) all applicable waiting periods (and any extensions thereof) under the HSR Act shall have expired or otherwise been terminated and, with respect to any necessary premerger filings with other Governmental Entities, any necessary filings have been made and approvals obtained, or waiting periods expired, as applicable;
- (c) no Legal Proceeding shall be pending or threatened in writing wherein an unfavorable judgment, order, decree, stipulation or injunction would (i) prevent consummation of the transactions contemplated by this Agreement or (ii) cause the transactions contemplated by this Agreement to be rescinded following consummation; and
- (d) the period during which Company Stockholders may demand and perfect their rights to appraisal in accordance with Section 262(d) of the Delaware General Corporation Law shall have lapsed.
- 5.2 <u>Conditions to Obligations of the Buyer and the Transitory Subsidiary</u>. The obligation of each of the Buyer and the Transitory Subsidiary to consummate the Merger is subject to the satisfaction (or waiver by the Buyer) of the following additional conditions:
- (a) the number of Dissenting Shares shall not exceed 15% of the number of outstanding Common Shares immediately prior to the Effective Time (calculated on a basis as if all outstanding Preferred Shares were converted into Common Shares);
- (b) Except as set forth on Section 5.2(b) of the Disclosure Schedule, the Company and the Subsidiaries shall have obtained (and shall have provided copies thereof to the Buyer) all of the waivers, permits, consents, approvals or other authorizations, and effected all of the registrations, filings and notices, referred to in Section 4.2 which are required on the part of the Company or the Subsidiaries;
- (c) the representations and warranties of the Company set forth in this Agreement shall be true and correct in all respects, in each case as of the date of this Agreement and as of the Closing as though made as of the Closing, except to the extent such representations and warranties are specifically made as of a particular date (in which case such representations and warranties shall be true and correct as of such date) and except to the extent such representations and warranties are not true and correct as a result of any Company Permitted Adverse Effect; provided that the condition set forth in this Section 5.2(c) shall only be deemed not to have been satisfied if the failure of such representation(s) or warranty(ies) to be true and correct has a Company Material Adverse Effect;
- (d) the Company shall have performed or complied in all material respects with its agreements and covenants, other than pursuant to Section 10.1(b), required to be performed or complied with under this Agreement as of or prior to the Closing;
- (e) no Legal Proceeding shall be pending or threatened in writing wherein an unfavorable judgment, order, decree, stipulation or injunction would have, individually or in the

-36-

aggregate, a Company Material Adverse Effect, and no such judgment, order, decree, stipulation or injunction shall be in effect;

- (f) the Company shall have delivered to the Buyer and the Transitory Subsidiary the Company Certificate;
- (g) the Buyer shall have received copies of the resignations, effective as of the Closing, of each director and officer of the Company and the Subsidiaries (other than any such resignations which the Buyer designates, by written notice to the Company, as unnecessary);
- (h) the Buyer shall have received from counsel to the Company an opinion in substantially the form attached hereto as Exhibit C, addressed to the Buyer dated as of the Closing Date;
- (i) the Buyer shall have received such other certificates and instruments (including certificates of good standing of the Company and the Subsidiaries in their jurisdiction of organization and the various foreign jurisdictions in which they are qualified, certified charter documents, certificates as to the incumbency of officers, documents necessary to transfer signing authority for all Company bank accounts and the adoption of authorizing resolutions) as it shall reasonably request in connection with the Closing; and
- (j) the Buyer shall have received such audited consolidated balance sheets and statements of income, changes in stockholders' equity and cash flows of the Company as are or will be required to be filed by the Buyer with the U.S. Securities and Exchange Commission, other than with respect to any filing by the Buyer under the Securities Act of 1933, as amended.
- 5.3 <u>Conditions to Obligations of the Company</u>. The obligation of the Company to consummate the Merger is subject to the satisfaction (or waiver by the Company) of the following additional conditions:
- (a) the Buyer shall have effected all of the registrations, filings and notices referred to in Section 4.2 which are required on the part of the Buyer;
- (b) the representations and warranties of the Buyer and the Transitory Subsidiary shall be true and correct in all respects, in each case as of the date of this Agreement and as of the Closing as though made as of the Closing, except to the extent such representations and warranties are specifically made as of a particular date (in which case such representations and warranties shall be true and correct as of such date) and except to the extent such representations and warranties are not true and correct as a result of any Buyer Permitted Adverse Effect; provided that the condition set forth in this Section 5.3(b) shall only be deemed not to have been satisfied if the failure of such representation(s) or warranty(ies) to be true and correct has a Buyer Material Adverse Effect;
- (c) each of the Buyer and the Transitory Subsidiary shall have performed or complied with in all material respects its agreements and covenants, other than pursuant to Section 10.1(b), required to be performed or complied with under this Agreement as of or prior to the Closing;

-37-

- (d) the Buyer shall have delivered to the Company the Buyer Certificate;
- (e) the Company shall have received from counsel to the Buyer and the Transitory Subsidiary an opinion in substantially the form attached hereto as Exhibit D, addressed to the Company and dated as of the Closing Date; and
- (f) the Company shall have received such other certificates and instruments (including certificates of good standing of the Buyer and the Transitory Subsidiary in their jurisdiction of organization, certified charter documents, certificates as to the incumbency of officers and the adoption of authorizing resolutions) as it shall reasonably request in connection with the Closing.

ARTICLE VI INDEMNIFICATION

- 6.1 <u>Indemnification by the Indemnifying Stockholders</u>. The Indemnifying Stockholders shall indemnify the Buyer in respect of, and hold it harmless against, any and all Damages incurred or suffered by the Surviving Corporation or the Buyer or any Affiliate thereof resulting from, relating to or constituting:
- (a) The failure of any representation or warranty made by the Company and the Subsidiaries in this Agreement or any other agreement or instrument furnished by the Company to the Buyer pursuant to this Agreement to be true and correct as of the date of this Agreement or as of the Closing Date with the same effect as though such representation or warranty had been made on and as of the Closing Date, except to the extent such representation or warranty specifically relates to a date prior to the Closing Date (in which case such representation or warranty shall be true and correct as of such prior date) and except to the extent such representations and warranties are not true and correct as a result of any Company Permitted Adverse Effect;
- (b) any failure to perform any covenant or agreement of the Company contained in this Agreement or any agreement or instrument furnished by the Company to the Buyer pursuant to this Agreement;
- (c) any failure of any Indemnifying Stockholder to have good and valid title to the issued and outstanding Company Shares issued in the name of such Indemnifying Stockholder, free and clear of all Security Interests;
- (d) any claim by a stockholder or former stockholder of the Company, or any other person or entity, seeking to assert, or based upon: (i) ownership or rights to ownership of any shares of stock of the Company; (ii) any rights of a stockholder (other than the right to receive the Consideration pursuant to this Agreement or appraisal rights under the applicable provisions of the Delaware General Corporation Law), including any option, preemptive rights or rights to notice or to vote; or (iii) any rights under the certificate of incorporation or by-laws of the Company; or
- (e) any demand by one or more Company Stockholders for appraisal under Section 262 of the Delaware General Corporation Law, if such Company Stockholders hold, in

-38-

the aggregate, more than 5% of the number of outstanding Common Shares immediately prior to the Effective Time (calculated on a basis as if all outstanding Preferred Shares were converted into Common Shares), it being understood that Damages in this case shall include, without limitation, litigation expenses incurred by the Buyer, the Company and the Surviving Corporation in connection with any such demand, as well as any amounts ultimately paid in settlement or satisfaction of the appraisal rights that exceed the Consideration otherwise issuable in respect of such Company Shares. Notwithstanding anything to the contrary herein, the Indemnifying Stockholders shall be liable only for the Excess Appraisal Rights Proportion of the Damages for any claim pursuant to this Section 6.1(e). For this purpose, the "Excess Appraisal Rights Proportion" shall be equal to a fraction (i) the numerator of which is (x) the percentage of the number of outstanding Common Shares represented by the Dissenting Shares less (y) 5%, and (ii) the denominator of which is the percentage of the number of outstanding Common Shares represented by the Dissenting Shares (with such percentages being calculated immediately prior to the Effective Time on a basis as if all outstanding Preferred Shares were converted into Common Shares).

- 6.2 <u>Indemnification by the Buyer</u>. The Buyer shall indemnify the Indemnifying Stockholders in respect of, and hold them harmless against, any and all Damages incurred or suffered by the Indemnifying Stockholders resulting from, relating to or constituting:
- (a) The failure of any representation or warranty made by the Buyer or the Transitory Subsidiary in this Agreement or any other agreement or instrument furnished by the Buyer or the Transitory Subsidiary to the Company pursuant to this Agreement to be true and correct as of the date of this Agreement or as of the Closing Date with the same effect as though such representation or warranty had been made on and as of the Closing Date, except to the extent such representation or warranty specifically relates to a date prior to the Closing Date (in which case such representation or warranty shall be true and correct as of such prior date) and except to the extent such representations and warranties are not true and correct as a result of any Buyer Permitted Adverse Effect; or
- (b) any failure to perform any covenant or agreement of the Buyer or the Transitory Subsidiary contained in this Agreement or any agreement or instrument furnished by the Buyer or the Transitory Subsidiary to the Company pursuant to this Agreement.

6.3 Indemnification Claims.

(a) An Indemnified Party shall give written notification to the Indemnifying Party of the commencement of any Third Party Action. Such notification shall be given within 25 days after receipt by the Indemnified Party of notice of such Third Party Action, and shall describe in reasonable detail (to the extent known by the Indemnified Party) the facts constituting the basis for such Third Party Action and the amount of the claimed damages; <u>provided</u>, <u>however</u>, that no delay or failure on the part of the Indemnified Party in so notifying the Indemnifying Party shall relieve the Indemnifying Party of any liability or obligation hereunder except to the extent of any damage or liability caused by or arising out of such delay or failure. Within 20 days after delivery of such notification, the Indemnifying Party may, upon written notice thereof to the Indemnified Party, assume control of the defense of such Third Party Action with counsel reasonably satisfactory to the Indemnified Party; provided that (i) the Indemnifying

-39-

Party may only assume control of such defense if (A) it acknowledges in writing to the Indemnified Party that any damages, fines, costs or other liabilities that may be assessed against the Indemnified Party in connection with such Third Party Action constitute Damages for which the Indemnified Party shall be indemnified pursuant to this Article VI and (B) the ad damnum is less than or equal to \$50 million and (ii) the Indemnifying Party may not assume control of the defense of a Third Party Action involving criminal liability or in which equitable relief is sought against the Indemnified Party. If the Indemnifying Party does not, or is not permitted under the terms hereof to, so assume control of the defense of a Third Party Action, the Indemnified Party shall control such defense. The Non-controlling Party may participate in, but not determine or conduct, such defense at its own expense. The Controlling Party shall keep the Non-controlling Party advised of the status of such Third Party Action and the defense thereof and shall consider in good faith recommendations made by the Non-controlling Party with respect thereto. The Non-controlling Party shall furnish the Controlling Party with such information as it may have with respect to such Third Party Action (including copies of any summons, complaint or other pleading which may have been served on such party and any written claim, demand, invoice, billing or other document evidencing or asserting the same) and shall otherwise fully cooperate with and assist the Controlling Party in the defense of such Third Party Action. The fees and expenses of counsel to the Indemnified Party with respect to a Third Party Action shall be considered Damages for purposes of this Agreement if (i) the Indemnified Party controls the defense of such Third Party Action pursuant to the terms of this Section 6.3(a) or (ii) the Indemnifying Party assumes control of such defense and the Indemnified Party reasonably concludes that the Indemnifying Party and the Indemnified Party have conflicting interests or different defenses available with respect to such Third Party Action. The Indemnifying Party shall not agree to any settlement of, or the entry of any judgment arising from, any Third Party Action without the prior written consent of the Indemnified Party, which shall not be unreasonably withheld, conditioned or delayed, provided, however, that such consent of the Indemnified Party shall not be required if (i) such settlement or judgment includes a complete release of the Indemnified Party from further liability and (ii) such settlement or judgment does not include any admission of liability on the part of the Indemnified Party (other than solely in respect of monetary payments, provided any such payments are paid by the Indemnifying Party) or impose any equitable relief upon the Indemnified Party. The Indemnified Party shall not agree to any settlement of, or the entry of any judgment arising from, any such Third Party Action without the prior written consent of the Indemnifying Party, which shall not be unreasonably withheld, conditioned or delayed.

- (b) In order to seek indemnification under this Article VI, an Indemnified Party shall deliver a Claim Notice to the Indemnifying Party. If the Indemnified Party is the Buyer, the Indemnifying Party shall deliver a copy of the Claim Notice to the Escrow Agent.
- (c) Within 20 days after delivery of a Claim Notice, the Indemnifying Party shall deliver to the Indemnified Party a Response, in which the Indemnifying Party shall: (i) agree that the Indemnified Party is entitled to receive all of the Claimed Amount (in which case the Response shall be accompanied by a payment by the Indemnifying Party to the Indemnified Party of the Claimed Amount, by check or by wire transfer; provided that if the Indemnified Party is the Buyer, the Indemnifying Party and the Indemnified Party shall deliver to the Escrow Agent, within three days following the delivery of the Response, a written notice executed by both parties instructing the Escrow Agent to disburse the Claimed Amount to the

-40-

- Buyer), (ii) agree that the Indemnified Party is entitled to receive the Agreed Amount (in which case the Response shall be accompanied by a payment by the Indemnifying Party to the Indemnified Party of the Agreed Amount, by check or by wire transfer; provided that if the Indemnified Party is the Buyer, the Indemnifying Party and the Indemnified Party shall deliver to the Escrow Agent, within three days following the delivery of the Response, a written notice executed by both parties instructing the Escrow Agent to disburse the Agreed Amount to the Buyer) or (iii) dispute that the Indemnified Party is entitled to receive any of the Claimed Amount.
- (d) During the 30-day period following the delivery of a Response that reflects a Dispute, the Indemnifying Party and the Indemnified Party shall use good faith efforts to resolve the Dispute. If the Dispute is not resolved within such 30-day period, such Dispute shall be resolved in a state or federal court sitting in Washington, D.C., in accordance with Section 10.11. If the Indemnified Party is the Buyer, the Indemnifying Party and the Indemnified Party shall deliver to the Escrow Agent, promptly following the resolution of the Dispute (whether by mutual agreement, arbitration, judicial decision or otherwise), a written notice executed by both parties instructing the Escrow Agent as to what (if any) portion of the Escrow Fund shall be disbursed to the Buyer and/or the Indemnifying Stockholders (which notice shall be consistent with the terms of the resolution of the Dispute).
- (e) For purposes of this Section 6.3 and the second and third sentences of Section 6.4, (i) if the Indemnifying Stockholders comprise the Indemnifying Party, any references to the Indemnifying Party (except provisions relating to an obligation to make any payments) shall be deemed to refer to the Indemnification Representatives, and (ii) if the Indemnifying Stockholders comprise the Indemnified Party, any references to the Indemnified Party (except provisions relating to an obligation to make or a right to receive any payments) shall be deemed to refer to the Indemnification Representatives. The Indemnification Representatives shall have full power and authority on behalf of each Indemnifying Stockholder to take any and all actions on behalf of, execute any and all instruments on behalf of, and execute or waive any and all rights of, the Indemnifying Stockholders under this Article VI. The Indemnification Representatives shall have no liability to any Indemnifying Stockholder for any action taken or omitted on behalf of the Indemnifying Stockholders pursuant to this Article VI.
- 6.4 <u>Survival of Representations and Warranties</u>. All representations and warranties that are covered by the indemnification agreements in Section 6.1(a) and Section 6.2(a) shall (a) survive the Closing and (b) expire on the date 12 months following the Closing Date; provided, however, that (i) Sections 2.2 and 2.3 shall survive for so long as applicable law shall permit a claim to be brought with respect thereto, and (ii) in the case of gross negligence, fraud or willful misconduct in the making of any representation or warranty, such representation or warranty shall survive for so long as applicable law shall permit a claim to be brought with respect thereto. If an Indemnified Party delivers to an Indemnifying Party, before expiration of a representation or warranty, either a Claim Notice based upon a breach of such representation or warranty, or an Expected Claim Notice based upon a breach of such representation or warranty, then the applicable representation or warranty shall survive until, but only for purposes of, the resolution of any claims arising from or related to the matter covered by such notice. If the legal proceeding or written claim (or a portion thereof) with respect to which an Expected Claim Notice has been given is definitively withdrawn or resolved in favor of the Indemnified Party,

-41-

the Indemnified Party shall promptly so notify the Indemnifying Party; and if the Indemnified Party has delivered a copy of the Expected Claim Notice to the Escrow Agent and funds have been retained in escrow after the Termination Date (as defined in the Escrow Agreement) with respect to such Expected Claim Notice, the Indemnifying Party and the Indemnified Party shall promptly deliver to the Escrow Agent a written notice executed by both parties instructing the Escrow Agent to disburse such retained funds (or appropriate portion thereof) to the Indemnifying Stockholders in accordance with the terms of the Escrow Agreement. The rights to indemnification set forth in this Article VI shall not be affected by (i) any investigation conducted by or on behalf of an Indemnified Party or any knowledge acquired (or capable of being acquired) by an Indemnified Party, whether before or after the date of this Agreement or the Closing Date, with respect to the inaccuracy or noncompliance with any representation, warranty, covenant or obligation which is the subject of indemnification hereunder or (ii) any waiver by an Indemnified Party of any closing condition relating to the accuracy of representations and warranties or the performance of or compliance with agreements and covenants.

6.5 Limitations.

- (a) Notwithstanding anything to the contrary herein, after the Closing (i) the aggregate liability of the Indemnifying Stockholders for Damages under this Article VI shall not exceed the amount of the Escrow Fund and the Escrow Fund will be the sole and exclusive source from which the Buyer may collect any Damages from the Indemnifying Stockholders for which it is entitled to indemnification under this Article VI, and (ii) the Indemnifying Stockholders shall not be liable under this Article VI unless and until the aggregate Damages for which they would otherwise be liable under this Article VI exceed \$375,000 (at which point the Indemnifying Stockholders shall become liable for the aggregate Damages under this Article VI in excess of \$250,000); provided that the limitations set forth in this sentence shall not apply (w) to a claim by the Buyer against the Company following a termination of this Agreement prior to Closing, (x) to a claim pursuant to Section 6.1(a) relating to a breach of the representations and warranties set forth in Sections 2.2 or 2.3, (y) to a claim pursuant to Section 6.1(c) or (z) in the case of gross negligence, fraud or willful misconduct; and further provided that the limitation in clause (ii) of this sentence shall not apply to a claim pursuant to Section 6.1(e). In addition, notwithstanding anything to the contrary herein, (i) except with respect to claims pursuant to Section 6.1(c), the aggregate liability of each Indemnifying Stockholder under this Article VI shall not exceed an amount equal to the value of the Consideration received by such Indemnifying Stockholder pursuant to this Agreement, (ii) no Indemnifying Stockholder shall be liable for any claim pursuant to Section 6.1(c) that relates to another stockholder of the Company, and (iii) except with respect to a claim pursuant to Section 6.1(c), the liability of the Indemnifying Stockholders under this Article VI shall be several, not joint, determined on a pro rata basis based on the percentage of the aggregate Consideration received by each such Indemnifying Stockholder pursuant to this Agreement.
- (b) Notwithstanding anything to the contrary herein, the Buyer shall not be liable under this Article VI unless and until the aggregate Damages for which it would otherwise be liable under this Article VI exceed \$375,000 (at which point the Buyer shall become liable for the aggregate Damages under this Article VI in excess of \$250,000); provided that the limitation set forth in this sentence shall not apply to a claim pursuant to Section 6.2(a) relating to a breach

-42-

of the representations and warranties set forth in Sections 3.2 or in the case of gross negligence, fraud or willful misconduct. Buyer's liability under this Section VI is also subject to the limitations set forth in Article VIII, provided that the limitations set forth in Article VIII shall not apply in the case of a claim brought after the Closing related to gross negligence, fraud or willful misconduct.

- (c) After the Closing, the rights of the Indemnified Parties under this Article VI shall be the exclusive remedy of the Indemnified Parties with respect to claims resulting from or relating to any misrepresentation, breach of warranty or failure to perform any covenant or agreement contained in this Agreement.
- (d) No Indemnifying Stockholder shall have any right of contribution against the Company or the Surviving Corporation with respect to any breach by the Company of any of its representations, warranties, covenants or agreements.
- (e) In determining the amount of any Damages for which the Indemnified Party is entitled to indemnification, the amount of any such Damages shall be determined after deducting therefrom the amount of any insurance proceeds or other third party recoveries actually received by such Indemnified Party or any of its subsidiaries in respect of such Damages. If an indemnification payment is received by an Indemnified Party, and such Indemnified Party later received insurance proceeds or other third party recoveries in respect of the related losses, such Indemnified Party shall immediately pay to the Indemnifying Party a sum equal to the lesser of (i) the actual amount of insurance proceeds or other third party recoveries or (ii) the actual amount of the indemnification payment previously paid by the Indemnifying Party with respect to such losses.
- (f) The Indemnifying Stockholders shall not be liable to the Buyer for punitive damages; provided, however that this provision shall not prevent the Buyer from obtaining indemnification for punitive damages claimed against the Buyer in a Third Party Action pursuant to this Article VI.

6.6 Limitation on Liability of Indemnification Representatives.

(a) The Indemnification Representatives shall incur no liability to the Indemnifying Stockholders with respect to any action taken or suffered by them in reliance upon any notice, direction, instruction, consent, statement or other documents believed by them to be genuinely and duly authorized, nor for other action or inaction except their own willful misconduct or gross negligence. The Indemnification Representatives may, in all questions arising under the Escrow Agreement, rely on the advice of counsel and the Indemnification Representatives shall not be liable to the Indemnifying Stockholders for anything done, omitted or suffered in good faith by the Indemnification Representatives based on such advice. The Indemnification Representatives are authorized by the Indemnifying Stockholders to incur expenses on behalf of the Indemnifying Stockholders in acting hereunder. If the Indemnification Representatives shall incur any loss, cost or expense in connection with their acting as such, the Indemnifying Stockholders will, on the written request of the Indemnification Representatives, reimburse the Indemnification Representatives for a proportionate amount thereof, determined on a pro rata basis based on the percentage of the aggregate Consideration received by each such

-43-

Indemnifying Stockholder pursuant to the Merger Agreement. Without limiting the generality of the foregoing, the Indemnification Representatives are authorized to deduct from any payment due to the Indemnifying Stockholders from the Escrow Fund the amount of such reimbursement, provided that, with respect to amounts still subject to claims by the Buyer, no such deduction from the Escrow Fund shall be made without the consent of the Buyer. The parties hereto understand and agree that the Indemnification Representatives are acting solely on behalf of and as agents for the Indemnifying Stockholders and not in their personal capacity, and in no event shall the Indemnification Representatives be personally liable to the Indemnifying Stockholders hereunder.

(b) Each Indemnifying Stockholder agrees to indemnify the Indemnification Representatives, on a pro rata basis based on the percentage of the aggregate Consideration received by each Indemnifying Stockholder pursuant to this Agreement, for any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever which may at any time be imposed on, incurred by or asserted against the Indemnification Representatives in any way relating to or arising out of this Agreement or any documents contemplated by or referred to herein or therein or the transactions contemplated hereby or thereby or the enforcement of any of the terms hereof or thereof; provided, however, that no Indemnifying Stockholder shall be liable for any of the foregoing to the extent they arise from an Indemnification Representative's gross negligence or willful misconduct. An Indemnification Representative shall be fully justified in refusing to take or to continue to take any action hereunder unless he shall first be fully indemnified to his reasonable satisfaction by the Indemnifying Stockholders against any and all liability and expense which may be incurred by him by reason of taking or continuing to take any such action.

ARTICLE VII TERMINATION

- 7.1 <u>Termination of Agreement</u>. The Parties may terminate this Agreement prior to the Closing (whether before or after Requisite Stockholder Approval), as provided below:
 - (a) the Parties may terminate this Agreement by mutual written consent;
- (b) the Buyer may terminate this Agreement (provided that the Buyer is not in material breach of this Agreement) by giving written notice to the Company in the event the Company is in breach of any representation, warranty or covenant contained in this Agreement, and such breach (i) individually or in combination with any other such breach, would cause the conditions set forth in clauses (c) or (d) of Section 5.2 not to be satisfied and (ii) is not cured within 20 days following delivery by the Buyer to the Company of written notice of such breach;
- (c) the Company may terminate this Agreement (provided that the Company is not in material breach of this Agreement) by giving written notice to the Buyer in the event the Buyer or the Transitory Subsidiary is in breach of any representation, warranty or covenant contained in this Agreement, and such breach (i) individually or in combination with any other such breach, would cause the conditions set forth in clauses (c) or (d) of Section 5.3 not to be satisfied and (ii) is not cured within 20 days following delivery by the Company to the Buyer of written notice of such breach;

-44-

- (d) any Party may terminate this Agreement by giving written notice to the other Parties at any time after (i) the stockholders of the Company have voted on whether to approve this Agreement and the Merger in the event this Agreement and the Merger failed to receive the Requisite Stockholder Approval or (ii) a court or Governmental Agency of competent jurisdiction over the Merger has issued a final, non-appealable order disapproving of the Merger or otherwise preventing the consummation of the Merger; and
- (e) the Buyer may terminate this Agreement by delivery of written notice to the Company at any time commencing 2 hours after the execution of this Agreement if the Requisite Stockholder Approval shall not have been obtained by the Company prior to delivery of such termination notice.
- 7.2 Effect of Termination. If any Party terminates this Agreement pursuant to Section 7.1, all obligations of the Parties hereunder shall terminate without any liability of any Party to any other Party (except for any liability of any Party for breaches of any covenants under this Agreement prior to such termination and except as provided in Sections 4.8 and 7.3). For avoidance of doubt, the election by a Party to exercise any termination right available to it under Section 7.1 shall not constitute a breach of this Agreement.

7.3 Liquidated Damages.

- (a) In the event that the Buyer terminates this Agreement other than pursuant to Section 7.1(a), (b), (d)(i) or (e), or the Buyer terminates this Agreement in breach of this Article VII, or if the Company terminates this Agreement pursuant to Section 7.1(c) or (d)(ii), then Buyer shall, within 15 days of such termination, pay (by wire transfer) to the Company liquidated damages of \$15,000,000, provided that in no event shall Buyer be liable for any amount in excess of \$15,000,000 pursuant to this Section 7.3. In addition, if Buyer terminates this Agreement pursuant to Section 7.1(b), the Company shall promptly, and in no event later than 15 days after the date of such termination, reimburse to Buyer all amounts previously paid by Buyer to the Company pursuant to Section 4.8(b).
- (b) Buyer and the Company agree that the agreements contained in Section 7.3(a) above are an integral part of the transactions contemplated by this Agreement and constitute liquidated damages and not a penalty.

ARTICLE VIII LIMITATIONS OF LIABILITY

- 8.1 Exclusions from Liability. Except as provided in Section 7.3, prior to the Closing Buyer shall not be liable to the Company, the Indemnifying Stockholders or any other person for indirect, consequential, exemplary, or punitive damages, including lost profits, regardless of the form of the action or the theory of recovery, even if Buyer has been advised of the possibility of such damages.
- 8.2 <u>Liability Cap</u>. Buyer's total aggregate liability for all claims arising from this Agreement, the negotiation of this Agreement, or the actions of any person in connection with the transactions contemplated by this Agreement, whether asserted by the Company, the Indemnifying Stockholders or any other person, including claims pursuant to Article VI or

-45-

Section 7.3, or claims alleging breach by the Buyer of this Agreement, shall be limited to \$15,000,000, regardless of the form of the action or the theory of recovery, except in the case of fraud or as provided in the last sentence of Section 6.5(b).

8.3 Exceptions to Limitations of Liability. The limitations set forth in Section 8.1 and 8.2 shall not apply with respect to obligations of Buyer pursuant to Section 4.8(b).

ARTICLE IX DEFINITIONS

For purposes of this Agreement, each of the following terms shall have the meaning set forth below.

"Affiliate" shall mean any affiliate, as defined in Rule 12b-2 under the Securities Exchange Act of 1934.

"Agreed Amount" shall mean part, but not all, of the Claimed Amount.

"Banking Fee Deduction" shall have the meaning set forth in Section 1.5(d).

"Buyer" shall have the meaning set forth in the first paragraph of this Agreement.

"Buyer Certificate" shall mean a certificate to the effect that each of the conditions specified in clause (c) of Section 5.1 (insofar as such clause relates to Legal Proceedings involving the Buyer or the Transitory Subsidiary) and clauses (a) through (c) of Section 5.3 is satisfied in all respects.

"Buyer Material Adverse Effect" shall mean any material adverse change, event, circumstance or development with respect to, or material adverse effect on, the business, assets, liabilities, capitalization, condition (financial or other), or results of operations of the Buyer. For the avoidance of doubt, the parties agree that the terms "material", "materially" or "materiality" as used in this Agreement with an initial lower case "m" shall have their respective customary and ordinary meanings, without regard to the meaning ascribed to Buyer Material Adverse Effect.

"Buyer Permitted Adverse Effect" shall mean any material adverse change, event, circumstance or development arising out of (a) changes in general economic or political conditions or the financing or capital markets in general or changes in currency exchange rates, (b) changes in laws or interpretations thereof by any Governmental Entity or changes in accounting requirements or principles, (c) changes affecting generally the industries or markets in which the Buyer conducts business which do not have a disproportionate impact on the Buyer, (d) the consummation of the transactions contemplated hereby or any actions by the Buyer taken pursuant to this Agreement or in connection with the transactions contemplated hereby, (e) any natural disaster, sabotage, military action or war (whether or not declared) or any escalation or worsening thereof, (f) any action required to be taken under any law or order or any existing agreement by which the Buyer (or any of its properties) is bound, or (g) the public announcement of this Agreement or the Merger, the pendency of the Merger or the performance by the Buyer of its obligations hereunder.

-46-

"Buyer Reports" shall mean (a) the Buyer's Annual Report on Form 10-K for the fiscal year ended December 31, 2004, as filed with the SEC, and (b) all other reports filed by the Buyer under Section 13 or subsections (a) or (c) of Section 14 of the Exchange Act with the SEC since January 1, 2005.

"CERCLA" shall mean the federal Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended.

"Certificate of Merger" shall mean the certificate of merger or other appropriate documents prepared and executed in accordance with Section 251(c) of the Delaware General Corporation Law.

"Certificates" shall mean stock certificates that, immediately prior to the Effective Time, represented Company Shares converted into Consideration pursuant to Section 1.5 (including any Company Shares referred to in the last sentence of Section 1.6(a)).

"Claim Notice" shall mean written notification which contains (i) a description of the Damages incurred or reasonably expected to be incurred by the Indemnified Party and the Claimed Amount of such Damages, to the extent then known, (ii) a statement that the Indemnified Party is entitled to indemnification under Article VI for such Damages and a reasonable explanation of the basis therefor, and (iii) a demand for payment in the amount of such Damages.

"Claimed Amount" shall mean the amount of any Damages incurred or reasonably expected to be incurred by the Indemnified Party.

"Closing" shall mean the closing of the transactions contemplated by this Agreement.

"Closing Date" shall mean the date two business days after the satisfaction or waiver of all of the conditions to the obligations of the Parties to consummate the transactions contemplated hereby (excluding the delivery at the Closing of any of the documents set forth in Article V), or such other date as may be mutually agreeable to the Parties; provided that if Buyer requires additional time to consummate the financing contemplated by the Commitment Letter, the "Closing Date" shall mean the date up to 30 days (or such lesser number of days as shall be required to consummate such financing) after the satisfaction or waiver of all of the conditions to the obligations of the Parties to consummate the transactions contemplated hereby (excluding the delivery at the Closing of any of the documents set forth in Article V), or such other date as may be mutually agreeable to the Parties.

"Code" shall mean the Internal Revenue Code of 1986, as amended.

"Commitment Letter" shall have the meaning set forth in Section 1.13.

"Common Shares" shall mean the shares of common stock, \$0.01 par value per share, of the Company.

"Company" shall have the meaning set forth in the first paragraph of this Agreement.

-47-

"Company Certificate" shall mean a certificate to the effect that each of the conditions specified in clause (a) of Section 5.1 and clauses (a) through (e) (insofar as clause (e) relates to Legal Proceedings involving the Company or a Subsidiary) of Section 5.2 is satisfied in all respects.

"Company Intellectual Property" shall mean the Intellectual Property owned or used by or licensed to the Company or a Subsidiary and covering, incorporated in, underlying or used in connection with the Customer Deliverables or the Internal Systems.

"Company Material Adverse Effect" shall mean any material adverse change, event, circumstance or development with respect to, or material adverse effect on, (i) the business, assets, liabilities, capitalization, condition (financial or other), or results of operations of the Company and the Subsidiaries, taken as a whole, (ii) the ability of the Buyer to operate the business of the Company and each of the Subsidiaries, taken as a whole, immediately after the Closing, substantially in the manner as such business was operated by the Company and the Subsidiaries, taken as a whole, immediately prior to the Closing or (iii) impairs the Company's ability to consummate the transactions contemplated by this Agreement. For the avoidance of doubt, the parties agree that the terms "material," "materially" or "materiality" as used in this Agreement with an initial lower case "m" shall have their respective customary and ordinary meanings, without regard to the meaning ascribed to Company Material Adverse Effect.

"Company Permitted Adverse Effect" shall mean any material adverse change, event, circumstance or development arising out of (a) changes in general economic or political conditions or the financing or capital markets in general or changes in currency exchange rates, (b) changes in laws or interpretations thereof by any Governmental Entity or changes in accounting requirements or principles, (c) changes affecting generally the industries or markets in which the Company and the Subsidiaries conduct business which do not have a disproportionate impact on the Company and the Subsidiaries, taken as a whole, (d) the consummation of the transactions contemplated hereby or any actions by the Company taken pursuant to this Agreement or in connection with the transactions contemplated hereby, (e) any natural disaster, sabotage, military action or war (whether or not declared) or any escalation or worsening thereof, (f) any action required to be taken under any law or order or any existing agreement by which the Company or any of the Subsidiaries (or any of their respective properties) is bound, or (g) the public announcement of this Agreement or the Merger, the pendency of the Merger or the performance by the Company of its obligations hereunder.

"Company Plan" shall mean any Employee Benefit Plan maintained, or contributed to, by the Company, any Subsidiary or any ERISA Affiliate.

"Company Shares" shall mean the Common Shares and the Preferred Shares together.

"Company Stock Plan" shall mean any stock option plan or other stock or equity-related plan of the Company.

"Company Stockholders" shall mean the stockholders of record of the Company immediately prior to the Effective Time.

"Consideration" shall have the meaning set forth in Section 1.5(b).

-48-

"Controlling Party" shall mean the party controlling the defense of any Third Party Action.

"Customer Deliverables" shall mean (a) the products that the Company or any Subsidiary (i) currently manufactures, markets, sells or licenses, or (ii) has manufactured, marketed, sold or licensed within the previous three years, or (iii) currently plans to manufacture, market, sell or license in the future and (b) the material services that the Company or any Subsidiary (i) currently provides, or (ii) has provided within the previous three years, or (iii) currently plans to provide in the future.

"<u>Damages</u>" shall mean any and all debts, obligations and other liabilities (whether absolute, accrued, contingent, fixed or otherwise, or whether known or unknown, or due or to become due or otherwise), diminution in value, monetary damages, fines, fees, penalties, interest obligations, deficiencies, losses and expenses (including amounts paid in settlement, interest, court costs, costs of investigators, fees and expenses of attorneys, accountants, financial advisors and other experts, and other expenses of litigation, arbitration or other dispute resolution proceedings relating to a Third Party Action or an indemnification claim under Article VI).

"<u>Disclosure Schedule</u>" shall mean the disclosure schedule provided by the Company to the Buyer on the date hereof and accepted in writing by the Buyer.

"<u>Disclosure Statement</u>" shall mean a written proxy or information statement containing the information prescribed by Section 4.3(a).

"<u>Dispute</u>" shall mean the dispute resulting if the Indemnifying Party in a Response disputes its liability for all or part of the Claimed Amount.

"<u>Dissenting Shares</u>" shall mean Company Shares held by a Company Stockholder who has not voted such Company Shares in favor of the adoption of this Agreement and with respect to which appraisal shall have been duly demanded and perfected in accordance with Section 262 of the Delaware General Corporation Law and not effectively withdrawn or forfeited prior to the Effective Time.

"Effective Time" shall mean the time at which the Surviving Corporation files the Certificate of Merger with the Secretary of State of the State of Delaware.

"Employee Benefit Plan" shall mean any "employee pension benefit plan" (as defined in Section 3(2) of ERISA), any "employee welfare benefit plan" (as defined in Section 3(1) of ERISA), and any other written or oral plan, agreement or arrangement involving direct or indirect compensation, including insurance coverage, severance benefits, disability benefits, deferred compensation, bonuses, stock options, stock purchase, phantom stock, stock appreciation or other forms of incentive compensation or post-retirement compensation.

"Environmental Law" shall mean any federal, state or local law, statute, rule, order, directive, judgment, Permit or regulation or the common law relating to the environment, occupational health and safety, or exposure of persons or property to Materials of Environmental Concern, including any statute, regulation, administrative decision or order pertaining to: (i) the presence of or the treatment, storage, disposal, generation, transportation, handling, distribution,

-49-

manufacture, processing, use, import, export, labeling, recycling, registration, investigation or remediation of Materials of Environmental Concern or documentation related to the foregoing; (ii) air, water and noise pollution; (iii) groundwater and soil contamination; (iv) the release, threatened release, or accidental release into the environment, the workplace or other areas of Materials of Environmental Concern, including emissions, discharges, injections, spills, escapes or dumping of Materials of Environmental Concern; (v) transfer of interests in or control of real property which may be contaminated; (vi) community or worker right-to-know disclosures with respect to Materials of Environmental Concern; (vii) the protection of wild life, marine life and wetlands, and endangered and threatened species; (viii) storage tanks, vessels, containers, abandoned or discarded barrels and other closed receptacles; and (ix) health and safety of employees and other persons. As used above, the term "release" shall have the meaning set forth in CERCLA.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" shall mean any entity which is, or at any applicable time was, a member of (1) a controlled group of corporations (as defined in Section 414(b) of the Code), (2) a group of trades or businesses under common control (as defined in Section 414(c) of the Code), or (3) an affiliated service group (as defined under Section 414(m) of the Code or the regulations under Section 414(o) of the Code), any of which includes or included the Company or a Subsidiary.

"Escrow Agreement" shall mean an escrow agreement in substantially the form attached hereto as Exhibit E.

"Escrow Agent" shall mean American Stock Transfer & Trust Company.

"Escrow Cash" shall have the meaning set forth in Section 1.5(b).

"Escrow Fund" shall have the meaning set forth in Section 1.9(a).

"Excess Appraisal Rights Proportion" shall have the meaning set forth in Section 6.1(e).

"Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.

"Expected Claim Notice" shall mean a notice that, as a result of a legal proceeding instituted by or written claim made by a third party, an Indemnified Party reasonably expects to incur Damages for which it is entitled to indemnification under Article VI.

"Financial Statements" shall mean:

- (a) the audited consolidated balance sheets and statements of income, changes in stockholders' equity and cash flows of the Company as of the end of and for each of the last three fiscal years, and
- (b) the Most Recent Balance Sheet and the unaudited consolidated statements of income and cash flows for the eight months ended as of the Most Recent Balance Sheet Date.

-50-

- "GAAP" shall mean United States generally accepted accounting principles.
- "Governmental Entity" shall mean any court, arbitrational tribunal, administrative agency or commission or other governmental or regulatory authority or agency.
 - "HSR Act" shall have the meaning set forth in Section 2.4.
 - "Indemnification Representatives" shall mean Dennis Beckingham and Marc Poirier.
 - "Indemnified Party" shall mean a party entitled, or seeking to assert rights, to indemnification under Article VI.
 - "Indemnifying Party" shall mean the party from whom indemnification is sought by the Indemnified Party.
 - "Indemnifying Stockholders" shall mean the Company Stockholders receiving Consideration pursuant to Section 1.5.
 - "Initial Cash" shall have the meaning set forth in Section 1.5(b).
 - "Initial Cash Amount" shall have the meaning set forth in Section 1.5(d).
 - "Initial Cash Per Share" shall have the meaning set forth in Section 1.5(d).
 - "Intellectual Property" shall mean all:
- (a) patents, patent applications, patent disclosures and all related continuation, continuation-in-part, divisional, reissue, reexamination, utility model and certificate of invention and design patents ("Patents");
- (b) trademarks, service marks, trade dress, Internet domain names, logos, trade names and corporate names and registrations and applications for registration thereof ("<u>Trademarks</u>");
 - (c) copyrightable works, copyrights and registrations and applications for registration thereof and renewal rights therefor;
 - (d) mask works and registrations and applications for registration thereof;
 - (e) schematics, computer software, firmware, technology and data;
- (f) trade secrets and confidential business information, whether patentable or nonpatentable and whether or not reduced to practice, know-how, manufacturing and product processes and techniques, research and development information, financial, marketing and business data, pricing and cost information, business and marketing plans and customer and supplier lists and information;

-51-

- (g) all documentation, records, programmers' notes and files relating to design, end-user documentation, manufacturing, quality control, sales, marketing or customer support for all intellectual property described herein; and
- (h) other proprietary rights relating to any of the foregoing (including remedies against infringements thereof and rights of protection of interest therein under the laws of all jurisdictions).

"Internal Systems" shall mean the internal systems of the Company or any Subsidiary that are used in its business or operations, including computer hardware systems, software applications and embedded systems.

"June 30, 2005 Balance Sheet" shall have the meaning set forth in Section 3.4.

"knowledge" or "to the knowledge" shall mean, with reference to the Company, the actual knowledge, after due and diligent inquiry, of one or more of the following employees of the Company: Carol Vallone, John Giordano, Barbara Ross, Peter Segall, Christopher Vento, Karen Gage, Mike Patsos, Mary Donohoe, Christopher Smith, Matthew Plasker, Karen Johnston, Mitchell Alcon and Eileen Burk.

"Lease" shall mean any lease or sublease pursuant to which the Company or a Subsidiary leases or subleases from another party any real property.

"Legal Proceeding" shall mean any action, suit, proceeding, claim, arbitration or investigation before any Governmental Entity or before any arbitrator.

"Materials of Environmental Concern" shall mean any: pollutants, contaminants or hazardous substances (as such terms are defined under CERCLA), pesticides (as such term is defined under the Federal Insecticide, Fungicide and Rodenticide Act), solid wastes and hazardous wastes (as such terms are defined under the Resource Conservation and Recovery Act), chemicals, other hazardous, radioactive or toxic materials, oil, petroleum and petroleum products (and fractions thereof), or any other material (or article containing such material) listed or subject to regulation under any law, statute, rule, regulation, order, Permit, or directive due to its potential, directly or indirectly, to harm the environment or the health of humans or other living beings.

"Merger" shall mean the merger of the Transitory Subsidiary with and into the Company in accordance with the terms of this Agreement.

"Most Recent Balance Sheet" shall mean the unaudited consolidated balance sheet of the Company as of the Most Recent Balance Sheet Date.

"Most Recent Balance Sheet Date" shall mean August 31, 2005.

"Non-controlling Party" shall mean the party not controlling the defense of any Third Party Action.

-52-

"Open Source Materials" shall mean all software or other material that is distributed as "free software," "open source software" or under a similar licensing or distribution model, including without limitation the GNU General Public License (GPL), GNU Lesser General Public License (LGPL), Mozilla Public License (MPL), BSD Licenses, the Artistic License, the Netscape Public License, the Sun Community Source License (SCSL) the Sun Industry Standards License (SISL) and the Apache License.

"Option" shall mean each option to purchase or acquire Common Shares, whether issued by the Company pursuant to the Option Plan or otherwise.

"Option Plan" shall mean the Company's 1997 Stock Plan, as amended.

"Ordinary Course of Business" shall mean the ordinary course of business consistent with past custom and practice (including with respect to frequency and amount).

"Owned Real Property" shall mean each item of real property owned by the Company or a Subsidiary.

"Parties" shall mean the Buyer, the Transitory Subsidiary and the Company.

"Patents" shall have the meaning set forth in the definition of Intellectual Property, above.

"Payment Agent" shall mean American Stock Transfer & Trust Company or such other entity mutually agreeable to the Parties.

"<u>Permits</u>" shall mean all permits, licenses, registrations, certificates, orders, approvals, franchises, variances and similar rights issued by or obtained from any Governmental Entity (including those issued or required under Environmental Laws and those relating to the occupancy or use of owned or leased real property).

"Preferred Shares" shall mean the shares of preferred stock, \$0.01 par value per share, of the Company.

"Reasonable Best Efforts" shall mean best efforts, to the extent commercially reasonable.

"<u>Requisite Stockholder Approval</u>" shall mean the adoption of this Agreement and the approval of the Merger by the requisite holders of Company Shares, as required by the Company's certificate of incorporation, as in effect at the time of such vote, and the Delaware General Corporation Law.

"Response" shall mean a written response containing the information provided for in Section 6.3(c).

"SEC" shall mean the Securities and Exchange Commission.

"Securities Act" shall mean the Securities Act of 1933, as amended.

-53-

"Security Interest" shall mean any mortgage, pledge, security interest, encumbrance, charge or other lien (whether arising by contract or by operation of law), other than (i) mechanic's, materialmen's, and similar liens, (ii) liens arising under worker's compensation, unemployment insurance, social security, retirement, and similar legislation and (iii) liens on goods in transit incurred pursuant to documentary letters of credit, in each case arising in the Ordinary Course of Business of the Company and not material to the Company.

"Software" shall mean any of the software owned by the Company or a Subsidiary.

"Subsidiary" shall mean any corporation, partnership, trust, limited liability company or other non-corporate business enterprise in which the Company (or another Subsidiary) holds stock or other ownership interests representing (a) more than 50% of the voting power of all outstanding stock or ownership interests of such entity or (b) the right to receive more than 50% of the net assets of such entity available for distribution to the holders of outstanding stock or ownership interests upon a liquidation or dissolution of such entity.

"Surviving Corporation" shall mean the Company, as the surviving corporation in the Merger.

"Taxes" shall mean all taxes, charges, fees, levies or other similar assessments or liabilities, including income, gross receipts, ad valorem, premium, value-added, excise, real property, personal property, sales, use, transfer, withholding, employment, unemployment, insurance, social security, business license, business organization, environmental, workers compensation, payroll, profits, license, lease, service, service use, severance, stamp, occupation, windfall profits, customs, duties, franchise and other taxes imposed by the United States of America or any state, local or foreign government, or any agency thereof, or other political subdivision of the United States or any such government, and any interest, fines, penalties, assessments or additions to tax resulting from, attributable to or incurred in connection with any tax or any contest or dispute thereof.

"<u>Tax Returns</u>" shall mean all reports, returns, declarations, statements or other information required to be supplied to a taxing authority in connection with Taxes.

"Third Party Action" shall mean any claim, demand, action, cause of action, suit or proceeding by a person or entity other than a Party for which indemnification may be sought by a Party under Article VI.

"Total Consideration" shall have the meaning set forth in Section 1.5(d).

"Total Consideration Per Share" shall have the meaning set forth in Section 1.5(d).

"Trademarks" shall have the meaning set forth in the definition of Intellectual Property, above.

"Transitory Subsidiary" shall have the meaning set forth in the first paragraph of this Agreement.

-54-

"Warrant" shall mean each warrant or other contractual right to purchase or acquire Company Shares, provided that Options and Preferred Shares shall not be considered Warrants.

ARTICLE X MISCELLANEOUS

10.1 Press Releases and Announcements.

- (a) No Party shall issue any press release or public announcement relating to the subject matter of this Agreement prior to the Closing without the prior written approval of the other Parties; <u>provided</u>, <u>however</u>, that any Party may make any public disclosure it believes in good faith is required by applicable law, regulation or stock market rule (in which case the disclosing Party shall use reasonable efforts to advise the other Parties and provide them with a copy of the proposed disclosure prior to making the disclosure) and the Buyer at its discretion shall be permitted to issue press releases after the Effective Time customary for a transaction of this nature.
- (b) The Parties have heretofore agreed to the content of a public communications plan relating to the period prior to the Effective Time. Notwithstanding anything to the contrary herein, each Party agrees, prior to the Effective Time, to use Reasonable Best Efforts to avoid making any statements to employees, customers, suppliers, business partners or other third parties in connection with this Agreement, as well as in any press releases or other public statements or disclosures relating thereto, that are materially inconsistent with the terms of such communications plan, except to the extent required to avoid making a material misstatement or omission. Each Party shall refrain from making any written statements that would reasonably be expected to create additional material contractual obligations to customers regarding the product and services offerings of any Party. Each Party will cooperate with the other Party in communicating any matters relating to the foregoing. Neither Party will be liable to the other Party in the event of a breach of this Section 10.1(b) for any indirect, consequential, exemplary, or punitive damages, including lost profits, regardless of the form of the action or the theory of recovery, even if such Party has been advised of the possibility of such damages.
- 10.2 No Third Party Beneficiaries. This Agreement shall not confer any rights or remedies upon any person other than the Parties and their respective successors and permitted assigns; provided, however, that (a) the provisions in Article I concerning issuance of the Consideration, (b) the provisions of Article VI concerning indemnification and (c) the provisions of Section 4.9 concerning indemnification are intended for the benefit of the individuals specified therein.
- 10.3 Entire Agreement. This Agreement (including the documents referred to herein) constitutes the entire agreement among the Parties and supersedes any prior understandings, agreements or representations by or among the Parties, written or oral, with respect to the subject matter hereof; provided that the Confidentiality Agreement dated February 11, 2005 between the Buyer and the Company shall remain in effect in accordance with its terms until the earlier of the Effective Time or December 31, 2007; and the Parties hereby agree to extend its term accordingly.

-55-

- 10.4 <u>Succession and Assignment</u>. This Agreement shall be binding upon and inure to the benefit of the Parties named herein and their respective successors and permitted assigns. No Party may assign any of its rights or delegate any of its performance obligations hereunder without the prior written approval of the other Parties; provided that the Transitory Subsidiary may assign its rights, interests and obligations hereunder to an Affiliate of the Buyer. Any purported assignment of rights or delegation of performance obligations in violation of this Section 10.4 is void.
- 10.5 <u>Counterparts and Facsimile Signature</u>. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same instrument. This Agreement may be executed by facsimile signature.
- 10.6 <u>Headings</u>. The section headings contained in this Agreement are inserted for convenience only and shall not affect in any way the meaning or interpretation of this Agreement.
- 10.7 <u>Notices</u>. All notices, requests, demands, claims, and other communications hereunder shall be in writing. Any notice, request, demand, claim or other communication hereunder shall be deemed duly delivered four business days after it is sent by registered or certified mail, return receipt requested, postage prepaid, or one business day after it is sent for next business day delivery via a reputable nationwide overnight courier service, in each case to the intended recipient as set forth below:

If to the Company:

WebCT, Inc 6 Kimball Lane Lynnfield, MA 01940

Attention: Chief Financial Officer

Facsimile: 781-224-4705

If to the Buyer or the Transitory Subsidiary:

Blackboard Inc. 1899 L Street, NW

Washington, DC 20036

Attention: General Counsel Facsimile: 202-318-1281

Copy to:

William B. Asher

Choate, Hall & Stewart LLP Two International Place Boston, MA 02110 Facsimile: 617-248-4000

Copy to:

Brent B. Siler

Wilmer Cutler Pickering Hale and Dorr LLP

1455 Pennsylvania Avenue, N.W.

Washington, DC 20004 Facsimile: 202-942-8484

Any Party may give any notice, request, demand, claim or other communication hereunder using any other means (including personal delivery, expedited courier, messenger service, telecopy, ordinary mail or electronic mail), but no such notice, request, demand, claim or other communication shall be deemed to have been duly given unless and until it actually is received by the party for whom it is intended. Any Party may change the address to which notices, requests, demands, claims, and other communications hereunder are to be delivered by giving the other Parties notice in the manner herein set forth.

- 10.8 Governing Law. All matters arising out of or relating to this Agreement and the transactions contemplated hereby (including without limitation its interpretation, construction, performance and enforcement) shall be governed by and construed in accordance with the internal laws of the State of Delaware without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of laws of any jurisdictions other than those of the State of Delaware.
- 10.9 Amendments and Waivers. The Parties may mutually amend any provision of this Agreement at any time prior to the Closing; provided, however, that any amendment effected subsequent to the Requisite Stockholder Approval shall be subject to any restrictions contained in the Delaware General Corporation Law. No amendment of any provision of this Agreement shall be valid unless the same shall be in writing and signed by all of the Parties. No waiver of any right or remedy hereunder shall be valid unless the same shall be in writing and signed by the Party giving such waiver. No waiver by any Party with respect to any default, misrepresentation or breach of warranty or covenant hereunder shall be deemed to extend to any prior or subsequent default, misrepresentation or breach of warranty or covenant hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence.
- 10.10 Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If the final judgment of a court of competent jurisdiction declares that any term or provision hereof is invalid or unenforceable, the Parties agree that the court making the determination of invalidity or unenforceability shall have the power to limit the term or provision, to delete specific words or phrases, or to replace any invalid or unenforceable term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be enforceable as so modified.
- 10.11 <u>Submission to Jurisdiction</u>. Each Party (a) submits to the jurisdiction of any state or federal court sitting in Washington, D.C. in any action or proceeding arising out of or relating to this Agreement, (b) agrees that all claims in respect of such action or proceeding may be heard and determined in any such court, (c) waives any claim of inconvenient forum or other challenge to venue in such court, and (d) agrees not to bring any action or proceeding arising out of or relating to this Agreement in any other court. Each Party agrees to accept service of any summons, complaint or other initial pleading made in the manner provided for the giving of notices in Section 10.7, provided that nothing in this Section 10.11 shall affect the right of any Party to serve such summons, complaint or other initial pleading in any other manner permitted by law.

10.12 Construction.

(a) The language used in this Agreement shall be deemed to be the language chosen by the Parties to express their mutual intent, and no rule of strict construction shall be applied against any Party.

-57-

- (b) Any reference to any federal, state, local or foreign statute or law shall be deemed also to refer to all rules and regulations promulgated thereunder, unless the context requires otherwise.
 - (c) Any reference herein to "including" shall be interpreted as "including without limitation".
- (d) Any reference to any Article, Section or paragraph shall be deemed to refer to an Article, Section or paragraph of this Agreement, unless the context clearly indicates otherwise.
- 10.13 <u>Further Assurances</u>. At any time and from time to time after the Closing, as requested by any Party hereto, the other Party shall promptly execute and deliver, or cause to be executed or delivered, all such documents, instruments and certificates, and shall take, or cause to be taken, all such further or other actions, as are necessary to evidence and effectuate the transactions contemplated by this Agreement.

-58-

IN WITNESS WHEREOF, the Parties have executed this Agreement as of the date first above written.

BLACKBOARD INC.

By: /s/ Michael L. Chasen

Title: CEO

COLLEGE ACQUISITION SUB, INC.

By: /s/ Matthew Small

Title: President

WEBCT, INC.

By: /s/ John Giordano

Title: Executive Vice President

Chief Financial Officer and Treasurer

-59-

Exhibit 99.2

EXECUTION VERSION

CREDIT SUISSE Eleven Madison Avenue New York, NY 10010

CONFIDENTIAL

October 12, 2005

Blackboard Inc. 1899 L Street, N.W. Washington D.C. 20036

Attention: Michael L. Chasen

Chief Executive Officer

<u>Project College</u> <u>\$80,000,000 Senior Secured Credit Facilities</u> Commitment Letter

Ladies and Gentlemen:

You have advised Credit Suisse (together with its affiliates, "Credit Suisse", "we" or "us") that Blackboard Inc. ("you" or the "Borrower") intends to acquire (the "Acquisition") all the capital stock of WebCT, Inc. a Delaware corporation (the "Company") from its current shareholders (together, the "Sellers"), and to consummate the other Transactions (such term and each other capitalized term used but not defined herein having the meaning assigned to such term in the Summary of Principal Terms and Conditions attached hereto as Exhibit A (the "Term Sheet").

You have further advised us that, in connection therewith, the Borrower will obtain the senior secured credit facilities (collectively, the "*Facilities*") described in the Term Sheet in an aggregate principal amount of up to \$80,000,000 consisting of (A) a \$10,000,000 senior secured revolving credit facility, and (B) a \$70,000,000 senior secured term loan facility.

1. Commitments.

In connection with the foregoing, Credit Suisse (the "*Initial Lender*") is pleased to advise you of its commitment to provide the entire principal amount of the Facilities, upon the terms and subject to the conditions set forth or referred to in this commitment letter (including the Term Sheet and other attachments hereto, this "*Commitment Letter*").

2. Agency Roles.

You hereby appoint Credit Suisse to act, and Credit Suisse hereby agrees to act, as sole administrative agent, sole collateral agent, sole bookrunner and sole lead arranger for the Facilities (collectively, the "Agent") on the terms and subject to the conditions set forth or referred to in this Commitment Letter. The Agent, in such capacities, will perform the duties and exercise the authority customarily performed and exercised by it in such roles. You agree that no other titles will be awarded and no compensation (other than that expressly contemplated by this Commitment Letter and the Fee Letter referred to below) will be paid in connection with the Facilities unless you and we shall so agree.

3. Syndication.

We reserve the right, prior to or after the execution of definitive documentation for the Facilities, to syndicate all or a portion of our commitments hereunder to a group of banks, financial institutions and other institutional lenders (together with the Initial Lender, the "Lenders") identified by us in consultation with you. We intend to commence syndication efforts promptly upon the execution of this Commitment Letter, and you agree actively to assist us in completing a satisfactory syndication. Such assistance shall include (a) your using commercially reasonable efforts to ensure that any syndication efforts benefit materially from your existing lending and investment banking relationships, (b) senior management of the Borrower being made available to the proposed Lenders as well as your using commercially reasonable efforts to cause representatives and advisors of the Company to be made available to the proposed Lenders, in each case, upon reasonable request during normal business hours, (c) assistance by you (and your using commercially reasonable efforts to cause the assistance by the Company) in the preparation of a Confidential Information Memorandum for each of the Facilities and other customary marketing materials to be used in connection with the syndications, (d) prior to the launch of the syndications your using commercially reasonable efforts to obtain ratings for each of the Facilities from each of Standard & Poor's Ratings Service ("S&P") and Moody's Investors Service, Inc. ("Moody's") and (e) the hosting, with the Agent, of one or more meetings of prospective Lenders as reasonably requested by the Agent. You agree, at the request of the Agent, to assist in the preparation of a version of the Confidential Information Memorandum and other marketing materials and presentations to be used in connection with the syndication of the Facilities, consisting exclusively of information and documentation that is either (i) publicly available or (ii) not material with respect to the Borrower, the Company or their respective subsidiaries or any of their respective securities for purposes of United States Federal and state securities laws (all such information and documentation being "Public Information"). Any information and documentation that is not Public Information is referred to herein as "Private Information". You further agree that each document to be disseminated by the Agent to any Lender in connection with the Facilities will be identified by you as either (i) containing Private Information or (ii) containing solely Public Information.

Subject to the terms hereof, the Agent will manage all aspects of any syndication, including decisions as to the selection of institutions to be approached and when they will be approached, when their commitments will be accepted, which institutions will participate, the allocation of the commitments among the Lenders, any naming rights and the amount and distribution of fees among the Lenders. To assist the Agent in its syndication efforts, you agree promptly to prepare and provide (and to use commercially reasonable efforts to cause the Company to provide) to the Agent all information with respect to the Borrower, the Company and their respective subsidiaries, the Transactions and the other transactions contemplated hereby, including all financial information and projections (the "*Projections*"), as we may reasonably request.

4. Representations and Warranties.

You hereby represent and covenant (and it shall be a condition to our commitment hereunder and agreement to perform the services described herein) that (a) all information other than the Projections (the "*Information*") that has been or will be made available to the Agent by or on behalf of you or any of your representatives is or will be, when furnished, complete and correct in all material respects and does not or will not, when furnished, contain any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements contained therein not materially misleading in light of the circumstances under which such statements are made and (b) the Projections that have been or will be made available to the Agent by or on behalf of you or any of your representatives have been or will be prepared in good faith based upon accounting principles consistent with the historical financial statements of the Borrower and upon assumptions that are reasonable at the time made and at the time the related

2

Projections are made available to the Agent, it being understood that actual results may vary materially from the Projections. You agree that if at any time prior to the closing of the Facilities you become aware that any of the representations in the preceding sentence would be incorrect if the Information and Projections were being furnished, and such representations were being made, at such time, then you will promptly supplement the Information and the Projections so that such representations will be correct under those circumstances. In arranging and syndicating the Facilities, we will be entitled to use and rely primarily on the Information and the Projections without responsibility for independent verification thereof.

5. <u>Fees</u>.

As consideration for the Initial Lender's commitment hereunder and the Agent's agreement to perform the services described herein, you agree to pay to the Agent and the Initial Lender the fees set forth in this Commitment Letter and in the fee letter dated the date hereof and delivered herewith with respect to the Facilities (the "Fee Letter").

6. Conditions Precedent.

The Initial Lender's commitment hereunder and the Agent's agreement to perform the services described herein are subject to, (a) our not having discovered or otherwise become aware of any information not previously disclosed to or known by us that is inconsistent with any Information (including, without limitation, the information contained in the financial model (and any actual or implicit assumptions thereunder) of the Borrower, the Company and their respective subsidiaries that was provided to us on October 5, 2005 (the "Bank Forecast"), which inconsistency or inconsistencies, individually or in the aggregate, constitutes or would reasonably be expected to constitute, a material adverse affect on the business, assets, liabilities, operations, financial condition, or historical, current or projected operating results (as set forth in the Bank Forecast) of the Borrower, the Company and their respective subsidiaries, taken as a whole, (b) except as previously disclosed to us, there not having occurred any event, change or condition since December 31, 2004 (the date of the most recent audited financial statements of the Company delivered to the Agent as of the date hereof) that, individually or in the aggregate, has had, or would reasonably be expected to have, a material adverse effect on the business, assets, liabilities, operations, financial condition, or the historical, current or projected operating results (as set forth in the Bank Forecast) of the Borrower and its subsidiaries and the Company and its subsidiaries, taken as a whole, (c) our satisfaction that, prior to and during the syndication of the Facilities, there shall be no other issues of debt securities or commercial bank or other credit facilities of the Borrower, the Company or their respective subsidiaries being offered, placed or arranged, (d) the negotiation, execution and delivery of definitive documentation with respect to the Facilities reasonably satisfactory to the Agent and its counsel, (e) the Agent's having been afforded a period of at least 30 consecutive days following the launch of the general syndication of the Facilities and immediately prior to the Closing Date to syndicate the Facilities (provided that such period shall not include any day from and including December 16, 2005 through and including January 2, 2006), (f) your providing a detailed business plan or projections of the Borrower and its subsidiaries for the years 2005 through 2010 and for the quarters beginning with the third fiscal quarter of 2005 and through the second fiscal quarter of 2007, in form and substance reasonably satisfactory to the Agent, (g) your compliance, in all material respects, with the terms of this Commitment Letter and the Fee Letter, and (h) the other conditions set forth or referred to in the Term Sheet and the other exhibits hereto. For the purposes of clauses (b) and (c) of this paragraph, any material adverse change, event, circumstance or development with respect to the Company and its subsidiaries arising out of or resulting from the public announcement of the Merger Agreement, the Acquisition or the pendency or consummation of the Acquisition (including the requirement for or timing of regulatory approvals) shall be disregarded.

3

7. Indemnification.

You agree (a) to indemnify and hold harmless the Agent, the Initial Lender and their respective affiliates and their respective officers, directors, employees, agents, advisors, controlling persons, members and successors and assigns (each, an "Indemnified Person") from and against any and all losses, claims, damages, liabilities and reasonable expenses, joint or several, to which any such Indemnified Person may become subject arising out of or in connection with this Commitment Letter, the Fee Letter, the Transactions, the Facilities or any related transaction or any claim, litigation, investigation or proceeding relating to any of the foregoing, regardless of whether any such Indemnified Person is a party thereto (and regardless of whether such matter is initiated by a third party or by the Borrower or any of its affiliates), and to reimburse each such Indemnified Person upon demand for any reasonable legal or other expenses incurred in connection with investigating or defending any of the foregoing (with, subject to customary exceptions for actual or potential conflicts of interest, all Indemnified Persons to use a single counsel in each jurisdiction), provided that the foregoing indemnity will not, as to any Indemnified Person, apply to losses, claims, damages, liabilities or related expenses to the extent they are found in a final, non-appealable judgment of a court of competent jurisdiction to have resulted primarily from the willful misconduct or gross negligence of such Indemnified Person, and (b) to reimburse the Agent and the Initial Lender from time to time, upon presentation of a summary statement, for all reasonable out-of-pocket expenses (including but not limited to expenses of the Agent's and the Initial Lender's due diligence investigation, consultants' fees, syndication expenses, travel expenses and fees, disbursements and other charges of counsel), in each case incurred in connection with the Facilities and the preparation of this Commitment Letter, the Fee Letter, the definitive documentation for the Facilities and any security arrangements in connection therewith; provided, however, that such expenses in excess of \$275,000 shall not be reimbursed unless approved in advance by the Borrower, which approval shall not be unreasonably withheld. Notwithstanding any other provision of this Commitment Letter, no Indemnified Person shall be liable for any damages arising from the unauthorized use by others of information or other materials obtained through electronic, telecommunications or other information transmission systems or for any indirect, special, punitive or consequential damages in connection with its activities related to the Facilities.

8. Sharing Information.

You acknowledge that the Agent, the Initial Lender and their respective affiliates may be providing debt financing, equity capital or other services (including financial advisory services) to other companies in respect of which you may have conflicting interests regarding the transactions described herein or otherwise. Neither we nor any of our affiliates will use confidential information obtained from you by virtue of the transactions contemplated by this Commitment Letter or our other relationships with you in connection with the performance by us of services for other companies, and we will not furnish any such information to other companies. You also acknowledge that neither we nor any of our affiliates has any obligation to use in connection with the transactions contemplated by this Commitment Letter, or to furnish to you, confidential information obtained by us from other companies.

You further acknowledge and agree that (a) no fiduciary, advisory or agency relationship between you and Credit Suisse has been created in respect of any of the transactions contemplated by this Commitment Letter, irrespective of whether Credit Suisse and/or its affiliates have advised or are advising you on other matters, (b) Credit Suisse, on the one hand, and you, on the other hand, have an arms-length business relationship that does not directly or indirectly give rise to, nor do you rely on, any fiduciary duty on the part of Credit Suisse, (c) you are capable of evaluating and understanding, and you do understand and accept, the terms, risks and conditions of the transactions contemplated by this Commitment Letter, (d) you have been advised that Credit Suisse and its affiliates are engaged in a broad range of transactions that may involve interests that differ from your interests and that Credit Suisse has

4

no obligation to disclose any such interests or transactions to you by virtue of any fiduciary, advisory or agency relationship, and (e) you waive, to the fullest extent permitted by law, any claims you may have against Credit Suisse for breach of fiduciary duty or alleged breach of fiduciary duty and agree that Credit Suisse shall have no liability (whether direct or indirect) to you in respect of such a fiduciary duty claim or to any person asserting a fiduciary duty claim on behalf of or in right of you, including your stockholders, employers or creditors.

You acknowledge that Credit Suisse First Boston LLC, an affiliate of Credit Suisse, is a full service securities firm engaged in securities trading and brokerage activities as well as providing investment banking and other financial services. In the ordinary course of business, Credit Suisse First Boston LLC and its affiliates may provide investment banking and other financial services to, and/or acquire, hold or sell, for their own accounts and the accounts of customers, equity, debt and other securities and financial instruments (including bank loans and other obligations) of, the Borrower and other companies with which the Borrower may have commercial or other relationships. With respect to any securities and/or financial instruments so held by Credit Suisse First Boston LLC or any of its affiliates or customers, all rights in respect of such securities and financial instruments, including any voting rights, will be exercised by the holder of the rights, in its sole discretion.

9. Assignments, Amendments, Governing Law, Etc.

This Commitment Letter shall not be assignable by you without the prior written consent of the Agent (and any attempted assignment without such consent shall be null and void), is intended to be solely for the benefit of the parties hereto (and Indemnified Persons) and is not intended to confer any benefits upon, or create any rights in favor of, any person other than the parties hereto (and Indemnified Persons). In connection with any syndication contemplated by Section 3 above, the Initial Lender may assign its commitment hereunder to any of its affiliates or any prospective Lender. Notwithstanding the foregoing, the Initial Lender shall not be released from the syndicated portion of its commitment hereunder by any such syndication unless and until the applicable assignee first funds such portion of the commitment hereunder; provided, however, that nothing herein shall be read to require the Initial Lender to maintain any portion of the loans under the facilities after the Closing Date. Any and all obligations of, and services to be provided by, the Agent or the Initial Lender hereunder (including, without limitation, the Initial Lender's commitment) may be performed and any and all rights of the Agent or the Initial Lender hereunder may be exercised by or through any of its affiliates or branches. This Commitment Letter may not be amended or any provision hereof waived or modified except by an instrument in writing signed by the Agent, the Initial Lender and you. This Commitment Letter may be executed in any number of counterparts, each of which shall be an original and all of which, when taken together, shall constitute one agreement. Delivery of an executed counterpart of a signature page of this Commitment Letter by facsimile transmission shall be effective as delivery of a manually executed counterpart hereof. This Commitment Letter and the Fee Letter supersede all prior understandings, whether written or oral, between us with respect to the Facilities. THIS COMMITMENT LETTER SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.

10. Jurisdiction.

Each of the parties hereto hereby irrevocably and unconditionally (a) submits, for itself and its property, to the exclusive jurisdiction of any New York State court or Federal court of the United States of America sitting in New York City, and any appellate court from any thereof, in any action or proceeding arising out of or relating to this Commitment Letter, the Fee Letter or the transactions contemplated hereby or thereby, or for recognition or enforcement of any judgment, and agrees that all claims in respect of any such action or proceeding may be heard and determined in such New York State

5

court or, to the extent permitted by law, in such Federal court, (b) waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the laying of venue of any suit, action or proceeding arising out of or relating to this Commitment Letter, the Fee Letter or the transactions contemplated hereby or thereby in any New York State court or in any such Federal court and (c) waives, to the fullest extent permitted by law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such court.

11. Confidentiality.

This Commitment Letter is delivered to you on the understanding that neither this Commitment Letter nor the Fee Letter nor any of their terms or substance, nor the activities of the Agent or the Initial Lender pursuant hereto, shall be disclosed, directly or indirectly, to any other person except (a) to your officers, directors, employees, attorneys, accountants and advisors on a confidential and need-to-know basis or (b) as required by applicable law or compulsory legal process (in which case you agree to inform us promptly thereof); *provided* that you may disclose this Commitment Letter and the contents hereof (but not the Fee Letter or the contents thereof) to the Company and its officers, directors, employees, attorneys, accountants and advisors on a confidential and need-to-know basis.

Notwithstanding anything herein to the contrary, any party to this Commitment Letter (and any employee, representative or other agent of such party) may disclose to any and all persons, without limitation of any kind, the tax treatment and tax structure of the transactions contemplated by this Commitment Letter and the Fee Letter and all materials of any kind (including opinions or other tax analyses) that are provided to it relating to such tax treatment and tax structure, except that (i) tax treatment and tax structure shall not include the identity of any existing or future party (or any affiliate of such party) to this Commitment Letter or the Fee Letter, and (ii) no party shall disclose any information relating to such tax treatment and tax structure to the extent nondisclosure is reasonably necessary in order to comply with applicable securities laws. For this purpose, the tax treatment of the transactions contemplated by this Commitment Letter and the Fee Letter is the purported or claimed U.S. federal income tax treatment of such transactions and the tax structure of such transactions is any fact that may be relevant to understanding the purported or claimed U.S. federal income tax treatment of such transactions.

12. Surviving Provisions.

The compensation, reimbursement, indemnification, confidentiality, jurisdiction and waiver of jury trial provisions contained herein and in the Fee Letter shall remain in full force and effect regardless of whether definitive financing documentation shall be executed and delivered and notwithstanding the termination of this Commitment Letter or the Initial Lender's commitments and the Agent's agreements hereunder.

13. Waiver of Jury Trial.

EACH OF THE PARTIES HERETO IRREVOCABLY WAIVES THE RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING, CLAIM OR COUNTERCLAIM BROUGHT BY OR ON BEHALF OF ANY PARTY RELATED TO OR ARISING OUT OF THIS COMMITMENT LETTER, THE FEE LETTER OR THE PERFORMANCE OF SERVICES HEREUNDER OR THEREUNDER.

6

14. PATRIOT Act Notification.

The Agent and the Initial Lender hereby notify you that pursuant to the requirements of the USA PATRIOT Act, Title III of Pub. L. 107-56 (signed into law October 26, 2001) (the "*PATRIOT Act*"), the Agent and each Lender is required to obtain, verify and record information that identifies the Borrower, which information includes the name, address, tax identification number and other information regarding the Borrower that will allow the Agent or such Lender to identify the Borrower in accordance with the PATRIOT Act. This notice is given in accordance with the requirements of the PATRIOT Act and is effective as to the Agent and each Lender.

15. Acceptance and Termination.

If the foregoing correctly sets forth our agreement, please indicate your acceptance of the terms of this Commitment Letter and of the Fee Letter by returning to us executed counterparts hereof and of the Fee Letter not later than 5:00 p.m., New York City time, on October 12, 2005. The Initial Lender's commitment hereunder and the Agent's agreements contained herein will expire at such time in the event that the Agent has not received such executed counterparts in accordance with the immediately preceding sentence. In the event that the initial borrowing in respect of the Facilities does not occur on or before the second anniversary of this Commitment Letter, then this Commitment Letter and the Initial Lender's commitment and the Agent's undertakings hereunder shall automatically terminate unless the Agent and the Initial Lender shall, in their sole discretion, agree to an extension. Before such date, if any event occurs or information becomes available that, in the reasonable and good faith judgment of the Agent and the Initial Lender, results or is likely to result in the failure to satisfy any condition precedent set forth or referred to in this Commitment Letter, the Agent shall so notify the Borrower in writing (including a detailed statement of the event or information that results or is likely to result in the failure to satisfy a condition precedent and identifying the specific condition precedent) and provide a period of at least sixty (60) days in which such failure (or likely failure) may be cured. If, following such cure period, it continues to be the reasonable and good faith judgment of the Agent and the Initial Lender that the event or information that was the subject of the notice results or is likely to result in the failure to satisfy any condition precedent set forth or referred to in this Commitment Letter, then the Agent and the Initial Lender may terminate this Commitment Letter and the Initial Lender's commitment and the Agent's undertakings hereunder.

[Remainder of this page intentionally left blank]

7

We are pleased to have been given the opportunity to assist you in connection with the financing for the Acquisition.

Very truly yours,

CREDIT SUISSE, CAYMAN ISLANDS BRANCH, as the Agent and the Initial Lender

By /s/ Robert Hetu

Name: Robert Hetu Title: Director

By /s/ Cassandra Droogan

Name: Cassandra Droogan

Title: Associate

Accepted and agreed to as of the date first above written:

BLACKBOARD INC.

By /s/ Michael L. Chasen

Name: Michael Chasen

Title: CEO

8

EXHIBIT A

Project College \$80,000,000 Senior Secured Credit Facilities Summary of Principal Terms and Conditions

Borrower:

Blackboard Inc. a Delaware corporation (the "Borrower").

Transactions:

The Borrower intends to acquire (the "Acquisition") all the capital stock of WebCT, Inc. (the "Company") pursuant to an agreement and plan of merger (the "Merger Agreement") proposed to be entered into by and between the Borrower and the Company and substantially in the form provided to the Agent on October 11, 2005. In connection with the Acquisition, (a) the Borrower will acquire the Company from the Sellers with the Sellers receiving an aggregate amount of \$180.0 million in cash (the "Acquisition Consideration"), (b) the Borrower will obtain the senior secured credit facilities described below under the caption "Facilities" (the "Facilities"), (c) fees and expenses (to be acceptable to the Arranger) incurred in connection with the foregoing in an aggregate amount not to exceed \$6.4 million (the "Transaction Expenses") will be paid, and (d) certain other costs (to be acceptable to the Arranger) in an aggregate amount not to exceed \$10.0 million (the "Other Costs") will be paid. The transactions described in this paragraph are collectively referred to herein as the "Transactions".

Sources and Uses:

The approximate sources and uses of the funds necessary to consummate the Transactions are set forth in Exhibit B to the Commitment Letter (the "Commitment Letter") to which this Term Sheet is attached.

Agent:

Credit Suisse, acting through one or more of its branches or affiliates ("*Credit Suisse*"), will act as sole and exclusive administrative agent and collateral agent (collectively, the "*Agent*") for a syndicate of banks, financial institutions and other institutional lenders (together with Credit Suisse, the "*Lenders*"), and will perform the duties customarily associated with such roles.

Sole Bookrunner and Sole Lead

Arranger:

Credit Suisse will act as sole bookrunner and sole lead arranger for the Facilities (the "*Arranger*"), and will perform the duties customarily associated with such roles.

Syndication Agent:

At the option of the Arranger, a financial institution identified by the Arranger and acceptable to the Borrower (the "Syndication Agent").

Documentation Agent:

At the option of the Arranger, a financial institution identified by the Arranger and acceptable to the Borrower (the "*Documentation Agent*").

Facilities:

(A) A first priority senior secured term loan facility in an aggregate principal amount of up to \$70.0 million (the "*Term Facility*").

ANNEX 1-1

(B) A first priority senior secured revolving credit facility in an aggregate principal amount of up to \$10.0 million (the "*Revolving Facility*"), of which up to an aggregate amount to be agreed upon will be available in the form of letters of credit.

In connection with the Revolving Facility, Credit Suisse (in such capacity, the "Swingline Lender") will make available to the Borrower a swingline facility under which the Borrower may make short-term borrowings of up to an aggregate amount to be agreed upon. Except for purposes of calculating the Commitment Fee described below, any such swingline borrowings will reduce availability under the Revolving Facility on a dollar-for-dollar basis. Each Lender under the Revolving Facility shall, promptly upon request by the Swingline Lender, fund to the Swingline Lender its pro rata share of any swingline borrowings.

Purpose:

- (A) The proceeds of the Term Facility will be used by the Borrower, on the date of the initial borrowing under the Facilities (the "Closing Date"), solely (a) to pay the Cash Consideration, (b) to refinance certain existing indebtedness, and (c) to pay the Transaction Expenses and Other Costs.
- (B) The proceeds of loans under the Revolving Facility will be used by the Borrower solely for working capital purposes.
- (C) Letters of credit will be used solely to support payment obligations incurred in the ordinary course of business by the Borrower and its subsidiaries.

Availability:

- (A) The full amount of the Term Facility must be drawn in a single drawing on the Closing Date. Amounts borrowed under the Term Facility that are repaid or prepaid may not be reborrowed.
- (B) No loans under the Revolving Facility may be made on the Closing Date. Thereafter, loans under the Revolving Facility will be available at any time prior to the final maturity of the Revolving Facility, in minimum principal amounts to be agreed upon. Amounts repaid under the Revolving Facility may be reborrowed.

Interest Rates and Fees:

As set forth on Annex I hereto.

Default Rate:

The applicable interest rate plus 2.0% per annum.

A-2

Letters of Credit:

Letters of credit under the Revolving Facility will be issued by Credit Suisse or another Lender acceptable to the Borrower and the Agent (the "Issuing Bank"). Each letter of credit shall expire not later than the earlier of (a) 12 months after its date of issuance and (b) the fifth business day prior to the final maturity of the Revolving Facility; provided, however, that any letter of credit may provide for renewal thereof for additional periods of up to 12 months (which in no event shall extend beyond the date referred to in clause (b) above).

Drawings under any letter of credit shall be reimbursed by the Borrower on the same business day. To the extent that the Borrower does not reimburse the Issuing Bank on the same business day, the Lenders under the Revolving Facility shall be irrevocably obligated to reimburse the Issuing Bank pro rata based upon their respective Revolving Facility commitments.

The issuance of all letters of credit shall be subject to the customary procedures of the Issuing Bank.

Final Maturity and Amortization:

(A) Term Facility

The Term Facility will mature on the date that is six years after the Closing Date, and will amortize in equal quarterly installments in an aggregate annual amount equal to 1.0% of the original principal amount of the Term Facility during the first five years thereof with the balance payable in equal quarterly installments during the final year of the Term Facility.

(B) Revolving Facility

The Revolving Facility will mature and the commitments thereunder will terminate on the date that is five years after the Closing Date.

Guarantees:

All obligations of the Borrower under the Facilities and under any interest rate protection or other hedging arrangements entered into with the Agent, the Arranger, a Lender or any affiliate of any of the foregoing (the "Hedging Arrangements") will be unconditionally guaranteed (the "Guarantees") by each existing and subsequently acquired or organized domestic subsidiary, including, without limitation, the Company and its subsidiaries (the "Subsidiary Guarantors").

Security:

The Facilities, the Guarantees and the Hedging Arrangements will be secured by substantially all the assets of the Borrower and each Subsidiary Guarantor (collectively, the "Collateral"), including but not limited to: (a) a perfected first-priority pledge of all the capital stock of the Company, (b) a perfected first-priority pledge of all the capital stock held by the Borrower or any Subsidiary Guarantor (which pledge, in the case of any foreign subsidiary, shall be limited to 100% of the non-voting stock (if any) and 65% of the voting stock

A-3

of such foreign subsidiary, and (c) perfected first-priority security interests in, and mortgages on, substantially all tangible and intangible assets of the Borrower and each Subsidiary Guarantor (including but not limited to accounts receivable, inventory, equipment, general intangibles, investment property, intellectual property, real property, cash, deposit and securities accounts, commercial tort claims, letter of credit rights, intercompany notes and proceeds of the foregoing).

All the above-described pledges, security interests and mortgages shall be created on terms, and pursuant to documentation, reasonably satisfactory to the Lenders, and none of the Collateral shall be subject to any other pledges, security interests or mortgages, subject to customary and limited exceptions to be agreed upon.

Mandatory Prepayments:

Loans under the Facilities shall be prepaid, in accordance with the order of application described below, with (a) if the ratio of total debt to EBITDA (to be defined) is greater than 2.0 to 1.0, 50% of Excess Cash Flow (to be defined), (b) 100% of the net cash proceeds of all asset sales or other dispositions of property by the Borrower and its subsidiaries (including insurance and condemnation proceeds) (subject to exceptions and reinvestment provisions to be agreed upon), (c) 100% of the net cash proceeds of issuances, offerings or placements of debt obligations of the Borrower and its subsidiaries (subject to exceptions to be agreed upon), and (d) if the ratio of total debt to EBITDA at the time of such issuance is greater than 2.0 to 1.0, 50% of the net cash proceeds of issuances of equity securities of the Borrower and its subsidiaries (subject to exceptions to be agreed upon).

The above-described mandatory prepayments shall be applied pro rata to the remaining amortization payments under the Term Facility. Each Lender under the Term Facility shall have the right to decline any such mandatory prepayment of its loans, in which case the amount of such prepayment that is declined shall be re-offered to the other Lenders under the Term Facility which shall have accepted such payment on a pro rata basis and, if further declined, such amount shall be applied as set forth below. When there are no longer outstanding loans under the Term Facility, such prepayments will be applied to prepay outstanding loans under the Revolving Facility and cash collateralize letters of credit, in each case with no corresponding permanent reduction of commitments under the Revolving Facility.

Voluntary Prepayments and Reductions in Commitments:

Voluntary reductions of the unutilized portion of the commitments under the Facilities and prepayments of borrowings thereunder will be permitted at any time, in minimum principal amounts to be agreed upon, without premium or penalty, subject to reimbursement of the Lenders' redeployment costs in the case of a prepayment of Adjusted LIBOR borrowings other than on the last day of the relevant interest period. All voluntary prepayments of the Term Facility will be applied pro rata to the remaining amortization payments under the

A-4

Term Facility.

Representations and Warranties:

Usual for facilities and transactions of this type and others to be reasonably specified by the Agent, including, without limitation, accuracy of financial statements and other information; no material adverse change; absence of litigation; no violation of agreements or instruments; compliance with laws (including ERISA, margin regulations and environmental laws); payment of taxes; ownership of properties; inapplicability of the Investment Company Act; solvency; effectiveness of governmental approvals; labor matters; environmental and other regulatory matters; and validity, priority and perfection of security interests in the Collateral.

Conditions Precedent to Initial Borrowing:

Delivery of such legal opinions, corporate documents and officers' and public officials' certifications as the Agent shall reasonably request; first-priority perfected security interests in the Collateral (free and clear of all liens, subject to customary exceptions and other limited exceptions to be agreed upon); receipt of lien and judgment searches reasonably satisfactory to the Agent; execution of the Guarantees and other definitive documentation, which shall be in full force and effect; accuracy of representations and warranties (subject to customary materiality and material adverse change qualifiers); absence of defaults, prepayment events or creation of liens under debt instruments or other agreements; evidence of authority; payment of fees and expenses; and obtaining of insurance reasonably satisfactory to the Agent.

The initial borrowing under the Facilities will also be subject to the conditions precedent set forth in Exhibit C to the Commitment Letter.

Conditions Precedent to all Borrowings:

Delivery of notice, accuracy of representations and warranties, and absence of defaults.

Affirmative Covenants:

Usual for facilities and transactions of this type and others to be reasonably specified by the Agent (to be applicable to the Borrower and its subsidiaries), including, without limitation, maintenance of corporate existence and rights; performance of obligations; delivery of consolidated and consolidating financial statements and other information, including information required under the PATRIOT Act; delivery of notices of default, litigation, ERISA events and material adverse change; maintenance of properties in good working order; maintenance of satisfactory insurance; maintenance of a rating of the Facilities by each of Standard & Poor's Ratings Service ("S&P") and Moody's Investors Service, Inc. ("Moody's") but, for the avoidance of doubt, with no obligation to maintain any particular minimum rating; compliance with laws; inspection of books and properties; hedging arrangements reasonably satisfactory to the Agent; further assurances; continued perfection of security interests in existing and subsequently acquired Collateral; and payment of

A-5

taxes.

Negative Covenants:

Usual for facilities and transactions of this type and others to be reasonably specified by the Agent (to be applicable to the Borrower and its subsidiaries), including, without limitation, limitations on dividends on, and redemptions and repurchases of, capital stock and other restricted payments; limitations on prepayments, redemptions and repurchases of debt; limitations on liens and sale leaseback transactions; limitations on loans and investments; limitations on debt, guarantees and hedging arrangements; limitations on mergers, acquisitions and asset sales; limitations on transactions with affiliates; limitations on changes in business conducted by the Borrower and its subsidiaries; limitations on restrictions on ability of subsidiaries to pay dividends or make distributions; limitations on amendments of debt and other material agreements; and limitations on capital expenditures.

Selected Financial Covenants:

Subject to the following levels and the financial definitions set forth in Annex 2 attached hereto, to be the following:

- (a) Maximum ratios of Total Debt to Consolidated EBITDA (for the period of four consecutive fiscal quarters most recently ended on or prior to such date) of no less than 3.25 to 1.0
- (b) Minimum Interest Coverage Ratio (for the period of four consecutive fiscal quarters most recently ended on or prior to such date) of no more than 4.00 to 1.0
- (c) Maximum Capital Expenditures, as determined as of the last day of each fiscal quarter set forth below, for the four fiscal quarters ending on such day, of no more than the amounts set forth opposite such fiscal quarter:

Fiscal Quarter Ending	Amount (expressed in millions)
June 30, 2005	\$14.6
September 30,	
2005	\$15.2
December 31,	
2005	\$15.7
March 31, 2006	\$15.4
June 30, 2006	\$15.3
September 30,	
2006	\$16.0
December 31,	
2006	\$17.0
March 31, 2007	\$17.4
June 30, 2007	\$17.9
September 30,	
2007	\$18.4

A-6

December 31, 2007	\$18.7
Each fiscal quarter ending in 2008	\$20.6
Each fiscal quarter ending in 2009	\$22.7
Each fiscal quarter ending in 2010	\$24.9
Each fiscal quarter ending in 2011	\$29.7

Events of Default:

Usual for facilities and transactions of this type (subject, where appropriate, to thresholds and grace periods to be agreed upon) and others to be reasonably specified by the Agent, including, without limitation, nonpayment of principal, interest or other amounts; violation of covenants; incorrectness of representations and warranties in any material respect; cross default and cross acceleration; bankruptcy; material judgments; ERISA events; actual or asserted invalidity of guarantees or security documents; and Change of Control (to be defined).

Voting:

Amendments and waivers of the definitive credit documentation will require the approval of Lenders holding more than 50% of the aggregate amount of the loans and commitments under the Facilities (with certain amendments and waivers also requiring class votes), except that the consent of each Lender shall be required with respect to, among other things, (a) increases in the commitment of such Lender, (b) reductions of principal, interest or fees, (c) extensions of final maturity or scheduled amortization and (d) releases of guarantors (other than in connection with permitted asset sales) or all or substantially all of the Collateral.

Cost and Yield Protection:

Usual for facilities and transactions of this type.

Assignments and Participations:

The Lenders will be permitted to assign (a) loans under the Term Facility without the consent of (but with notice to) the Borrower and (b) loans and commitments under the Revolving Facility with the consent of the Borrower, the Swingline Lender and the Issuing Bank, in each case not to be unreasonably withheld or delayed; *provided* that such consent of the Borrower shall not be required (i) if such assignment is made to another Lender or an affiliate of a Lender or (ii) after the occurrence and during the continuance of a default. All assignments will require the consent of the Agent, not to be unreasonably withheld or delayed. Each assignment will be in an amount of an integral multiple of \$1,000,000. Assignments will be by novation and will not be required to be pro rata between the Facilities.

The Lenders will be permitted to sell participations in loans and commitments without restriction. Voting rights of participants shall be limited to matters in respect of (a) increases in commitments,

A-7

(b) reductions of principal, interest or fees, (c) extensions of final maturity or scheduled amortization and (d) releases of guarantors (other than in connection with permitted asset sales) or all or substantially all of the Collateral.

Expenses and Indemnification:

The Borrower will indemnify the Arranger, the Agent, the Syndication Agent, the Documentation Agent, the Lenders, their respective affiliates, successors and assigns and the officers, directors, employees, agents, advisors, controlling persons and members of each of the foregoing (each an "Indemnified Person") and hold them harmless from and against all costs, expenses (including reasonable fees, disbursements and other charges of counsel) and liabilities of such Indemnified Person arising out of or relating to any claim or any litigation or other proceeding (regardless of whether such Indemnified Person is a party thereto and regardless of whether such matter is initiated by a third party or by the Borrower or any of its affiliates) that relates to the Transactions, including the financing contemplated hereby, the Acquisition or any transactions connected therewith, provided that no Indemnified Person will be indemnified for any cost, expense or liability to the extent determined in the final, non-appealable judgment of a court of competent jurisdiction to have resulted primarily from its gross negligence or willful misconduct. In addition, all out-of-pocket expenses (including, without limitation, fees, disbursements and other charges of counsel) of the Arranger, the Agent, the Syndication Agent, the Documentation Agent and the Lenders for enforcement costs and documentary taxes associated with the Facilities will be paid by the Borrower.

Governing Law and Forum:

New York.

Counsel to Agent and Arranger:

Latham & Watkins LLP.

A-8

ANNEX 1

Interest Rates:

The interest rates under the Facilities will be as follows:

Revolving Facility

If the Facilities are rated at or above the Ratings Threshold (as defined below) on the Closing Date, at the option of the Borrower, Adjusted LIBOR plus 2.25-2.50% or ABR plus 1.25-1.50%. If the Facilities are rated below the Ratings Threshold on the Closing, at the option of the Borrower, Adjusted LIBOR plus 2.50-2.75% or ABR plus 1.50-1.75%.

Term Facility

If the Facilities are rated at or above the Ratings Threshold on the Closing Date, at the option of the Borrower, Adjusted LIBOR plus 2.25-2.50% or ABR plus 1.25-1.50%. If the Facilities are rated below the Ratings Threshold on the Closing, at the option of the Borrower, Adjusted LIBOR plus 2.50-2.75% or ABR plus 1.50-1.75%.

Facilities

The Borrower may elect interest periods of 1, 2, 3 or 6 months for Adjusted LIBOR borrowings.

Calculation of interest shall be on the basis of the actual days elapsed in a year of 360 days (or 365 or 366 days, as the case may be, in the case of ABR loans based on the Prime Rate) and interest shall be payable at the end of each interest period and, in any event, at least every three months.

ABR is the Alternate Base Rate, which is the higher of Credit Suisse's Prime Rate and the Federal Funds Effective Rate plus 1/2 of 1.0%.

Adjusted LIBOR will at all times include statutory reserves.

As used herein, "*Ratings Threshold*" means a rating of at least B+ by S&P and a rating of a least B1 by Moody's, in each case with no negative outlook.

Letter of Credit Fee:

A per annum fee equal to the spread over Adjusted LIBOR under the Revolving Facility will accrue on the aggregate face amount of outstanding letters of credit under the Revolving Facility, payable in arrears at the end of each quarter and upon the termination of the Revolving Facility, in each case for the actual number of days elapsed over a 360-day year. Such fees shall be distributed to the Lenders participating in the Revolving Facility pro rata in accordance with the amount of each such Lender's Revolving Facility commitment. In addition, the Borrower shall pay to the Issuing Bank, for its own account, (a) a fronting fee equal to a percentage per annum to be agreed upon of the aggregate face

ANNEX 1-1

amount of outstanding letters of credit, payable in arrears at the end of each quarter and upon the termination of the Revolving Facility, calculated based upon the actual number of days elapsed over a 360-day year, and (b) customary issuance and administration fees.

Commitment Fees:

0.50% per annum on the undrawn portion of the commitments in respect of the Revolving Facility, payable quarterly in arrears after the Closing Date and upon the termination of the commitments, calculated based on the number of days elapsed in a 360-day year.

Changes in Interest Rate Margins:

The definitive credit documentation for the Facilities will contain provisions under which, from and after the date of delivery of the Borrower's financial statements covering a period of at least six full months after the Closing Date, and so long as no default shall have occurred and be continuing, interest rate margins under the Revolving Facility will be subject to change in increments to be agreed upon based upon performance goals to be agreed upon.

Annex 1-2

ANNEX 2

For purposes of the "Selected Financial Covenants" section in the Term Sheet to which this Annex 2 is attached, the following terms shall have the following meanings (subject, in each case, to final negotiation in the definitive documentation for the Facilities):

"Consolidated EBITDA" shall mean, for any period, Consolidated Net Income for such period plus

- (a) without duplication and to the extent deducted in determining such Consolidated Net Income, the sum of
 - (i) Consolidated Interest Expense for such period,
 - (ii) consolidated income tax expense for such period,
 - (iii) all amounts attributable to depreciation and amortization for such period,
 - (iv) any non-cash charges and expenses (other than the write-down of current assets) for such period (provided, that to the extent that all or any portion of the income of any person is excluded from Consolidated Net Income pursuant to the definition thereof for all or any portion of such period any amounts set forth in the preceding clauses (i) through (iv) that are attributable to such person shall not be included for purposes of this definition for such period or portion thereof),
 - (v) non-cash compensation charge arising from any grant of stock, stock options or other equity-based awards,
 - (vi) non-recurring charges and expenses related to the Acquisition, including (A) employee compensation for the period from the Closing Date to the date that is 150 days thereafter to be paid to employees identified for termination, (B) other severance costs and accrued vacation for terminated employees, (C) retention bonuses, (D) lease termination expenses and other restructuring charges, and (E) fees for Acquisition-related and/or integration-related professional services; *provided*, *however*, in each case, that such charges and expenses are (A) incurred within the first year following the Closing Date and (B) identified in reasonable detail in a certificate of the Borrower's chief financial officer,
 - (vii) any non-cash decrease in consolidated revenues during such period resulting from the following purchase accounting adjustments made in accordance with GAAP in connection with the Acquisition: (A) loss of deferred revenue of the Company representing the difference between revenues of the Company which would have been amortized into income from deferred revenue (without giving effect to purchase accounting adjustments) and the actual recognized revenues of the Company as of the Closing Date (giving effect to purchase accounting adjustments), and (B) the adjustment resulting from changes in revenue recognition as a result of transitional changes in licensing methodology,
 - (viii) any other extraordinary, unusual or non-recurring losses or charges; *provided*, *however*, in each case, that such charges and expenses are identified in reasonable detail in a certificate of the Borrower's chief financial officer, and
 - (ix) without duplication, any losses on the sale of assets or businesses outside of the ordinary course of business,

ANNEX 2-1

minus

- (b) without duplication,
 - (i) all cash payments made during such period on account of reserves, restructuring charges and other non-cash charges added to Consolidated Net Income pursuant to clause (a)(iv) above in a previous period, and
 - (ii) to the extent included in determining such Consolidated Net Income, any extraordinary gains and all non-cash gains for such period, all determined on a consolidated basis in accordance with GAAP;

provided, that for purposes of calculating Consolidated EBITDA for any period (A) the Consolidated EBITDA of any entity acquired by the Borrower or any of its subsidiaries pursuant to an acquisition permitted by the definitive documentation during such period shall be included on a Pro Forma Basis for such period (assuming the consummation of such acquisition and the incurrence or assumption of any indebtedness in connection therewith occurred as of the first day of such period) and (B) the Consolidated EBITDA of any person or line of business sold or otherwise disposed of by the Borrower or any of its subsidiaries during such period for shall be excluded for such period (assuming the consummation of such sale or other disposition and the repayment of any indebtedness in connection therewith occurred as of the first day of such period).

"Consolidated Net Income" shall mean, for any period, the net income or loss of the Borrower and its subsidiaries for such period determined on a consolidated basis in accordance with GAAP; provided, that there shall be excluded (a) the income of any such subsidiary to the extent that the declaration or payment of dividends or similar distributions by such subsidiary of that income is not at the time permitted by operation of the terms of its charter or any agreement, instrument, judgment, decree, statute, rule or governmental regulation applicable to such subsidiary, (b) except to the extent otherwise provided in paragraph (A) of the proviso in the definition of 'Consolidated EBITDA', the income or loss of any person accrued prior to the date it becomes a subsidiary or is merged into or consolidated with the Borrower or any of its subsidiaries or the date that such person's assets are acquired by the Borrower or any subsidiary, (c) the income of any person (other than any subsidiary of the Borrower) in which any other person (other than the Borrower or a wholly owned subsidiary of the Borrower or any director holding qualifying shares in accordance with applicable law) has an interest, except to the extent of the amount of dividends or other distributions actually paid to the Borrower or one of its wholly owned subsidiaries by such person during such period, and (d) any gains attributable to sales of assets out of the ordinary course of business.

"Consolidated Interest Expense" shall mean, for any period, the sum of (a) the interest expense (including imputed interest expense in respect of Capital Lease Obligations and synthetic lease obligations) of the Borrower and its subsidiaries for such period (including all commissions, discounts and other fees and charges owed by the Borrower and its subsidiaries with respect to letters of credit and bankers' acceptance financing), net of interest income, in each case determined on a consolidated basis in accordance with U.S. GAAP, plus (b) any interest accrued during such period in respect of indebtedness of the Borrower or any of its subsidiaries that is required to be capitalized rather than included in consolidated interest expense for such period in accordance with U.S. GAAP. For purposes of the foregoing, interest expense shall be determined after giving effect to any net payments made or received by the Borrower or any of its subsidiaries with respect to interest rate hedging agreements.

"Capital Expenditures" shall mean, for any period, with respect to any person, (i) the additions to property, plant and equipment and other capital expenditures of such person and its consolidated

Annex 1-2

subsidiaries that are (or should be set forth in a consolidated statement of cash flows for such person prepared in accordance with U.S. GAAP and (ii) Capital Lease Obligations incurred by such person and its consolidated subsidiaries for such period.

"Capital Lease Obligations" shall mean, for any period, with respect to any person, the obligations of such person to pay rent or other amounts under any lease of (or other arrangement conveying the right to use) real or personal property, or a combination thereof, which obligations are required to be classified and accounted for as capital leases on the balance sheet of such person under U.S. GAAP, and the amount of such obligations at any time shall be the capitalized amount thereof at such time determined in accordance with U.S. GAAP.

"Interest Coverage Ratio" shall mean, on any date, the ratio of (a) Consolidated EBITDA for the period of four consecutive fiscal quarters most recently ended on or prior to such date, taken as one accounting period, to (b) Consolidated Interest Expense (excluding interest expense in connection with any prepayment by the Borrower of any of the Facilities) for the period of four consecutive fiscal quarters ended on or prior to such date, taken as one accounting period.

Annex 1-3

EXHIBIT B

Sources and Uses of Funds (in millions of dollars) (all figures are approximate)

Sources of Funds		Uses of Funds	
Existing Cash	\$ 122.0	Acquisition Consideration	\$ 180.0
Revolving Facility ¹	0.0	Transaction Expenses	5.7
Term Facility	70.0	Repayment of Existing Revolving Facility	0.4
		Other Costs	6.0
Total Sources	\$ 192.0	Total Uses	\$ 192.0

Represents amount to be drawn under the \$10.0 million Revolving Facility on the Closing Date.

B-1

EXHIBIT C

Project College \$80,000,000 Senior Secured Credit Facilities Summary of Additional Conditions Precedent2

Except as otherwise set forth below, the initial borrowing under the Facilities shall be subject to the following additional conditions precedent:

- 1. The Acquisition and the other Transactions shall be consummated simultaneously with the closing under the Facilities in accordance with applicable law and on the terms described in the Term Sheet; the Merger Agreement entered into, and all other related documentation shall be substantially in the form disclosed to the Agent, on October 12, 2005, with such changes and additions as are reasonably satisfactory to the Agent.
- 2. After giving effect to the Transactions and the other transactions contemplated hereby, the Borrower and its subsidiaries shall have outstanding no indebtedness or preferred stock other than (a) the loans and other extensions of credit under the Facilities and (b) other limited indebtedness to be agreed upon.
- 3. The Agent shall have received (a) the unaudited consolidated balance sheet of the Company as of August 31, 2005 and the unaudited consolidated statements of income and cash flows for the eight months ended as of August 31, 2005, and (b) for each month ending after August 31, 2005, an unaudited income statement of the Company for each such month and a balance sheet as of the end of such month, in each case prepared in accordance with US GAAP applied on a consistent basis throughout the periods covered thereby. The financial statements described in clauses (a) and (b) are subject to normal recurring year-end adjustments and do not include footnotes.
- 4. The Agent shall have received a pro forma consolidated and consolidating balance sheet and related pro forma consolidated and consolidating statements of income and cash flows of the Borrower as of and for the twelve-month period ending on the last day of the most recently completed four-fiscal quarter period, prepared after giving effect to the Transactions as if the Transactions had occurred as of such date (in the case of such balance sheet) or at the beginning of such period (in the case of such other financial statements), which financial statements shall not be materially inconsistent with the forecasts previously provided to the Agent in a manner that constitutes or would reasonably be expected to constitute a material adverse effect on the business, assets, liabilities, operations, financial condition, or historical, current or projected operating results of the Borrower, the Company and their respective subsidiaries, taken as a whole, as reflected in the Bank Forecast.
- 5. The Agent shall be satisfied that the Borrower's consolidated pro forma EBITDA for the four-fiscal quarter period most recently ended prior to the Closing Date (prepared in accordance with Regulation S-X and with such further adjustments in form and substance reasonably satisfactory to the Agent, in each case, to give pro forma effect to the Transactions as if they had occurred at the beginning of such four-fiscal quarter period) (such consolidated pro forma EBITDA, "*Pro Forma EBITDA*") shall not be less than \$21.5 million.
- 6. The Agent shall be satisfied that the Borrower's ratio of Senior Secured Debt (to be defined) on the Closing Date to Pro Forma EBITDA shall be no more than 3.25 to 1.0.
- All capitalized terms used but not defined herein have the meanings given to them in the Commitment Letter to which this term sheet is attached, including Exhibit A thereto.

C-1

- 7. The Agent shall have received a certificate from the chief financial officer of the Borrower certifying that the Borrower and its subsidiaries, on a consolidated basis after giving effect to the Transactions and the other transactions contemplated hereby, are solvent.
- 8. All requisite material non-governmental third parties shall have approved or consented to the Transactions and the other transactions contemplated hereby to the extent required, all applicable appeal periods shall have expired and there shall be no litigation, administrative or judicial action, actual or threatened, that could reasonably be expected to restrain, prevent or impose burdensome conditions on the Transactions or the other transactions contemplated hereby.
- 9. All applicable waiting periods (and any extensions thereof) under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, shall have expired or otherwise been terminated and, with respect to any necessary premerger filings with any other administrative agency or commission, court, arbitrational tribunal, or other governmental or regulatory authority or agency, any necessary filings have been made and approvals obtained, or waiting periods shall have expired, as applicable. No action, suit, proceeding, claim, arbitration or investigation before any court, arbitrational tribunal, administrative agency or commission or other governmental or regulatory authority or agency, or before any arbitrator, shall be pending or threatened in writing wherein an unfavorable judgment, order, decree, stipulation or injunction would (i) prevent consummation of the transactions contemplated by the Merger Agreement to be rescinded following consummation.
- 10. The Agent shall have received, at least five business days prior to the Closing Date, all documentation and other information required by regulatory authorities under applicable "know your customer" and anti-money laundering rules and regulations, including without limitation the PATRIOT Act.

C-2

Exhibit 99.3



News Release

Corporate and Investor Contact: Michael J. Stanton

Blackboard Inc. +1 (202) 463-4860 ext. 2305

Blackboard Inc. Provides Initial FY 2006 Guidance and Reaffirmed Guidance for 2005

Washington, DC – October 12, 2005 – Blackboard Inc. (NASDAQ: BBBB), a leading provider of enterprise software and services to education institutions, today reaffirmed its financial guidance for the remainder of 2005 and provided initial full year guidance for 2006. The company's guidance excludes any impact of its planned acquisition of WebCT, Inc., which was announced earlier today.

Outlook for the Third Quarter and Full Year of 2005

The following forward-looking statements regarding future financial performance are based on current expectations and actual results may differ materially. These statements do not reflect the potential impact of mergers, acquisitions or other business combinations that may be completed after the date of this release.

Blackboard expects that its effective tax rate will continue to be in the range of 4 to 7 percent through the end of 2005. Additionally, the Company's guidance does not incorporate the impact of expensing stock-based compensation under FAS 123(R), which the Company will adopt beginning January 1, 2006.

Blackboard provides cash net income and cash net income per share in this press release as additional information regarding Blackboard's operating results. These measures are not in accordance with, nor are they an alternative for, generally accepted accounting principles (GAAP) and may be different from cash net income and other non-GAAP measures used by other companies. Blackboard believes that this presentation of cash net income and cash net income per share provides useful information to investors regarding additional financial and business trends relating to Blackboard's financial condition and results of operations.

For the third quarter of 2005 the Company reaffirms its prior guidance and expects:

- Revenue of approximately \$35.4 to \$35.9 million;
- Net income of approximately \$6.2 to \$6.5 million, resulting in diluted EPS of approximately \$0.21 to \$0.23 per share. This is based on an estimated 28.9 million diluted shares and a 4% effective tax rate for the quarter; and
- Cash net income of approximately \$6.2 to \$6.6 million after adding back the tax adjusted amortization of intangibles of approximately \$75,000, which would result in cash EPS of approximately \$0.22 to \$0.23 per share. This is based on an estimated 28.9 million diluted shares and an estimated 4% effective tax rate for the quarter.

The Company reaffirms its prior guidance for the full year of 2005 and expects:

- Revenue of approximately \$134.0 to \$135.0 million;
- Net income of approximately \$24.0 to \$24.6 million, resulting in diluted EPS of approximately \$0.84 to \$0.86 per share, which is based on an estimated 28.5 million diluted shares and a 4% effective tax rate for the full year; and
- Cash net income of approximately \$24.3 to \$24.9 million after adding back the tax adjusted amortization of intangibles of approximately \$300,000, which would result in cash EPS of approximately \$0.85 to \$0.87 per share based on an estimated 28.5 million diluted shares and an estimated 4% effective tax rate for the full year.

Initial Guidance for the Full Year of 2006

The Company's full year 2006 guidance does not incorporate the impact of expensing stock-based compensation under FAS 123(R), which the Company will adopt beginning January 1, 2006 and assumes an effective tax rate of 35 to 38% for the full year 2006. The higher effective tax rate assumes Blackboard no longer maintains valuation allowances against the net operating losses even though the net operating losses will continue to provide cash tax benefits. If this assumption is incorrect our effective rate could be materially lower.

For the full year of 2006, we expect:

- Revenue of approximately \$155.0 to \$159.0 million;
- Operating margins of approximately 20%;
- Net income of approximately \$21.7.0 to \$23.0 million, resulting in diluted EPS of approximately \$0.73 to \$0.77 per share, which is based on an estimated 29.7 million diluted shares and an effective tax rate of 38%; and
- Cash net income of approximately \$21.9 to \$23.2 million after adding back the tax adjusted amortization of intangibles of approximately \$200,000, which would result in cash EPS of approximately \$0.74 to \$0.78 per share based on an estimated 29.7 million diluted shares and an effective tax rate of 38%.

Planned Acquisition of WebCT, Inc.

Blackboard expects that its planned acquisition of WebCT, Inc. will close late this year or in early 2006. Assuming the merger closes by early 2006, Blackboard expects the transaction to be moderately accretive to cash earnings on a non-GAAP basis and dilutive on a GAAP basis for calendar year 2006, and significantly accretive to cash earnings on a non-GAAP basis and dilutive on a GAAP basis for calendar year 2007. Non-GAAP cash earnings exclude the impact of certain acquisition and integration costs, amortization of intangibles, stock-based compensation expense and the impact of purchase accounting adjustments.

Conference Call & Webcast

The Blackboard management team will host a financial analyst and investor conference call today at 5:00 p.m. EDT. The call can be accessed at (866) 578-5788 (U.S.) or (617) 213-8057 (International) and using the reference code 56789081. Interested parties can also access the webcast through the Investor Relations section of the Company's Web site at http://investor.blackboard.com. Please access the Web site at least 15 minutes prior to the start of the call to register, download and install any necessary software.

For those unable to listen to the live conference call, a telephone replay will be available at (888) 286-8010 or (617) 801-6888 (reference code 18422925), through October 19, 2005.

About Blackboard

Blackboard is a leading provider of enterprise software and services to the education industry. The Company's product line consists of five software applications bundled in two suites, the *Blackboard Academic Suite* TM and the *Blackboard Commerce Suite*. Blackboard's clients include colleges, universities, schools and other education providers, as well as textbook publishers and student-focused merchants that serve education providers and their students. Blackboard is headquartered in Washington, D.C., with offices and staff in North America, Europe and Asia.

Forward Looking Statements

Any statements in this press release about future expectations, plans and prospects for Blackboard and other statements containing the words "believes," "anticipates," "plans," "expects," "will," and similar expressions, including statements about the expected impact of the acquisition on Blackboard's earnings per share in future periods, constitute forward-looking statements within the meaning of The Private Securities Litigation Reform Act of 1995. Actual results may differ materially from those indicated by such forward-looking statements as a result of various important factors, including the timing and extent of regulatory review, the timing of the closing, the ability of Blackboard to integrate the business, operations and personnel of WebCT following the acquisition, and the ability of both companies to retain their existing customers and gain new customers before and after the closing of the acquisition. Other factors that could affect the results discussed in our forward-looking statements include those set forth in the "Risk Factors" section of our most recent 10-Q filed with the SEC. In addition, the forward-looking statements included in this press release represent the Company's views as of October 12, 2005. The Company anticipates that subsequent events and developments will cause the Company's views to change. However, while the Company may elect to update these forward-looking statements at some point in the future, the Company specifically disclaims any obligation to do so. These forward-looking statements should not be relied upon as representing the Company's views as of any date subsequent to October 12, 2005.

###

Exhibit 99.4

News Release

Blackboard and WebCT Announce Agreement to Merge

The combined Companies' 3,700 academic clients will create an unparalleled Community of Practice enabling collaboration and innovation around the world

WASHINGTON, DC – October 12, 2005 – Two leading providers of enterprise software and services to the education community, Blackboard Inc. (NASDAQ: BBBB) and WebCT, Inc., today announced a definitive agreement to merge. Under terms of the agreement, Blackboard will acquire WebCT in a cash transaction for \$180 million, which values the offer at approximately \$154 million, net of WebCT's August 31, 2005 cash balance of \$26 million. The ultimate value of the offer will vary depending on WebCT's cash balance at closing. The transaction will combine two pioneering academic e-Learning organizations into a single company with the client base, resources and expertise to uniquely meet the rapidly evolving needs of educators around the world.

The merger of Blackboard and WebCT marks a major milestone in the build-out of networked learning environments by colleges, universities, schools and other education providers. It creates a large, collaborative community of academic institutions—a global e-Learning "Community of Practice"—supported by unsurpassed resources for innovation and support. Currently, more than 3,700 higher education, K-12, corporate, government and commercial academic institutions enhance their learning environments with solutions offered by the two organizations. Bringing Blackboard and WebCT together provides a powerful platform for innovating and supporting what at many institutions is the least developed, yet one of the most strategic segments of their technology infrastructure. Further, it will break down barriers and enable collaboration across institutions by developers and end-users at a critical time in the evolution of e-Learning technologies.

"This merger makes tremendous sense for our clients, shareholders and employees. It marks an unparalleled opportunity for two successful, mission-driven organizations to unify with a singular focus on being the premier partner and platform for educators on a global basis," said Michael Chasen, Blackboard President and Chief Executive Officer. "Together with our clients, we have

one of those rare and special opportunities to truly improve the access, quality and efficiency of education on a global scale."

"Both companies are passionate about the academic market and, alongside our respective clients, have truly pioneered the category together," said Carol Vallone, WebCT President and Chief Executive Officer. "Given the alignment of our visions, technologies, and overall strategies, the combination of our two companies will advance the teaching and learning technology industry, benefiting customers worldwide."

Both companies' Boards of Directors have approved the merger. Subject to regulatory approval and other customary closing conditions, the transaction is expected to close late this year or in early 2006. The combined companies will operate under the Blackboard name and brand with corporate headquarters located in Washington, DC. Operations will continue in WebCT's facilities in Massachusetts and Canada. Michael Chasen will continue to serve as President and CEO.

Complementary Strengths and Synergies

The merger of Blackboard and WebCT unites two innovators servicing academic institutions, both with complementary heritages in academia. Key strengths of the combination include:

- Top rated, proven e-Learning solutions recognized for ease-of-use, flexibility and scalability
- A broad, unified platform for publishers and e-Learning software developers that spans the globe
- A world-class service and support organization benefiting from the best-practices and infrastructure investments of both companies
- A stronger presence in the global education markets, including an expanded international footprint, including the UK, Canada, Australia, Japan, South Africa, Finland, Netherlands, Singapore, Spain, Hong Kong and Ireland
- Expanded base of Blackboard clients, allowing Blackboard to better serve those WebCT e-Learning institutions that also use the *Blackboard Commerce Suite* TM product set
- Larger revenue streams, enabling greater investment for innovation

Client Continuity and Product Strategy

Blackboard plans to enhance and support the existing products of both WebCT and Blackboard. New releases and ongoing software maintenance for WebCT and Blackboard will continue to be a key focus of the combined company. At the same time, Blackboard will leverage the *Blackboard Building Blocks*® architecture and WebCT's PowerLinks framework as the joint foundation for these enabling interfaces that will allow the existing product lines to interoperate with one another and with other applications. Over time, Blackboard will incorporate the best features and usability characteristics from the two product lines into a new, standards-based product set that will incorporate a scalable, architecture, the use of Web services, unparalleled ease of use, and flexible customization features.

"I have had experience with both companies and view this merger as combining excellence with excellence to advance the e-Learning industry," said Jack Wilson, President of The University of Massachusetts. "I also see this combination as a way to break down barriers across the Commonwealth of Massachusetts, and to open the door to new opportunities for collaboration among institutions using different e-learning platforms."

Financial Details

WebCT offers many of the same financial characteristics of Blackboard's business model, including an annual recurring subscription-based licensing model, ratable revenue recognition, a stable educational institutional client base and historically high renewal rates. As a result, the combination is expected to enhance growth and profitability over time. Blackboard expects the acquisition to close late this year or in early 2006. Assuming the merger closes by early 2006, Blackboard expects the transaction to be moderately accretive to earnings on a non-GAAP basis and dilutive on a GAAP basis for calendar year 2006, and significantly accretive to earnings on a non-GAAP basis and dilutive on a GAAP basis for calendar year 2007. The combined company expects to realize significant efficiencies by leveraging shared development infrastructure, and mitigating duplicative marketing initiatives and administrative expenditures.

Blackboard retained Credit Suisse First Boston as its financial advisor and WilmerHale and McDermott, Will & Emery as its legal advisors. WebCT retained Goldman Sachs & Co. as its financial advisor and Choate, Hall & Stewart and White & Case as its legal advisors.

Additional Information

The Blackboard management team will host a financial analyst and investor conference call today at 5:00 p.m. EDT. The call can be accessed at (866) 578-5788 (U.S.) or (617) 213-8057 (International) and using the reference code 56789081. The conference call will also be web cast at http://investor.Blackboard.com. For those unable to listen to the live conference call, a telephone replay will be available at (888) 286-8010 or (617) 801-6888 (reference code 18422925), through October 19, 2005.

Blackboard's CEO, Michael Chasen, and WebCT's CEO, Carol Vallone, will hold multiple meetings open to clients and other educators to discuss the merger and answer questions. These meetings will take place at the Educause conference during each company's corporate presentations scheduled for Wednesday, October 19, 2005, 11:40 AM EDT in Meeting Room W204A and on Thursday, October 20, 2005, 2:20 PM EDT in Meeting Room W204A.

In addition, Blackboard has launched a new portion of its Web site, www.blackboard.com/webct, where anyone can track the latest news and information about the merger.

About Blackboard

Blackboard is a leading provider of enterprise software and services to the education industry. The Company's product line consists of five software applications bundled in two suites, the *Blackboard Academic Suite* TM and the *Blackboard Commerce Suite*. Blackboard's clients include colleges, universities, schools and other education providers, as well as textbook publishers and student-focused merchants that serve education providers and their students. Blackboard is headquartered in Washington, D.C., with offices and staff in North America, Europe and Asia.

About WebCT

Based in Lynnfield, MA, WebCT provides a highly flexible e-learning environment that empowers institutions across the educational spectrum with the ability to achieve their unique objectives for faculty empowerment, administrative efficiency and student outcomes. Colleges and universities around the world, from community colleges to large university consortia, are using WebCT to expand the boundaries of teaching and learning.

Forward Looking Statements

Any statements in this press release about future expectations, plans and prospects for Blackboard and other statements containing the words "believes," "anticipates," "plans," "expects," "will," and similar expressions, including statements about the expected impact of the acquisition on Blackboard's earnings per share in future periods, constitute forward-looking statements within the meaning of The Private Securities Litigation Reform Act of 1995. Actual results may differ materially from those indicated by such forward-looking statements as a result of various important factors, including the timing and extent of regulatory review, the timing of the closing, the ability of Blackboard to integrate the business, operations and personnel of WebCT following the acquisition, and the ability of both companies to retain their existing customers and gain new customers before and after the closing of the acquisition. Other factors that could affect the results discussed in our forward-looking statements include those set forth in the "Risk Factors" section of our most recent 10-Q filed with the SEC. In addition, the forward-looking statements included in this press release represent the Company's views as of October 12, 2005. The Company anticipates that subsequent events and developments will cause the Company's views to change. However, while the Company may elect to update these forward-looking statements at some point in the future, the Company specifically disclaims any obligation to do so. These forward-looking statements should not be relied upon as representing the Company's views as of any date subsequent to October 12, 2005.

###

